

Delft University of Technology

## Integrated Circuits for 3D High-Frame-Rate Intracardiac Echocardiography Probes

Hopf, Y.M.

DOI 10.4233/uuid:7016e74d-c7df-42a8-8257-11326940ad7f

Publication date 2023

**Document Version** Final published version

#### Citation (APA)

Hopf, Y. M. (2023). Integrated Circuits for 3D High-Frame-Rate Intracardiac Echocardiography Probes. [Dissertation (TU Delft), Delft University of Technology]. https://doi.org/10.4233/uuid:7016e74d-c7df-42a8-8257-11326940ad7f

#### Important note

To cite this publication, please use the final published version (if applicable). Please check the document version above.

Copyright Other than for strictly personal use, it is not permitted to download, forward or distribute the text or part of it, without the consent of the author(s) and/or copyright holder(s), unless the work is under an open content license such as Creative Commons.

#### Takedown policy

Please contact us and provide details if you believe this document breaches copyrights. We will remove access to the work immediately and investigate your claim.

This work is downloaded from Delft University of Technology. For technical reasons the number of authors shown on this cover page is limited to a maximum of 10.

# INTEGRATED CIRCUITS FOR 3D HIGH-FRAME-RATE INTRACARDIAC ECHOCARDIOGRAPHY PROBES

# INTEGRATED CIRCUITS FOR 3D HIGH-FRAME-RATE INTRACARDIAC ECHOCARDIOGRAPHY PROBES

# Dissertation

for the purpose of obtaining the degree of doctor at Delft University of Technology by the authority of the Rector Magnificus, Prof.dr.ir. T.H.J.J. van der Hagen, chair of the Board for Doctorates to be defended publicly on Friday 3 March 2023 at 10:00 o'clock

by

# Yannick Matthias HOPF

Master of Science in Electrical Engineering and Information Technology, Technische Universität Darmstadt, Germany born in Seligenstadt, Germany. This dissertation has been approved by the promotors.

Composition of the doctoral committee:

| Rector Magnificus,        | chairperson                              |
|---------------------------|--|
| Dr. ir. M.A.P. Pertijs,   | Delft University of Technology, promotor |
| Prof. dr. ir. N. de Jong, | Delft University of Technology, promotor |

Independent Members: Prof. Dr.-Ing. M. Ortmanns, Prof. dr. ir. W.A. Serdijn, Prof. dr. ir. R. Dekker, Dr. ir. P.J.A. Harpe,

University of Ulm, Germany Delft University of Technology Delft University of Technology Eindhoven University of Technology

Other Members: Dr. ir. J.G. Bosch,

Erasmus University Medical Center

This research is part of the project 3D-ICE with project number 14279 of the Open Technology Programme which is partly financed by the Dutch Research Council (NWO).





Keywords:Ultrasound, ICE, ASIC, 3D, High-Frame-RatePrinted by:Proefschrift Specialist | www.proefschriftspecialist.nlCover by:Yannick Hopf

Copyright © 2023 by Y. Hopf ISBN 978-94-6384-419-2 An electronic version of this dissertation is available at http://repository.tudelft.nl/.

To Lea and my Family

# CONTENTS

| INTRODUCTION   | 1  |
|--|----|
| 1.1 Motivation & Background                            | 1  |
| 1.1.1 Cardiovascular Conditions                        | 1  |
| 1.1.2 Impact of Ultrasound Imaging                     | 2  |
| 1.1.3 Conventional Ultrasound Probes                   | 5  |
| 1.1.4 Integrated Circuits in Ultrasound Probes         | 7  |
| 1.2 Intracardiac Echocardiography Probes               | 10 |
| 1.2.1 Developments                                     | 10 |
| 1.2.2 Challenges & Opportunities                       | 11 |
| 1.3 Thesis Organization                                | 13 |
| References   | 15 |
| INA CINC COURSE FOR 2D HIGH FRAME RATE ICE             | 25 |
| IMAGING SCHEME FOR 3D HIGH-FRAME-RATE ICE              | 20 |
| 2.1 Introduction                                       | 25 |
| 2.1.1 Clinical Background                              | 25 |
| 2.1.2 Electromechanical Wave Imaging                   | 26 |
| 2.1.3 High-Frame-Rate 3D Intracardiac Echocardiography | 27 |
| 2.1.4 Data Rate Reduction by Micro-Beamforming         | 28 |
| 2.1.5 3D ICE Imaging Scheme Design                     | 30 |
| 2.2 Materials and Methods                              | 30 |
| 2.2.1 Imaging Scheme Design: Parameters                | 30 |
| 2.2.2 Steerable Fan-Shaped Transmit Beam               | 31 |
| 2.2.3 Angular Weighted Averaging                       | 31 |
| 2.2.4 Simulation Setup                                 | 32 |
| 2.2.5 Numerical Phantoms                               | 34 |
| 2.2.6 Evaluation Criteria                              | 35 |
| 2.3 Results  | 36 |
| 2.3.1 Imaging Scheme Optimization                      | 36 |
| 2.3.2 Imaging Performance Evaluation                   | 39 |
| 2.3.3 Imaging of a Beating Atrium Phantom              | 41 |
| 2.4 Discussion and Conclusion                          | 44 |
| Appendix I: Fan-Shaped Transmit Beam Generation        | 48 |
| References   | 49 |

| TRASOUND PROBES                        | 55     |
|--|--------|
| 3.1 Introduction                       | 55     |
| 3.2 Circuit Implementation             |        |
| 3.2.1 High-Voltage Pulser              |        |
| 3.2.2 High-Voltage Level Shifter       |        |
| 3.2.3 High-Side Regulator              |        |
| 3.3 Measurement Results                |        |
| 3.4 Conclusion                         |        |
| References                             |        |
| A                                      |        |
| AN ASIC WITH SHARED HYBRID BEAMFORMING | G ADC  |
| FOR HIGH-FRAME-RATE 3D ICE             | 67     |
| 4.1 Introduction                       |        |
| 4.2 System Design                      |        |
| 4.2.1 Overview                         |        |
| 4.2.2 Architecture                     |        |
| 4.2.3 Hybrid ADC                       |        |
| 4.2.4 Datalink                         |        |
| 4.3 Circuit Implementation             |        |
| 4.3.1 Timing                           |        |
| 4.3.2 Comparator                       |        |
| 4.3.3 Logic & CDAC                     |        |
| 4.3.4 Calibration                      |        |
| 4.4 Experimental Results               |        |
| 4.4.1 Electrical Measurements          |        |
| 4.4.2 Acoustic Measurements            |        |
| 4.5 Conclusion                         |        |
| References                             |        |
|  |        |
| A HIGH-FRAME-RATE ULTRASOUND ASIC WITH | MULTI- |
| LEVEL LOAD MODULATION DATALINK         | 105    |
| 5.1 Introduction                       |        |
| 5.2 System Design                      |        |
| 5.2.1 Overview                         |        |
| 5.2.2 Architecture                     |        |
| 5.3 Circuit Implementation             |        |
| 5.3.1 TX Beamformer                    |        |
| 5.3.2 LNA                              |        |
| 5.3.3 PGA                              |        |
| 5.3.4 Datalink                         |        |

| 5.4 Experimental Results      |     |
|-------------------------------|-----|
| 5.4.1 Electrical Measurements |     |
| 5.4.2 Acoustic Measurements   |     |
| 5.5 Conclusion                |     |
| References                    |     |
| CONCLUSION                    | 143 |
| 6.1 Main Contributions        |     |
| 6.2 General Findings          |     |
| 6.3 Future Work               |     |
| References                    |     |
| SUMMARY                       | 153 |
| SAMENVATTING                  | 157 |
| LIST OF ABBREVIATIONS         | 161 |
| LIST OF PUBLICATIONS          | 165 |
| ACKNOWLEDGEMENTS              | 167 |
| About the author              | 171 |

# INTRODUCTION

## 1.1. Motivation & Background

#### 1.1.1. Cardiovascular Conditions

I n 2019, the World Health Organization (WHO) evaluated that cardiovascular diseases (CVDs) are the leading cause of death in the world, with a rising share as numbers further increased while deaths due to communicable diseases have been decreasing over the last two decades [1]. Cardiovascular diseases include a wide range of conditions in the vascular system and heart but the cause of death commonly is the acute event of a stroke or heart attack. These are usually caused by restricted blood flow to the brain or heart, e.g. due to cerebrovascular or coronary heart disease [2].

Next to lifestyle changes and medicinal treatment, cardiovascular interventions are an important tool in dealing with cardiovascular diseases [3]. These include heavily invasive procedures like heart transplantation, artificial heart operations, or coronary bypass surgery that require a complex open-chest intervention.

However, there are also less intrusive, minimally-invasive procedures that are applied to target specific conditions in the heart and vascular system. Balloon angioplasty uses a catheter to move through an artery and inflate a balloon at the place where a blockage has been identified. This can open the artery to increase the blood flow and a stent can be placed to prolong the effect [4]. Other examples of the treatment of structural heart conditions include the transcatheter closure of a patent foramen ovale, atrial septal defect [5], or ventricular septal defect [6] and pulmonary [7] or mitral [8] valvuloplasty. Additionally, minimally invasive procedures can also be applied to target electrical heart defects that cause atrial fibrillation, the most common heart arrhythmia with a 33% increase over the last 20 years and a risk factor for having a stroke [9]. During this so-called electrophysiology with catheter ablation, an electrical map of the heart is obtained and afterwards faulty tissue is ablated to restore a regular heartbeat [10].

While not all cardiac interventions can be performed with minimally-invasive procedures, they have shown a reduced risk, can typically work on local anesthesia, and will see a rising share compared to traditional surgery [11].

#### 1.1.2. Impact of Ultrasound Imaging

The use of minimally-invasive interventions in the heart requires that procedure guidance can move from direct vision, as predominantly used in open-chest surgery, to imaging tools. In the following, several medical imaging tools are discussed.

In 1895, x-rays were first discovered by Wilhelm Conrad Röntgen and used to observe the structure and function of the heart [12]. X-rays are a form of electromagnetic radiation with wavelengths between 0.01 to 10 nm and can generally pass through human tissue. An image is formed by recording the remaining radiation after passing through the body and thus displays the local density variation in the medium on a capturing screen [13]. Due to their high energy, x-rays belong to the class of ionizing radiation, meaning that they can harm human tissue, being able to cause cancer for prolonged exposure [14]. In current minimally-invasive procedures, x-rays are still frequently used during fluoroscopy to image the process inside the body [15]. Fluoroscopy generally works with the continuous application of radiation and its capturing on a fluorescing screen to provide real-time investigation and procedure guidance. Fig. 1.1(a) shows a typical setup with a table for the patient to lie on and the scanner above. While plain x-ray imaging and fluoroscopy generally suffer from superimposition in the depth of the image, x-rays can also be used in a computed tomography (CT) procedure to generate highquality cross-sections of the body. During CT imaging, with a scanner as shown in Fig. 1.1(b), several acquisitions are obtained at different angles and combined to reconstruct the data of each voxel in the chosen plane [13].

Magnetic resonance imaging (MRI) is another popular tool in medical imaging

 $\mathbf{2}$ 



Figure 1.1: (a) Classical patient-side fluoroscopy system (Precision 600, GE Healthcare, USA); (b) CT scanner (CT 5000 Ingenuity, Philips, The Netherlands); (c) MRI scanner (MAGNETOM Free.Max, Siemens Healthineers, Germany); (d) Hand-held ultrasound probe (PM1-6A, Samsung, Seoul); (e) TEE probe (Adult 4D TEE Probe Premium, Oldefit Ultrasound, The Netherlands); (f) ICE catheter (ACUSON AcuNav Volume ICE Catheter, Siemens Healthineers, Germany)

and can produce high-resolution images without the application of ionizing radiation. An MRI scanner, such as displayed in Fig. 1.1(c), instead applies a strong magnetic field in a pulsed fashion to force the alignment of the hydrogen atoms in the body to it. Subsequently, it measures the radiofrequency waves emitted from the body when the atoms break alignment again. However, a major limitation to the application in minimally-invasive procedures is that the strong magnetic fields prevent imaging in the presence of implants, such as cardiac pacemakers, and restrict the use of tools with metallic components [13], [16].

Ultrasound imaging can in many cases present an alternative without this limitation and a low usage threshold. While ultrasound was already used for medical therapeutic purposes in the 1930s, it wasn't until 1947 that Karl Theo Dussik was able to create the first ultrasonically generated images. Since then, it has risen to be a common medical imaging tool [17]. Medical ultrasonic imaging is based on sound waves ranging from several MHz to several tens of MHz. A transducer transmits pressure waves and subsequently captures the returning echoes from the medium. No need for harmful ionizing radiation and the ability to generate highresolution 3D images are clear advantages over fluoroscopy. Compared to CT and MRI, ultrasonic imaging requires less hardware and acquisition time, translating into a lower cost and better compatibility for active procedure guidance [18].

One of the most common applications of ultrasound for imaging the heart is in a transthoracic echocardiography (TTE) procedure with a cardiac probe similar to Fig. 1.1(d). During the investigation, the hand-held probe is positioned on the chest of the patient and acquisitions are obtained through the rib cage [19]. However, as the pressure wave attenuation is proportional to the traveled distance as well as, for heart tissue, approximately the ultrasound frequency, only relatively low frequencies can be applied [20]. This limits the obtained resolution and, combined with the limited acoustic window through the ribs and the need for a dedicated operator, makes TTE less favorable for the guidance of minimallyinvasive interventions [21].

To enable higher image quality, special probes for use in transesophageal echocardiography (TEE) or intracardiac echocardiography (ICE) procedures were first developed in the 1960s [17]. TEE probes have a transducer array at the tip of a gastroscopic tube as shown in Fig. 1.1(e) and can get close to the heart by moving through the esophagus. This enables them to obtain high-resolution cardiac images while still providing substantial space for the integration at its tip and inside the tube for connection with an imaging system outside the body [22]. However, the application can lead to substantial discomfort and conscious sedation is frequently applied to diminish pain and anxiety at the cost of an increased number of potential complications [23]. This is avoided in the use of ICE probes, as displayed in Fig. 1.1(f), that can be introduced into the vascular system via a small incision in the groin area and then image from within the heart [24], [25]. Additionally, even higher spatial resolution can potentially be reached as the distance to the area of interest decreases further [26]. As the transducer is integrated at the tip of a catheter with a diameter of only around 3 mm diameter, integration



Figure 1.2: Conceptual drawing of a typical ultrasound imaging setup.

becomes more challenging though [27].

In this work, the possibilities for ultrasonic imaging with intracardiac echocardiography procedures and the design of probes are therefore further investigated.

#### **1.1.3.** Conventional Ultrasound Probes

Traditional ultrasound imaging setups include a passive transducer connected with a cable to an imaging system as conceptually shown in Fig. 1.2 [28]. The transducer is conventionally made from lead zirconate titanate (PZT) but transducer designs based on microelectromechanical systems (MEMS), such as capacitive micromachined ultrasonic transducers (CMUTs) [29] and piezoelectric micromachined ultrasonic transducers (PMUTs) [30], are also entering the field. While these can achieve comparable performance, are compatible with wafer-level volume production, and can be monolithically integrated with CMOS technologies, they do come with their own challenges. These include the requirement of a high-voltage transducer bias for CMUTs and high process variation for PMUTs [31], [32].

The transducer elements are then connected with individual micro-coaxial cables within the probe cable. In the imaging system, a transmit/receive (T/R)switch is connecting the element either to the transmit (TX) path for the excitation of ultrasound waves or to the receive (RX) path to capture the resulting echoes from the imaged medium. During TX, the imaging system applies electrical transmit patterns with voltage amplitudes in the order of 10s to 100s of volts to enable a sufficient signal-to-noise ratio (SNR) for image reconstruction. The transmission is controlled by a TX beamformer that applies delays to the signals of the individual elements in order to determine the direction and divergence of



Figure 1.3: (a) Illustration of transmit beamforming of a diverging wave with six transducer elements; (b) Conceptual drawing of delay-and-sum beamforming.

the ultrasound beam as shown in Fig. 1.3(a).

During RX, on the other hand, returned signals down to the level of  $\mu$ Vs have to be amplified before further processing is possible. Since the attenuation of the ultrasound waves is dependent on the distance traveled and thus the time passed after TX, the gain is commonly adjusted with time in a so-called time gain compensation (TGC). This can reduce the total dynamic range (DR) requirement, which can go beyond 100 dB, for the following system to the instantaneous DR of interest as shown in Fig. 1.4. The received signals are subsequently digitized at ideally 4 to 10 times the transducer center frequency [33] and fed to an image processing block. Fig. 1.3(b) conceptually illustrates how the information from each voxel is, in the simplest case, reconstructed by delay-and-sum (DAS) beamforming [28].



Figure 1.4: Illustration of how time gain compensation can reduce the total dynamic range of the received signal to only the range of interest.

A major drawback of the traditional probe configuration is that each transducer element requires an individual micro-coaxial cable for connection. This leads to a significant cost and complexity factor in the assembly but also adds substantial loading to the transducer, resulting in signal attenuation [34]. This is particularly concerning for CMUTs as they have a relatively high impedance [35] but also generally an issue when going to 3D imaging. While 2D plane imaging requires only a 1D array of transducer elements (the second dimension being the recorded time of arrival of the echoes) 3D imaging generally requires a 2D matrix of elements. As both azimuth and elevation direction have a pitch limitation to stay below an acceptable level of unwanted effects such as grating lobes, the elements of the 2D matrix are quite small and have a higher impedance [36].

The interconnect to the imaging system also becomes a bottleneck simply because of the limited diameter of the probe shaft. Hand-held devices shouldn't become too stiff to operate and catheter-based probes are fundamentally limited by the dimensions of the human body. Thinner cables can sometimes be applied but generally introduce stronger loading and more attenuation [37], [38]. The consequence is that there is a limit to the number of elements of any probe if implemented in the traditional way.

#### 1.1.4. Integrated Circuits in Ultrasound Probes

Application-specific integrated circuits (ASICs) can be employed to alleviate the challenges of conventional ultrasound imagers and enable 3D imaging even with catheter-based probes. Fig. 1.5 gives an overview of a possible ultrasound system design with an ASIC at the tip of the probe [27], [39]. The loading of the transducer element is reduced by moving part of the electronics in front of the cable [40].



Figure 1.5: Example of an ultrasound system with ASIC inside the imaging probe.

This usually includes the T/R switch as well as TX- and RX-related circuitry. The transmitter can be implemented as high-voltage (HV) pulsers [41], [42] or also just as switches gating a high-voltage signal from the external system [43], [44]. The receiver typically at least features a low-noise amplifier (LNA) that can be tailored to the applied transducer [45]. A variable gain stage is often also applied to enable time gain compensation [32], [46]. Digitization with analog-todigital converters (ADCs) can moreover be added to make the sensitive received signals more robust, allow digital data transmission, and facilitate upcoming signal processing techniques such as machine-learned data reduction [47], [48].

Another major opportunity provided by in-probe ASICs is cable-count reduction. It can enable 2D transducer arrays with a high element count for 3D imaging but also be applied to just reduce the cost of a cable assembly or reduce the stiffness of the probe [49], [50]. However, the cable count is actually higher to start with as each element in Fig. 1.5 theoretically requires a separate connection for TX and RX while this was shared for the probe without an ASIC. This can be more than compensated by the use of circuit design techniques though.

In TX, it has been shown that, despite a large number of elements, beamforming is possible with only a few cables with the implementation of an on-chip beamformer. This can be configured with an efficient serial link into local registers [51], [52] or a compact shift register [39], [53]. Another possibility to reduce the hardware, albeit with restricted TX beamforming capabilities, is the use of a row/column approach to generate transmit patterns [43], [54], [55].

In RX, the channel-count reduction can't be as efficient due to the larger output bandwidth but significant factors have still been achieved with multiplexing

1



Figure 1.6: Conceptual overview of subarray beamforming operation in the image reconstruction.

and subarray beamforming<sup>1</sup>. Analog output signal multiplexing has been shown in the time [56], [57] and frequency domain [58]. Reduction rates in the order of 10x could be achieved but the sensitive analog signals suffer from bandwidth limitation and channel-to-channel crosstalk on the small cables [59]. This can be addressed by moving to digital time-division multiplexing (TDM) with better tolerance to cross-talk, interference, and noise [60]. Another beneficial feature of multiplexing is that it is generally compatible with subarray beamforming. Subarray beamforming does receive beamforming of subarrays of the transducer matrix already in the probe instead of completely in the external imaging system as shown in Fig. 1.6 [61]. The process is also based on DAS operations with delay configurations to pre-steer the receive beam and can reduce the number of channels by only transmitting the combined signal. The combination between TDM and  $\mu BF$  has been shown in digital beamforming of element-level signals [62]–[64] and analog beamforming followed by digitization and TDM [65]–[67]. This has shown channelcount reductions of up to 36x and is thus of particular interest for the following discussion on the design of ICE probes with cable diameters in the order of 3 mm [27].

<sup>&</sup>lt;sup>1</sup>Also known as micro-beamforming (µBF) subaperture beamforming, or subaperture processing (SAP).

## **1.2.** Intracardiac Echocardiography Probes

#### **1.2.1.** Developments

The first recorded intracardiac ultrasound experiments go back to 1956, not even ten years after the first general ultrasound imaging [17], [68]. The early study already applied catheters with a single-element ultrasound tip and saw the development of rotating and multi-element probes shortly after, in the mid-1960s [69]. This was followed by the first 32-element, 9-Fr catheter which could obtain realtime, two-dimensional cross-sections from within the heart in 1971 [70]. However, the transducer elements were still assembled around the catheter tip and it took until 1996 before the first 128-element, side-looking phased-array ICE catheter was reported [71]. Since then, several phased-array transducers for intracardiac imaging, such as the AcuNav catheter from Siemens/Biosense Webster, ViewFlex from St. Jude Medical, and SOUNDSTAR from Biosense Webster, have become commercially available and are being used in clinical procedures [34], [72]. All of these are 64-element, side-looking probes ranging from 8 to 10 Fr in size but are also all still limiting the field of view to 2D cross-sections of the heart.

As 3D visualization is quite interesting in the guidance of minimally-invasive procedures [15], it was initially explored in research how 3D images could be obtained from manual rotation of a 1D array [73]. The method was able to successfully reconstruct 3D images of the heart but not in real-time, given a mapping period in the order of 20 minutes. This could be alleviated with a probe that replaced the manual rotation with a rotating 1D transducer array at the tip [74]. The design achieved real-time volume acquisition but was still limited by a frame rate of 7 volumes/s at an opening angle of 40°. In the device described in [75], another approach was taken by applying a twisted 1D array instead of device rotation. While this can only provide a limited opening angle of 22°, there is no inherent limitation to the frame rate associated with it.

To enable 3D intracardiac ultrasound imaging without the deficiencies of 1D transducer arrays, 2D matrices have been investigated in parallel<sup>2</sup>. The first reported design goes back to 2001 and featured 64 active channels individually wired through the catheter [77]. More designs with up to 512 elements followed but saw a rising complexity in the interconnect with only limited image quality [37], [38].

1

<sup>&</sup>lt;sup>2</sup>In the class of forward-looking probes [42], [76], 3D imaging has also been shown but is not further discussed as they typically serve in close conjunction with therapy catheters and have limited imaging capabilities.

It wasn't until the co-integration with electronics at the tip of the catheter to reduce the cable count and buffer the received signals that the first real-time 3D ICE probe was shown by Wildes *et al.* in 2016 [27]. It featured a transducer matrix of  $60 \times 14$  elements in a 10-Fr catheter, could cover a field of view of  $90^{\circ} \times 90^{\circ} \times 8$ cm, and could image at a rate of up to 50 volumes/s. Since then, several commercial parties, such as Oldelft Ultrasound with its 4D ICE Transducer, Siemens Healthineers with its ACUSON AcuNav Volume ICE Catheter, and Philips with its VeriSight Pro, have announced their new 3D ICE probes.

The following section gives an overview of the different challenges in the design of these 3D probes and also presents opportunities for future ICE catheters.

#### 1.2.2. Challenges & Opportunities

Something to generally take into account is the acceptable probe temperature and thus power consumption. While conventional, passive probes typically only face the heating from thermal losses in the transduction process, the new generation of probes with co-integrated ASICs also have to deal with the additional power consumption of the electronics. There is no clearly defined limit to the power an intracardiac catheter may dissipate in order to gain clearance but requirements for a maximum surface temperature of 43°C have been formulated by the US Food and Drug Administration (FDA) [78], [79]. This absolute value poses a tighter requirement on the power consumption per element for 3D probes with 2D transducers compared to phased array catheters for 2D imaging. While the implications aren't as severe at the frontend as the smaller elements also reduce the power per element in TX and ease the noise requirement per channel in RX. circuit overhead and especially the possibly higher output data bandwidth need to be considered [46], [65]. Ultimately, it is quite complicated to derive a power consumption limit from the temperature regulation as the translation depends heavily on the applied environment, thermal design of the probe, and imaging conditions. Even a small blood flow can for example already lead to a significant temperature distribution and short burst acquisitions at high power consumption can only result in a minor temperature increase due to the low-pass characteristic of the mechanical structure [80]. The general approach in this thesis is therefore to minimize power consumption as much as possible.

Similar to the power overhead when progressing from 2D to 3D imaging catheters, there is also a relative increase in the need for circuit area in the transition which can result in a significant challenge. The circuitry of each channel or unit cell should ideally be matched to the pitch of the transducer elements to ease the system design and enable scalability to larger apertures [81]. The pitch is typically determined by the transmission frequency in order to minimize the effect of grating lobes that are at angles

$$\Theta_g = \pm \arcsin\left(\frac{m\lambda}{p}\right) \tag{1.1}$$

where  $\Theta_{\rm g}$  is the angle of the grating lobe for m  $\neq 0$ , p is the pitch and  $\lambda$  is the wavelength of the ultrasound wave. The maximum pitch, in order to not have grating lobes in the image, thus is half of the wavelength and commonly in the order of 100 µm to 200 µm for ICE probes [20]. This forms a tough requirement for the circuitry and has led to many designs that either couldn't match the electronics to the transducer pitch [63], [67], sized the elements substantially larger than half of the wavelength [32], [62] or didn't implement all of the desirable features [65], [82]. Particularly the transmit-related circuits are very area-intense as the application of high voltages to excite sufficiently strong pressure waves requires the use of transistors with large isolation structures [39]. As a consequence, one of the main goals of this work is to explore architectures with a high area efficiency and optimize blocks accordingly.

Next to the limited area, the interconnect to the imaging system becomes a major concern with the larger number of elements in 2D transducer matrices and the small intracardiac catheters in the order of 10 Fr [37], [38]. Other than with power and area, there is no inherent scaling factor with the element size but a linear increase of necessary connection bandwidth with the number of elements if the signal is to be used for image reconstruction in the external unit unharmed. As designs are often implemented in mature technologies to make use of their HV transistor options and apply thin micro-coaxial cables, both imposing severe bandwidth limitations, an unfeasible number of cables would be required to transfer all the generated data in real-time [43], [46].

To cope with the discussed power, area, and interconnect requirements, two main methods have previously been shown. The first is to not implement all circuitry at the element level and allocate a fixed channel in the probe cable to each element. Instead, the elements are multiplexed over the course of multiple pulse-echo cycles [39], [50]. This is a direct relief to all discussed aspects but trades with a reduced frame rate. The second is subarray beamforming as described in section 1.1.4 and also applied in imaging catheters such as the first full 3D ICE

probe from Wildes *et al.* [27]. While compared to multiplexing there is more overhead in the delay and summation of element-level signals, this is also very effective in the reduction of output data bandwidth, can still reduce the required circuitry significantly, and theoretically reads out all the elements in real-time [81]. However, it does not provide the raw data of the full array and introduces focusing errors that result in increased grating and side lobe levels as well as broadening of the main beam. Narrower transmit beams can be used to mitigate these effects but require more acquisitions per volume, again limiting the frame rate [83].

These inherent trade-offs with frame rate are why one aspect that has not yet been reported in conjunction with high-image-quality 3D intracardiac probes as introduced in section 1.2.1 is high frame rate. A high frame rate is interesting to record phenomena with very short time windows, such as electromechanical or shear waves, in the heart. Their investigation can lead to additional diagnostic potential by e.g. being able to detect abnormal electrical activation patterns or create a stiffness map of the heart. However, they travel at speeds in the order of 1 m/s along the heart tissue and require a frame rate of at least 1000 volumes/s to be captured [84], [85].

Consequently, techniques to address the identified main challenges of power consumption, circuit area, and cable-count reduction without excessively limiting the frame rate are being investigated.

### **1.3.** Thesis Organization

The aim of this work is to explore opportunities in intracardiac ultrasound probes and arrive at a concept for a next-generation imaging catheter. This starts with the discussion of a desired transducer array design as well as an imaging scheme to enable the targeted acquisition modes. The focus is thereafter laid on innovations in the ASIC architecture and its circuit-level building blocks to address current challenges in the field. The thesis is organized in six chapters to present different aspects in logical units from the idea to the evaluation. The division of the chapters is as follows.

Chapter 2 discusses the value of a high-frame-rate 3D ICE probe design and proposes an imaging scheme to enable it. The scheme employs 1D subarray beamforming with seven steps in the elevation direction to achieve a high frame rate while reducing the channel count sufficiently for realization in a 10-Fr catheter. It is presented how the transmit beam divergence and center frequency are optimized based on the required image quality for electromechanical wave imaging (EWI). Numerical simulation results show that a set of 7 fan-shaped transmission beams can provide a frame rate of 1000 volumes/s and a sufficient spatial resolution to visualize the electromechanical wave propagation on a large 3D surface.

Chapter 3 introduces a compact HV transmit circuit for dense 2D transducer arrays used in 3D ultrasonic imaging systems. The chapter shows how stringent area requirements can be addressed by a unipolar pulser with an embedded transmit/receive switch. Combined with a capacitive HV level shifter, the circuit forms the ultrasonic HV transmitter with the lowest reported HV transistor count and area. It is shown how the novel level shifter can reduce sensitivity to transients on the HV supply caused by pulsing and how hardware sharing can further reduce the active circuit area. The description of a prototype of  $8 \times 9$  elements and its characterization is included to demonstrate the functionality and performance.

Chapter 4 presents an ASIC for 3D, high-frame-rate ultrasound imaging probes. The design is the first to combine element-level, high-voltage transmitters and analog frontends, subarray beamforming, and in-probe digitization in a scalable fashion for catheter-based probes. The integration challenge is met by a hybrid analog-to-digital converter, combining an efficient charge-sharing successiveapproximation-register (SAR) first stage and a compact single-slope (SS) second stage. Merging with the subarray beamformer and additional hardware-sharing between neighboring channels ultimately leads to the lowest reported area and power consumption across miniature ultrasound probe ADCs. Electrical, acoustic, and imaging experiments are shown to verify the design's functionality at up to 1000 volumes/s.

Chapter 5 discusses a second-generation ASIC for catheter-based high-framerate 3D ultrasound imaging probes. It presents a method to reduce the number of cables in the catheter while maintaining a small footprint per element by realizing transmit beamforming on the chip with a combination of a shift register (SR) and a row/column (R/C) approach. To explore an additional cable-count reduction in the receiver part of the design, a channel with a combination of time-division multiplexing, subarray beamforming, and 4-level pulse amplitude modulation (PAM-4) data transmission is also included. This achieves an 18-fold total cable-count reduction and minimizes the power consumption in the catheter by a load modulation (LM) cable driver. The functionality and performance are evaluated in the characterization of a prototype of  $16 \times 18$  transducer elements.

Chapter 6 concludes the thesis by summarizing the main findings and presenting suggestions for future research.

# References

- World Health Organization. "The Top 10 Causes of Death." (2020), [Online]. Available: https://www.who.int/news-room/fact-sheets/detail/thetop-10-causes-of-death (visited on 09/13/2022).
- [2] —, "Cardiovascular Diseases (CVDs)." (2021), [Online]. Available: https: //www.who.int/en/news-room/fact-sheets/detail/cardiovasculardiseases-(cvds) (visited on 09/13/2022).
- [3] D. P. Leong, P. G. Joseph, M. McKee, et al., "Reducing the Global Burden of Cardiovascular Disease, Part 2," *Circulation Research*, vol. 121, no. 6, pp. 695–710, Sep. 2017. DOI: 10.1161/CIRCRESAHA.117.311849.
- [4] P. W. Serruys, P. de Jaegere, F. Kiemeneij, et al., "A Comparison of Balloon-Expandable-Stent Implantation with Balloon Angioplasty in Patients with Coronary Artery Disease," New England Journal of Medicine, vol. 331, no. 8, pp. 489–495, Aug. 1994. DOI: 10.1056/NEJM199408253310801.
- [5] J. P. Giblett, L. K. Williams, S. Kyranis, L. M. Shapiro, and P. A. Calvert, "Patent Foramen Ovale Closure: State of the Art," *Radcliffe Cardiology*, vol. 15, no. 15, Nov. 2020. DOI: 10.15420/icr.2019.27.
- [6] B. H. Morray, "Ventricular Septal Defect Closure Devices, Techniques, and Outcomes," *Interventional Cardiology Clinics*, vol. 8, no. 1, pp. 1–10, Jan. 2019. DOI: https://doi.org/10.1016/j.iccl.2018.08.002.
- P. S. Rao, "Percutaneous balloon pulmonary valvuloplasty: State of the art," *Catheterization and Cardiovascular Interventions*, vol. 69, no. 5, pp. 747– 763, Feb. 2007. DOI: https://doi.org/10.1002/ccd.20982.
- [8] M. Nobuyoshi, T. Arita, S.-i. Shirai, et al., "Percutaneous Balloon Mitral Valvuloplasty," *Circulation*, vol. 119, no. 8, e211–e219, Dec. 2009. DOI: 10. 1161/CIRCULATIONAHA.108.792952.

- [9] G. Lippi, F. Sanchis-Gomar, and G. Cervellin, "Global epidemiology of atrial fibrillation: An increasing epidemic and public health challenge," *International Journal of Stroke*, vol. 16, no. 2, pp. 217–221, 2021. DOI: 10.1177/ 1747493019897870.
- [10] K. Nademanee, J. McKenzie, E. Kosar, et al., "A new approach for catheter ablation of atrial fibrillation: mapping of the electrophysiologic substrate," *Journal of the American College of Cardiology*, vol. 43, no. 11, pp. 2044– 2053, Jun. 2004. DOI: 10.1016/j.jacc.2003.12.054.
- [11] M. J. Mack, "Minimally invasive cardiac surgery," Surgical Endoscopy And Other Interventional Techniques, vol. 20, no. 2, S488–S492, Mar. 2006.
- [12] M. AlGhatrif and J. Lindsay, "A brief review: history to understand fundamentals of electrocardiography," *Journal of Community Hospital Internal Medicine Perspectives*, vol. 2, no. 1, Apr. 2012. DOI: 10.3402/jchimp.v2i1. 14383.
- [13] R. A. Novelline, Squire's Fundamentals of Radiology, 6th ed. Cambridge, UK, Harvard University Press, 2004.
- M. Mahesh, "Fluoroscopy: Patient Radiation Exposure Issues," RadioGraphics, vol. 21, no. 4, pp. 1033–1045, Jul. 2001. DOI: 10.1148/radiographics. 21.4.g01j1271033.
- [15] Z. M. Hijazi, K. Shivkumar, and D. J. Sahn, "Intracardiac Echocardiography (ICE) During Interventional & Electrophysiological Cardiac Catheterization," *Circulation*, vol. 119, no. 4, pp. 587–596, Feb. 2009. DOI: 10.1161/ CIRCULATIONAHA.107.753046.
- [16] P. Nordbeck, I. Weiss, P. Ehses, et al., "Measuring RF-induced currents inside implants: Impact of device configuration on MRI safety of cardiac pacemaker leads," *Magnetic Resonance in Medicine*, vol. 61, no. 3, pp. 570–578, Jan. 2009. DOI: https://doi.org/10.1002/mrm.21881.
- [17] I. Edler and K. Lindström, "The history of echocardiography," Ultrasound in Medicine & Biology, vol. 30, no. 12, pp. 1565–1644, Dec. 2004. DOI: https: //doi.org/10.1016/S0301-5629(99)00056-3.
- [18] S. M. Bierig and A. Jones, "Accuracy and Cost Comparison of Ultrasound Versus Alternative Imaging Modalities, Including CT, MR, PET, and Angiography," *Journal of Diagnostic Medical Sonography*, vol. 25, no. 3, pp. 138– 144, May 2009. DOI: 10.1177/8756479309336240.

- [19] G. S. Hillis and P. Bloomfield, "Basic Transthoracic Echocardiography," British Medical Journal, vol. 330, no. 7505, pp. 1432–1436, Jun. 2005.
- [20] T. Szabo, Diagnostic ultrasound imaging: inside out, 2nd ed. Oxford, UK: Academic Press, 2014.
- [21] S. S. Kim, Z. M. Hijazi, R. M. Lang, and B. P. Knight, "The Use of Intracardiac Echocardiography and Other Intracardiac Imaging Tools to Guide Noncoronary Cardiac Interventions," *Journal of the American College of Cardiology*, vol. 53, no. 23, pp. 2117–2128, Jun. 2009. DOI: https://doi. org/10.1016/j.jacc.2009.01.071.
- [22] J. Mathew, M. Swaminathan, and C. Ayoub, *Clinical Manual and Review of Transesophageal Echocardiography*, 3rd ed. New York, USA, McGraw-Hill Education, 2019.
- [23] O. Khalid, R. Srivastava, A. Mulhall, A. Paladugu, M. Stoddard, and S. Lippmann, "Conscious Sedation: For a TEE, Is It Always Required?" *Echocar-diography*, vol. 27, no. 1, pp. 74–76, Feb. 2010. DOI: https://doi.org/10.1111/j.1540-8175.2009.01049.x.
- [24] T. Bartel, S. Müller, A. Biviano, and R. T. Hahn, "Why is intracardiac echocardiography helpful? Benefits, costs, and how to learn," *European Heart Journal*, vol. 35, no. 2, pp. 69–76, Oct. 2013. DOI: 10.1093/eurheartj/ eht411.
- [25] T. Bartel, T. Konorza, J. Arjumand, et al., "Intracardiac Echocardiography Is Superior to Conventional Monitoring for Guiding Device Closure of Interatrial Communications," *Circulation*, vol. 107, no. 6, pp. 795–797, Feb. 2003. DOI: https://doi.org/10.1161/01.CIR.0000057547.00909.1C.
- [26] "Comparison of intracardiac echocardiography and transesophageal echocardiography for imaging of the right and left atrial appendages," *Heart Rhythm*, vol. 11, no. 11, pp. 1890–1897, Nov. 2014. DOI: https://doi.org/10.1016/ j.hrthm.2014.07.015.
- [27] D. Wildes, W. Lee, B. Haider, et al., "4-D ICE: A 2-D Array Transducer With Integrated ASIC in a 10-Fr Catheter for Real-Time 3-D Intracardiac Echocardiography," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 63, no. 12, pp. 2159–2173, Oct. 2016. DOI: 10.1109/ TUFFC.2016.2615602.

- [28] K. Iniewski, Medical Imaging Principles, Detectors, and Electronics, 1st ed. Hoboken, USA, John Wiley & Sons, Inc., 2009.
- [29] B. T. Khuri-Yakub and Ö. Oralkan, "Capacitive micromachined ultrasonic transducers for medical imaging and therapy," *Journal of Micromechanics* and Microengineering, vol. 21, no. 5, pp. 054004–054014, Apr. 2011. DOI: 10.1088/0960-1317/21/5/054004.
- [30] P. Muralt, N. Ledermann, J. Paborowski, et al., "Piezoelectric micromachined ultrasonic transducers based on PZT thin films," *IEEE Transactions on Ul*trasonics, Ferroelectrics, and Frequency Control, vol. 52, no. 12, pp. 2276– 2288, Dec. 2005. DOI: 10.1109/TUFFC.2005.1563270.
- [31] J. M. Rothberg, T. S. Ralston, A. G. Rothberg, et al., "Ultrasound-on-chip platform for medical imaging, analysis, and collective intelligence," Proceedings of the National Academy of Sciences, vol. 118, no. 27, Jul. 2021. DOI: 10.1073/pnas.2019339118.
- [32] J. Lee, K.-R. Lee, B. E. Eovino, et al., "A 36-Channel Auto-Calibrated Front-End ASIC for a pMUT-Based Miniaturized 3-D Ultrasound System," *IEEE Journal of Solid-State Circuits*, vol. 56, no. 6, pp. 1910–1923, Jan. 2021. DOI: 10.1109/JSSC.2021.3049560.
- [33] B. Steinberg, "Digital beamforming in ultrasound," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 39, no. 6, pp. 716–721, Nov. 1992. DOI: 10.1109/58.165556.
- [34] T. Proulx, D. Tasker, and J. Bartlett-Roberto, "Advances in catheter-based ultrasound imaging Intracardiac Echocardiography and the ACUSON Acu-NavTM Ultrasound Catheter," in *IEEE Ultrasonics Symposium*, 2005., vol. 1, Mar. 2005, pp. 669–678. DOI: 10.1109/ULTSYM.2005.1602941.
- [35] T. Matéo, N. Sénégond, C. Meynier, et al., "A 1-D CMUT transducer with front-end ASIC in a 9 French catheter for Intracardiac Echocardiography: Acoustic and Imaging evaluation," in 2020 IEEE International Ultrasonics Symposium (IUS), Nov. 2020, pp. 1–8. DOI: 10.1109/IUS46767.2020. 9251715.
- [36] C. Oakley, "Calculation of ultrasonic transducer signal-to-noise ratios using the KLM model," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 44, no. 5, pp. 1018–1026, Sep. 1997. DOI: 10.1109/ 58.655627.

- [37] W. Lee, S. Idriss, P. Wolf, and S. Smith, "A miniaturized catheter 2-D array for real-time, 3-D intracardiac echocardiography," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 51, no. 10, pp. 1334– 1346, Oct. 2004. DOI: 10.1109/TUFFC.2004.1350962.
- [38] D. E. Dausch, K. H. Gilchrist, J. B. Carlson, S. D. Hall, J. B. Castellucci, and O. T. von Ramm, "In vivo real-time 3-D intracardiac echo using PMUT arrays," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 61, no. 10, pp. 1754–1764, Oct. 2014. DOI: 10.1109/TUFFC. 2014.006452.
- [39] M. Tan, E. Kang, J.-S. An, et al., "A 64-Channel Transmit Beamformer With ±30-V Bipolar High-Voltage Pulsers for Catheter-Based Ultrasound Probes," *IEEE Journal of Solid-State Circuits*, vol. 55, no. 7, pp. 1796–1806, Apr. 2020. DOI: 10.1109/JSSC.2020.2987719.
- [40] G. Gurun, P. Hasler, and F. L. Degertekin, "Front-end receiver electronics for high-frequency monolithic CMUT-on-CMOS imaging arrays," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 58, no. 8, pp. 1658–1668, Aug. 2011. DOI: 10.1109/TUFFC.2011.1993.
- [41] K. Chen, H.-S. Lee, A. P. Chandrakasan, and C. G. Sodini, "Ultrasonic Imaging Transceiver Design for CMUT: A Three-Level 30-Vpp Pulse-Shaping Pulser With Improved Efficiency and a Noise-Optimized Receiver," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 11, pp. 2734–2745, Nov. 2013. DOI: 10.1109/JSSC.2013.2274895.
- [42] G. Gurun, C. Tekes, J. Zahorian, et al., "Single-chip CMUT-on-CMOS frontend system for real-time volumetric IVUS and ICE imaging," *IEEE Trans*actions on Ultrasonics, Ferroelectrics, and Frequency Control, vol. 61, no. 2, pp. 239–250, Feb. 2014. DOI: 10.1109/TUFFC.2014.6722610.
- [43] E. Kang, Q. Ding, M. Shabanimotlagh, et al., "A Reconfigurable Ultrasound Transceiver ASIC With 24 × 40 Elements for 3-D Carotid Artery Imaging," *IEEE Journal of Solid-State Circuits*, vol. 53, no. 7, pp. 2065–2075, Jul. 2018. DOI: 10.1109/JSSC.2018.2820156.
- [44] M. Tan, C. Chen, Z. Chen, et al., "A Front-End ASIC With High-Voltage Transmit Switching and Receive Digitization for 3-D Forward-Looking Intravascular Ultrasound Imaging," *IEEE Journal of Solid-State Circuits*, vol. 53, no. 8, pp. 2284–2297, Aug. 2018. DOI: 10.1109/JSSC.2018.2828826.

- [45] C. Chen and M. A. P. Pertijs, "Integrated Transceivers for Emerging Medical Ultrasound Imaging Devices: A Review," *IEEE Open Journal of the Solid-State Circuits Society*, vol. 1, pp. 104–114, Sep. 2021. DOI: 10.1109/0JSSCS. 2021.3115398.
- [46] E. Kang, M. Tan, J.-S. An, et al., "A Variable-Gain Low-Noise Transimpedance Amplifier for Miniature Ultrasound Probes," *IEEE Journal of Solid-State Circuits*, vol. 55, no. 12, pp. 3157–3168, Dec. 2020. DOI: 10.1109/JSSC. 2020.3023618.
- [47] G. Pilikos, L. Horchens, J. Batenburg, T. van Leeuwen, and F. Lucka, "Deep data compression for approximate ultrasonic image formation," in 2020 IEEE International Ultrasonics Symposium, Sep. 2020. DOI: 10.1109/IUS46767. 2020.9251753.
- [48] N. Shlezinger, A. Amar, B. Luijten, R. J. G. van Sloun, and Y. C. Eldar, Deep Task-Based Analog-to-Digital Conversion, Jan. 2022. DOI: 10.48550/ ARXIV.2201.12634.
- [49] D. Van Willigen, J. Janjic, E. Kang, et al., "ASIC Design for a Single-Cable 64-Element Ultrasound Probe," in 2018 IEEE International Ultrasonics Symposium (IUS), Feb. 2018, pp. 1–4. DOI: 10.1109/ULTSYM.2018.8580162.
- [50] D. M. van Willigen, E. Kang, J. Janjic, et al., "A Transceiver ASIC for a Single-Cable 64-Element Intra-Vascular Ultrasound Probe," *IEEE Journal* of Solid-State Circuits, vol. 56, no. 10, pp. 3157–3166, Jun. 2021. DOI: 10. 1109/JSSC.2021.3083217.
- [51] J. Kang, C. Yoon, J. Lee, et al., "A System-on-Chip Solution for Pointof-Care Ultrasound Imaging Systems: Architecture and ASIC Implementation," *IEEE Transactions on Biomedical Circuits and Systems*, vol. 10, no. 2, pp. 412–423, Apr. 2016. DOI: 10.1109/TBCAS.2015.2431272.
- [52] Y. Igarashi, S. Kajiyama, Y. Katsube, et al., "Single-Chip 3072-Element-Channel Transceiver/128-Subarray-Channel 2-D Array IC With Analog RX and All-Digital TX Beamformer for Echocardiography," *IEEE Journal of Solid-State Circuits*, vol. 54, no. 9, pp. 2555–2567, Sep. 2019. DOI: 10.1109/ JSSC.2019.2921697.
- [53] G. Jung, C. Tekes, M. W. Rashid, et al., "A Reduced-Wire ICE Catheter ASIC With Tx Beamforming and Rx Time-Division Multiplexing," *IEEE Transactions on Biomedical Circuits and Systems*, vol. 12, no. 6, pp. 1246–1255, Dec. 2018. DOI: 10.1109/TECAS.2018.2881909.

- [54] K. Chen, B. C. Lee, K. E. Thomenius, B. T. Khuri-Yakub, H.-S. Lee, and C. G. Sodini, "A Column-Row-Parallel Ultrasound Imaging Architecture for 3-D Plane-Wave Imaging and Tx Second-Order Harmonic Distortion Reduction," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 65, no. 5, pp. 828–843, May 2018. DOI: 10.1109/TUFFC.2018. 2811393.
- [55] P. Guo, F. Fool, E. Noothout, et al., "A 1.2mW/channel 100µm-Pitch-Matched Transceiver ASIC with Boxcar-Integration-Based RX Micro-Beamformer for High-Resolution 3D Ultrasound Imaging," in 2022 IEEE International Solid-State Circuits Conference (ISSCC), vol. 65, Feb. 2022, pp. 496–498. DOI: 10.1109/ISSCC42614.2022.9731784.
- [56] C. Daft, P. Wagner, S. Panda, and I. Ladabaum, "Two Approaches to Electronically Scanned 3D Imaging Using cMUTs," in 2006 IEEE Ultrasonics Symposium, Oct. 2006, pp. 685–688. DOI: 10.1109/ULTSYM.2006.185.
- [57] T. M. Carpenter, M. W. Rashid, M. Ghovanloo, D. M. J. Cowell, S. Freear, and F. L. Degertekin, "Direct Digital Demultiplexing of Analog TDM Signals for Cable Reduction in Ultrasound Imaging Catheters," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 63, no. 8, pp. 1078–1085, Aug. 2016. DOI: 10.1109/TUFFC.2016.2557622.
- [58] M. W. Rashid, C. Tekes, M. Ghovanloo, and F. L. Degertekin, "Design of frequency-division multiplexing front-end receiver electronics for CMUT-on-CMOS based intracardiac echocardiography," in 2014 IEEE International Ultrasonics Symposium, Sep. 2014, pp. 1540–1543. DOI: 10.1109/ULTSYM. 2014.0381.
- [59] Q. Liu, C. Chen, Z.-y. Chang, C. Prins, and M. A. P. Pertijs, "A mixed-signal multiplexing system for cable-count reduction in ultrasound probes," in 2015 *IEEE International Ultrasonics Symposium (IUS)*, Oct. 2015, pp. 1–4. DOI: 10.1109/ULTSYM.2015.0141.
- [60] Y. M. Hopf, O. Boudewine, S. Mehdi, et al., "A Pitch-Matched Transceiver ASIC with Shared Hybrid Beamforming ADC for High-Frame-Rate 3D Intracardiac Echocardiography," *IEEE Journal of Solid-State Circuits*, vol. 57, no. 11, pp. 3228–3242, Nov. 2022. DOI: 10.1109/JSSC.2022.3201758.
- B. Savord and R. Solomon, "Fully sampled matrix transducer for real time 3D ultrasonic imaging," in *IEEE Symposium on Ultrasonics*, 2003, Oct. 2003, pp. 945–953. DOI: 10.1109/ULTSYM.2003.1293556.

- [62] M.-C. Chen, A. Peña Perez, S.-R. Kothapalli, et al., "A Pixel Pitch-Matched Ultrasound Receiver for 3-D Photoacoustic Imaging With Integrated Delta-Sigma Beamformer in 28-nm UTBB FD-SOI," *IEEE Journal of Solid-State Circuits*, vol. 52, no. 11, pp. 2843–2856, Nov. 2017. DOI: 10.1109/JSSC. 2017.2749425.
- [63] Y.-J. Kim, S.-E. Cho, J.-Y. Um, et al., "A Single-Chip 64-Channel Ultrasound RX-Beamformer Including Analog Front-End and an LUT for Non-Uniform ADC-Sample-Clock Generation," *IEEE Transactions on Biomedical Circuits and Systems*, vol. 11, no. 1, pp. 87–97, Feb. 2017. DOI: 10.1109/ TBCAS.2016.2571739.
- [64] M. D'Urbino, C. Chen, Z. Chen, et al., "An Element-Matched Electromechanical ΔΣ ADC for Ultrasound Imaging," *IEEE Journal of Solid-State Circuits*, vol. 53, no. 10, pp. 2795–2805, Oct. 2018. DOI: 10.1109/JSSC. 2018.2859961.
- [65] C. Chen, Z. Chen, D. Bera, et al., "A Pitch-Matched Front-End ASIC With Integrated Subarray Beamforming ADC for Miniature 3-D Ultrasound Probes," *IEEE Journal of Solid-State Circuits*, vol. 53, no. 11, pp. 3050–3064, Sep. 2018. DOI: 10.1109/JSSC.2018.2864295.
- [66] J.-Y. Um, Y.-J. Kim, S.-E. Cho, et al., "An Analog-Digital Hybrid RX Beamformer Chip With Non-Uniform Sampling for Ultrasound Medical Imaging With 2D CMUT Array," *IEEE Transactions on Biomedical Circuits and* Systems, vol. 8, no. 6, pp. 799–809, Dec. 2014. DOI: 10.1109/TBCAS.2014. 2375958.
- [67] T. Kim, S. Shin, and S. Kim, "An 80.2 dB DR 23.25 mW/Channel 8-Channel Ultrasound Receiver With a Beamforming Embedded SAR ADC," *IEEE Transactions on Circuits and Systems II: Express Briefs*, vol. 66, no. 9, pp. 1487–1491, Sep. 2019. DOI: 10.1109/TCSII.2018.2889810.
- [68] T. Cieszynski, "Intracardiac method for the investigation of structure of the heart with the aid of ultrasonics," Archivum immunologiae et therapiae experimentalis, vol. 8, pp. 551–7, 1960.
- [69] N. Bom, H. ten Hoff, C. T. Lancée, W. J. Gussenhoven, and J. G. Bosch, "Early and recent intraluminal ultrasound devices," *The International Journal of Cardiac Imaging*, vol. 4, no. 4, pp. 79–88, Jun. 1989.

- [70] N. Bom, C. Lancée, and F. Van Egmond, "An ultrasonic intracardiac scanner," Ultrasonics, vol. 10, no. 2, pp. 72–76, Mar. 1972. DOI: https://doi. org/10.1016/0041-624X(72)90250-8.
- [71] J. B. Seward, D. L. Packer, R. C. Chan, M. Curley, and A. J. Tajik, "Ultrasound Cardioscopy: Embarking on a New Journey," *Mayo Clinic Proceedings*, vol. 71, no. 7, pp. 629–635, Jul. 1996. DOI: https://doi.org/10.1016/ S0025-6196(11)62999-6.
- [72] I. Gonzalez, Q.-L. Cao, and Z. M. Hijazi, Role of intracardiac echocardiography (ICE) in transcatheter occlusion of atrial septal defects. London, UK, IntechOpen, 2012, pp. 99–118.
- [73] D. Packer, S. Johnson, M. Kolasa, T. Bunch, B. Henz, and Y. Okumura, "New generation of electro-anatomic mapping: full intracardiac ultrasound image integration," *Europace*, vol. 10, pp. iii35–41, Dec. 2008. DOI: 10.1093/ europace/eun231.
- [74] W. Lee, W. Griffin, D. Wildes, et al., "A 10-Fr ultrasound catheter with integrated micromotor for4-D intracardiac echocardiography," *IEEE Trans*actions on Ultrasonics, Ferroelectrics, and Frequency Control, vol. 58, no. 7, pp. 1478–1491, Jul. 2011. DOI: 10.1109/TUFFC.2011.1967.
- [75] W. T. Wilser, S. R. Barnes, and L. Garbini, "Helical acoustic array for medical ultrasound," U.S. Patent US8449467B2, May 2013.
- [76] J. W. Choe, O. Oralkan, A. Nikoozadeh, et al., "Volumetric real-time imaging using a CMUT ring array," *IEEE Transactions on Ultrasonics, Ferroelectrics,* and Frequency Control, vol. 59, no. 6, pp. 1201–1211, Jun. 2012. DOI: 10. 1109/TUFFC.2012.2310.
- [77] E. D. Light, S. F. Idriss, P. D. Wolf, and S. W. Smith, "Real-time threedimensional intracardiac echocardiography," *Ultrasound in Medicine & Biology*, vol. 27, no. 9, pp. 1177–1183, Sep. 2001. DOI: https://doi.org/10. 1016/S0301-5629(01)00421-5.
- [78] Marketing Clearance of Diagnostic Ultrasound Systems and Transducers, U.S. Department of Health, Human Services, Food, and Drug Administration, Silver Spring, MD, USA, Jun. 2019.
- [79] IEC 60601-2-37/A1, International Electrotechnical Commission, Geneva, Switzerland, Aug. 2015.

- [80] C. Tangwongsan, J. Will, J. Webster, K. Meredith, and D. Mahvi, "In vivo measurement of swine endocardial convective heat transfer coefficient," *IEEE Transactions on Biomedical Engineering*, vol. 51, no. 8, pp. 1478–1486, Aug. 2004. DOI: 10.1109/TBME.2004.828035.
- [81] C. Chen, Z. Chen, D. Bera, et al., "A Front-End ASIC With Receive Subarray Beamforming Integrated With a 32×32 PZT Matrix Transducer for 3-D Transesophageal Echocardiography," *IEEE Journal of Solid-State Circuits*, vol. 52, no. 4, pp. 994–1006, Apr. 2017. DOI: 10.1109/JSSC.2016.2638433.
- [82] J. Li, Z. Chen, M. Tan, et al., "A 1.54mW/Element 150µm-Pitch-Matched Receiver ASIC with Element-Level SAR/Shared-Single-Slope Hybrid ADCs for Miniature 3D Ultrasound Probes," in 2019 Symposium on VLSI Circuits, Aug. 2019, pp. C220–C221. DOI: 10.23919/VLSIC.2019.8778200.
- [83] P. Santos, G. U. Haugen, L. Løvstakken, E. Samset, and J. D'hooge, "Diverging Wave Volumetric Imaging Using Subaperture Beamforming," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 63, no. 12, pp. 2114–2124, Dec. 2016. DOI: 10.1109/TUFFC.2016.2616172.
- [84] E. E. Konofagou and J. Provost, "Electromechanical wave imaging for noninvasive mapping of the 3D electrical activation sequence in canines and humans in vivo," *Journal of Biomechanics*, vol. 45, no. 5, pp. 856–864, Mar. 2012, Special Issue on Cardiovascular Solid Mechanics. DOI: https: //doi.org/10.1016/j.jbiomech.2011.11.027.
- [85] O. Villemain, J. Baranger, M. K. Friedberg, et al., "Ultrafast Ultrasound Imaging in Pediatric and Adult Cardiology: Techniques, Applications, and Perspectives," JACC: Cardiovascular Imaging, vol. 13, no. 8, pp. 1771–1791, Aug. 2008. DOI: https://doi.org/10.1016/j.jcmg.2019.09.019.

# IMAGING SCHEME FOR 3D HIGH-FRAME-RATE ICE

This chapter is based on the publication "Imaging Scheme for 3-D High Frame Rate Intracardiac Echography: A Simulation Study" in IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control, vol. 69, no. 10, pp. 2862-2874, Oct. 2022.

## 2.1. Introduction

#### 2.1.1. Clinical Background

 $\mathbf{T}$  he heart pumps blood with a rhythm determined by a group of pacemaking cells in the sinoatrial node. These cells generate an action potential that depolarizes cardiomyocytes. This action potential triggers the adjacent cells and propagates along the cardiac wall to generate global contraction. The conduction properties of cardiomyocytes and electrical pathways can be changed by normal aging or certain diseases which lead to various types of arrhythmias. Atrial fibrillation (AF) is the most common cardiac arrhythmia and is the major cardiac precursor of stroke [1]. AF can be treated by RF ablation. A high-resolution 3D anatomical mapping of electrical activity is required for localizing sources generating AF to
plan the ablation procedure [2], [3]. A key feature of atrial fibrillation is its irregularity in time and space. Because the patterns of atrial activation change on a beat-to-beat basis, a series of consecutive activation maps are needed to cover the spatiotemporal variation in activation [4]. Intracardiac electro-anatomical mapping is widely exploited to generate a 3D surface map of atrial electrical activity [5].

#### 2.1.2. Electromechanical Wave Imaging

Electrical excitation of cardiomyocytes is followed by a transient contraction after about 20-40 ms [6]. The electrical excitation is passed to neighboring cells and propagates along the atrial wall as a reaction-diffusion wave with a velocity of 0.5-2 m/s [7]. A high correlation between electrical and resulting mechanical activations has been reported in several studies [8]–[11] and this activation propagates as a wave-like phenomenon over the atrial wall: the electromechanical wave (EW). Electromechanical wave imaging (EWI) was introduced and has been developed by the group of Konofagou [7], [12]–[19] as a non-invasive ultrasound-based imaging method to map the electro-mechanical activations in the heart. Recent studies demonstrated a high correlation between the cardiac electrical activity and its consequent electromechanical wave for healthy and arrhythmic cases in a simulation study [16], in the left ventricular wall [17], [18], atrial wall [15], and also in the walls of all four chambers [20].

Electromechanical wave imaging is accomplished in a two-stage process on ultrasound data [12]–[15]. In the first stage, tissue strain is estimated, commonly by a cross-correlation technique, along the axial direction on consecutive RF signals which are acquired at a motion-estimation rate (500 - 2000 Hz) [19]. In the second stage, electromechanical wave onset is estimated by measuring the time at which the axial incremental strain departs from or crosses zero at a given pixel. We intend to image the propagation of the EW wavefront with a spatiotemporal resolution suitable for the identification of local electrophysiological phenomena. Since the EW velocities are in the order of 0.5 - 2 m/s and the desired spatial resolution is in the order of millimeters, we aim at a volume rate of 1,000 Hz and a lateral resolution of 2 to 5 mm.

Current electromechanical wave imaging can provide single-beat 2D maps of the electromechanical wavefront (i.e. a cross-sectional image with activation times) with a lateral resolution of < 5 mm. Since the electromechanical waves propagate throughout the entire heart, volumetric imaging is desirable to fully visualize their patterns. In previous studies, several 2D images were acquired and processed separately. Then, a pseudo-3D map was generated by combining these 2D maps through electrocardiogram (ECG) gating [20] and spatial interpolation. Recently, Grondin *et al.* used a transthoracic matrix transducer to achieve 3D EWI of the entire heart from the apical view [21]. The narrow intercostal space and channel count limitations restricted their transducer size to 9.6 mm × 9.6 mm. In consequence, the lateral resolution was limited, to 5.8 mm at 40 mm. For the atria which are located much deeper (> 10 cm) the lateral resolution will be lower. It would be interesting to see if a higher resolution can improve strain mapping and reduce local inaccuracies in the electromechanical wave patterns in the atria.

#### 2.1.3. High-Frame-Rate 3D Intracardiac Echocardiography

Using an intracardiac echocardiography (ICE) transducer can drastically reduce the required imaging depth for atrial EWI compared to the transthoracic apical view. Moreover, the possibility of using a higher central frequency compared to transthoracic transducers allows a higher spatial resolution in the atrial regions. Nowadays, ICE is widely exploited for RF ablation procedures to guide electrophysiologists in septal puncture and to navigate the ablation catheter. However, this is limited to low-frame-rate 2D-ICE. Several 3D-ICE designs have been proposed [22]–[24], but these are not capable of imaging at high frame rates. A high-frame-rate 3D ICE probe, in addition to delivering at least the functionality of conventional 2D ICE, could provide a single-beat 3D electromechanical wave map. However, high-frame-rate 3D ultrasound imaging introduces several challenges, including handling a high data rate generated by a large number of elements. Moreover, a high volume rate allows just a few transmissions to illuminate the total volume, so a diverging wave transmission scheme is required. Possibilities for compounding are limited, and thus SNR will be relatively low [25]. Furthermore, an intracardiac catheter imposes additional constraints on the transducer size, cable count, and power dissipation. ICE catheters typically have diameters in the order of 3 mm [22] and need to accommodate the entire imaging matrix as well as associated electronics at the tip as well as all interconnect in the shaft. Therefore, data reduction and an educated choice of design tradeoffs are crucial for high-frame-rate 3D ICE.

#### 2.1.4. Data Rate Reduction by Micro-Beamforming

Micro-beamforming is a well-established method to reduce the channel count while preserving the image quality and SNR for volumetric imaging [26], [27]. In this method, element signals in a sub-aperture are mutually delayed and summed to form a single micro-beamformed signal from a predefined –steered– direction. The entire region of interest can thus be imaged by steering the sub-apertures to a sequence of different directions in, potentially, both azimuth and elevation, and implementing a parallel beamforming technique at the back-end system [28]–[30]. Since the final beamforming is based on the received data from sub-apertures with a relatively large effective pitch, grating lobes can occur. Using relatively narrow transmit beams that are steered to the pre-steering directions can reduce the grating lobe artifacts. Hence, a significant number of such narrow beams should be utilized to preserve the image quality [26]. Thus, increasing the number of elements per sub-aperture reduces the channel count but also reduces the achievable volume rate or field of view.

Wildes *et al.* developed a 2D transducer consisting of  $60 \times 14$  elements with application-specific integrated electronics for volumetric intracardiac echocardiography based on a 10-Fr (3.3 mm) catheter [22]. A 2D micro-beamforming approach was used to reduce the total receive channel count to 48 (along with power, transmit, and auxiliary signals leading to a total of 88 connections). Their method provided an imaging sector of  $90^{\circ} \times 60^{\circ} \times 8$  cm with a spatial resolution of 1.7 mm × 4.9 mm (azimuth × elevation) at a depth of 50 mm. They concluded that the achieved spatial resolution was sufficient for atrial imaging. To suppress the grating lobes formed by micro-beamforming, narrow beams were used in transmission, which limited the frame rate to 30 vol/s [22].

Implementing a 1D (i.e. a 1xN sub-aperture) rather than a 2D micro-beamformer introduces the grating lobes only in the direction in which the micro-beamforming is implemented. Therefore, the transmit beams need to be narrow only in the corresponding direction. Consequently, 1D micro-beamforming could provide a relatively high volume rate 3D imaging by employing a set of fan-shaped diverging transmit beams that are narrow in the micro-beamforming direction and wide in the perpendicular direction, see Fig. 2.1. These asymmetric fan-shaped beams can cover a large volume of interest with a limited number of transmissions. These fan-shaped beams can be generated by a dual virtual source similar to the method proposed by Chen *et al.* [31] or a single virtual source behind a rectangular transducer as illustrated in Fig. 2.1(b). Since a dual virtual source approach generates



Figure 2.1: (a) A schematic representation of the proposed side-looking transducer mounted in an intracardiac catheter. Azimuth (Az), elevation (El), and axial (Ax) directions are indicated on the corresponding axes. A set of 3 out of 7 fan-shaped beams steered to different directions in elevation is shown to illustrate the transmit scheme. (b) A fan-shaped transmit beam is generated from a virtual source behind the transducer.

a curved transmit fan beam, it is less appropriate to be combined with a microbeamforming approach. Therefore, we use a single virtual source approach in this study.

The size of the 1D micro-beamformer should be chosen such that the needed reduction in channel count is achieved while grating lobe levels are kept sufficiently low. In the design of the high-frame-rate 3D ICE, many trade-offs need to be made to realize an imaging scheme that satisfies the requirements for EWI while remaining within the constraints on transducer size and cable count, resulting from the small catheter size. Furthermore, the elements will be mounted on a pitch-matched application-specific integrated circuit (ASIC) containing all the required electronics, including high-voltage pulsers, low-noise amplifiers, time-gain compensation circuits, analog-to-digital converters (ADCs), and micro-beamformers, which are necessary to realize the proposed method [27], [32], [33]. The minimum pitch of the transducer elements is limited to 160  $\mu$ m × 160  $\mu$ m by the area required for the designed electronics.

In this study, we did not consider electronic noise and ultrasonic attenuation in the simulations, since we are assuming they are not limiting factors on the clutter and PSF image quality parameters we are studying here. They have been considered in the electronics design procedure [33] to achieve the desired imaging depth. Although these circuit design choices put some limits on the transducer layout and the imaging scheme as mentioned in the introduction and methods section, fully describing these choices is out of the scope of this manuscript.  $\mathbf{2}$ 

#### 2.1.5. 3D ICE Imaging Scheme Design

In this study, we develop a novel imaging scheme for high-frame-rate 3D ultrasound imaging with a matrix intracardiac catheter as sketched in Section 2.1.4, and gear the transducer design and imaging scheme towards intracardiac electromechanical wave imaging. The main goal is to enable volumetric imaging with a sufficient frame rate and image quality for electromechanical wave imaging while reducing the data rate to realistic values. The proposed method consists of implementing 1D micro-beamforming in the elevation direction to reduce the data rate. By combining this with the transmission of fan-shaped beams, a high frame rate is enabled while grating lobes in the elevation direction are suppressed. The proposed method achieves a further reduction of the grating lobe artifacts and improved CNR by implementing angular-weighted coherent compounding. We will discuss the decision on critical imaging scheme parameters (transmit frequency, microbeamformer size, and transmit beam divergence) to achieve the desired channel count reduction while providing a sufficient frame rate and image quality for EWI. Finally, we evaluate the proposed imaging scheme in a series of simulations. The novelty of this work lies in an imaging scheme designed for dedicated integrated circuitry to realize high-frame-rate 3D ICE suitable for EWI.

# 2.2. Materials and Methods

#### 2.2.1. Imaging Scheme Design: Parameters

We strive for a 3D ICE design with a frame rate of 1000 Hz, imaging depth of 10 cm, opening angle of  $70^{\circ}$  by  $70^{\circ}$ , and lateral resolution of 5 mm to be able to realize 3D EWI of the left atrium from the right atrium.

For a depth of 10 cm, the pulse repetition frequency (PRF) is limited to 7.7 kHz by the round-trip travel time of the ultrasound waves assuming a speed of sound of 1540 m/s. Therefore, a maximum number of 7 transmissions is available to acquire the entire region of interest, and we will always use this number of transmissions in our scheme.

We base the design of the matrix transducer on a 10-Fr intracardiac catheter, which limits the transducer size to 3 mm in elevation direction (Fig. 2.1(a) and the number of cables to a maximum of 100 [22], [34]. The aperture of the matrix array is rectangular, with the shortest axis perpendicular to the ICE probe shaft.

We consider square elements with a pitch of 160  $\mu$ m, imposed by the area needed for the pitch-matched application-specific electronics [32]. Choosing such a large pitch (50% to 80% of the wavelength for the assumed frequency range of 5 to 8 MHz) introduces grating lobes in image reconstruction and also secondary waves in diverging wave transmission. To counteract these, we optimize the central frequency in the given range to suppress the secondary waves. In addition, an angular weighting function is applied to a conventional delay-and-sum to reduce the grating lobes. Based on the chosen pitch and catheter size constraints we consider a matrix array of 18×64 elements.

The imaging scheme consisting of steered fan-shaped diverging transmit beams will be used with 1D micro-beamforming in elevation direction and acquisition of all element-level signals in the azimuthal direction. Furthermore, angular weighted coherent compounding will be used to reduce grating lobe levels. Transducer size, pitch, number of transmissions, and imaging depth are fixed in this study, whereas the other features are optimized in a simulation study.

#### 2.2.2. Steerable Fan-Shaped Transmit Beam

A single virtual source was used to generate the desired fan-shaped beam. In this approach, the beam divergence in azimuth and elevation are not independent. First, we determined the virtual source location based on the transducer size and desired beam divergence in the elevational direction, and then the effective transducer size in azimuth was determined to achieve the required beam divergence in that direction (see appendix I).

#### 2.2.3. Angular Weighted Averaging

A voxel-based delay-and-sum is proposed to reconstruct an intermediate full volumetric image corresponding to each transmission. Since only a narrow region is insonified in elevation direction for each transmission, reconstructing regions far from this transmission direction only adds noise to intermediate images. In addition, the receive beam profile contains strong grating lobes at directions corresponding to the large effective pitch associated with the micro-beamformer size [26]. Hence, the final volumetric image is obtained by applying angular weighting functions to the intermediate images in order to suppress the noise and grating



Figure 2.2: Applied weights  $W(\Theta_r,\Theta_i)$  as a function of the reconstruction elevation direction  $\Theta_r$  for different transmit beams. Blue arrows show all transmission directions and the red arrow shows the current transmission direction with its corresponding angular weighting function. Black arrows show the reconstruction directions and the bold black arrow shows an example of receive line with its corresponding weights (black cross) when the transmit direction is  $\Theta_i$ .

lobes in non-insonified regions and also improve the image quality by coherent compounding of overlapping regions using

$$I(R,\Theta_r,\varphi) = \sum_{i=1}^{7} W(\Theta_r,\Theta_i) \sum_{m=1}^{N_{\mu BF}} \sum_{n=1}^{N_{AZ}} S_{m,n}(t - \sigma_i - \sigma_{m,n})$$
(2.1)

where I(R,  $\Theta_r$ ,  $\varphi$ ) is a reconstructed RF sample in a spherical coordinate system, W( $\Theta_r$ , $\Theta_i$ ) is the weighting function (as illustrated in Fig. 2.2) corresponding to the transmit direction  $\Theta_i$ , S<sub>m, n</sub> is the micro-beamformed signal of a subgroup at elevation position m and azimuthal position n in the transducer array,  $\sigma_i(R, \Theta_r, \varphi)$ is transmit delay,  $\sigma_{m, n}(R, \Theta_r, \varphi)$  is receive delay and  $\Theta_i$  is transmit/pre-steering direction.

Fig. 2.2 shows the angular weights as a function of the reconstruction line direction for different transmission directions. To avoid abrupt changes in image intensity, the angular weights have some overlap with their neighboring transmissions [33].

#### 2.2.4. Simulation Setup

The simulation study consists of three steps. First, the azimuthal beam divergence and central frequency in a range of  $20^{\circ}$  -  $45^{\circ}$  and 5 - 8 MHz respectively are mutually optimized to provide an acceptable trade-off among the imaging opening angle,

| Parameters               | Value  |
|--------------------------|--|
| Transducer size          | 2.88 mm x 10.24 mm   |
| Pitch                    | 160 μm x 160 μm  |
| Kerf                     | 20 µm  |
| Number of elements       | 18 x 64  |
| Micro-beamformer size    | 1 - 4 elements *   |
| Imaging depth            | 10 cm  |
| Frame rate               | > 1000 Hz  |
| Transmit beam            | Steerable Fan beam   |
| Number of transmissions  | 7  |
| Transmission beam        | Direction: 0°  |
| (azimuth)                | Divergence: 20° - 45° *                                      |
| Transmission beam        | Direction: $-30^{\circ}$ to $30^{\circ}$ ; step $10^{\circ}$ |
| (elevation)              | Divergence: 10.7°, 12.5° - 20° *; step: 2.5°                 |
| Central frequency        | 5 - 8 MHz *  |
| Sampling frequency       | 4 x central frequency  |
| Transmit apodization     | 2D Tukey window with   |
| - Tunishint up suizution | cosine fraction 0.2  |

TABLE 2.1: IMAGING PARAMETERS

\* These parameters are evaluated within the given ranges to optimize the imaging scheme.

transmit beam quality, and spatial resolution. Transmit beam quality in diverging wave imaging is known to be susceptible to the occurrence of so-called axial lobes or secondary pulses [35], [36], related to imperfect signal cancellation of late-arriving element signals from arrays with pitch larger than half a wavelength. Since the quality of the transmitted wave is very important for our image quality, we will first investigate and optimize the temporal profile of the transmitted diverging wave in our simulations.

Second, transmit beam divergence in elevation will be determined based on grating lobe artifacts, and the intensity uniformity between transmissions. Finally, the image quality will be evaluated for a micro-beamformer size ranging between 1 (no micro-beamforming) and 4 to achieve an acceptable trade-off between the desired image quality and data rate reduction. The imaging parameters are summarized in Table 2.1. All simulations are performed in Field II [37].

 $\mathbf{2}$ 



Figure 2.3: Schematic cross-sections of the 3D left atrium numerical phantom (radius 50 mm, thickness 3 mm). Left: a wave is introduced as a radial wall thickening (blue arrows). Red arrows show the circumferential electromechanical wavefront propagation direction. Right: wavefront propagation after 10ms (thickness/excursions not to scale).

#### 2.2.5. Numerical Phantoms

Static phantom: A 3D numerical phantom, a cube with an edge length of 100 mm containing hyperechoic and hypoechoic spherical regions with diameters of 2, 4, 6, and 10 mm, highly reflective points, and background point scatterers with an average density of  $0.2/\lambda$  has been used in Field II simulations. The relatively low point scatterer density is chosen to limit simulation time for this large-size cubic phantom and the high number of transducer elements. In addition, a single point scatterer at a depth of 50 mm is simulated to determine the Point Spread Function (PSF).

Beating left atrium: To evaluate the proposed high-frame-rate imaging scheme in a dynamic environment, a spherical shell with an outer diameter of 50 mm and a thickness of 3 mm has been simulated to mimic a left atrium. The center of the sphere is at an axial distance of 50 mm, equivalent to imaging the left atrium from the center of the right atrium. A circumferentially propagating wave has been introduced to this phantom by applying a radial thickening with an amplitude of 10 mm/s (10  $\mu$ m/frame). The atrial thickening initiates from a single point and propagates omnidirectionally along the atrial wall with a velocity of 2 m/s [7]. A cross-section of the beating left atrium numerical phantom is illustrated in Fig. 2.3.

We use 1D axial cross-correlation to estimate the frame-to-frame tissue displacement. Since simulating a 3D numerical phantom with a matrix array in Field II is very time-consuming, we limit our simulations to only 3 pairs of consecutive volumetric data sets acquired at an interval of 1 ms within each pair (i.e., mimicking a frame rate of 1000 Hz). The consecutive pairs of data sets have been simulated for 3 time-points, t = 0 ms, 10 ms, and 20 ms, to show the concept of electromechanical wave imaging. The final map of tissue displacement (actually displacement over 1ms in the direction of the ultrasound beam, so the axial tissue velocity component) has been generated by applying an image intensity-based mask to the displacement data to only visualize the atrial wall. In addition, another mask based on tissue displacement has been applied to the displacement map to exclude non-moving tissue.

#### 2.2.6. Evaluation Criteria

The performance of various combinations of transmission and micro-beamforming schemes has been evaluated, as assessed by the widths of PSF at -6 dB (azimuthal and elevational resolutions), the side lobe level, the grating lobe level, and the contrast-to-noise ratio (CNR). The side lobe level is defined as the peak level of the highest side lobe compared to the main lobe in dB, and the grating lobe level is defined as the peak level at the theoretically-expected grating lobe direction. These two were measured by imaging only one scatterer in the field of view. The CNR is calculated using [38]

$$CNR = \frac{\mu_s - \mu_c}{\sqrt{\frac{\sigma_s^2 + \sigma_c^2}{2}}} \tag{2.2}$$

where  $\mu_s$  and  $\mu_c$  are the mean amplitudes and  $\sigma_s$  and  $\sigma_c$  are the standard deviation of the gray levels in the speckle and anechoic cyst regions, respectively, of the 3D numerical phantom.

Imaging with a limited number of diverging beams and micro-beamforming may result in angle-dependent changes in image characteristics. To evaluate these effects, a lateral shift-variance plot (LSV-plot) [35] is used. An LSV-plot is constructed for a given imaging system by imaging a point scatterer shifting laterally. For each lateral position, a PSF is calculated and stacked to the other PSFs to form a 2D image. The entire image is normalized to its maximum value and plotted in dB scale. If there is little position dependence, the plot will be invariant along the diagonal. Otherwise, deviating structures will be visible revealing the nature of the position dependence. In this study, a point scatterer at a distance of 50 mm from the center of the transducer is shifted along the elevation direction from -45° to 45° with a step of 1° to create the LSV-plot.



Figure 2.4: Transmitted pressure profile of a diverging wave as a function of time and azimuth angle at typical imaging depth of 50 mm, f=7.5 MHz, divergence=  $45^{\circ}$ , (a) envelope of the pressure from  $-45^{\circ}$  to  $45^{\circ}$  (red line indicates the position of the point used to acquire the 1D signal in (b). (b) time-domain pressure signal at 40° azimuth. The secondary pulse is observed as a longer pulse with lower amplitude compared to the main pulse.

# 2.3. Results

#### 2.3.1. Imaging Scheme Optimization

#### Transmit Central Frequency and Azimuthal Divergence Angle

When evaluating the transmitted diverging waves, we found significant secondary pulse levels in the transmitted time signals. In Fig 2.4(a), where the envelope of the transmit pulse is shown for all azimuth angles, it can be seen that the secondary pulse is present over a large range of the opening angle and gets longer for larger angles, following the primary short pulse after a few microseconds. Fig. 2.4(b) shows the transmit pulse generated by a transducer with a 45° divergence and 7.5-MHz center frequency at a point 50 mm from the transducer, at an angle of 40°.

We evaluated how the pulse shape depended on the beam divergence and transmit central frequency. The level of the secondary pulse has been measured at points located on a spherical surface with a radius of 50 mm ranging from  $-45^{\circ}$ to  $45^{\circ}$  in both azimuth and elevation direction. The maximum of the secondary pulses measured in different directions is shown in Fig. 2.5(a) as a function of central frequency and beam divergence. The function has a triangular local minimum

 $\mathbf{2}$ 



Figure 2.5: (a) Maximum secondary pressure peaks as a function of central frequency and azimuthal beam divergence. Reconstructed images based on a fully sampled array for a central frequency and azimuthal beam divergence of (b). f = 7.5 MHz and  $\varphi_{az} = 45^{\circ}$  (black point in (a)); and (c). f = 6 MHz and  $\varphi_{az} = 35^{\circ}$  (white point in (a)). Red arrows show the secondary reflection and grating lobe caused by poor destructive interference of waves transmitted from each individual element.

region in the frequency band of 5.3 - 7.5 MHz and the azimuthal beam divergence of  $25^{\circ} - 45^{\circ}$ . Any local minimum close to the triangle hypotenuse will achieve low secondary pressure levels for a relatively high central frequency and opening angle. To show the effect of these secondary pulses, images of point scatterers made with transmit beams with different secondary pressure levels are reconstructed. Fig. 2.5(b) shows that the larger secondary pulse (corresponding to the black dot in Fig. 2.5(a) will interact with the scatterers to show prominent secondary echoes of scatterers (arrows) whereas in Fig. 2.5(c), where the transmit beam has a lower secondary-pulse pressure level (corresponding to the white dot in Fig. 2.5(a), the secondary echoes are almost invisible. We choose a central frequency of 6 MHz and an azimuthal beam divergence of  $35^{\circ}$  and these values will be used as bases for the rest of the simulations.

#### Elevational Beam Divergence and Intensity Ripple

With the proposed transducer size and the chosen azimuthal beam divergence of 35°, the single virtual source technique can provide an elevational beam of 10.7°. The transmit beam profile for this elevational beam divergence is illustrated in Fig. 2.6 which shows an intensity ripple over an elevational angle of -3.6 dB. To reduce this ripple, we would need to increase elevational beam divergence. To



Figure 2.6: Transmit beam profile in elevation direction for an elevational beam divergence of 10.7°. The horizontal dash line shows an intensity ripple level of -3.6 dB.

 TABLE 2.2: TRANSMIT BEAM PROFILE NON-UNIFORMITY MEASURED BY THE INTENSITY RIPPLE

 BETWEEN TWO CONSECUTIVE TRANSMISSIONS

| Elevational beam divergence | Intensity ripple (dB) |
|-----------------------------|-----------------------|
| 10.7° (possible minimum)    | -3.6                  |
| 12.5°                       | -3.3                  |
| 15°                         | -2.4                  |
| 17.5°                       | -1                    |
| 20°                         | -0.4                  |

achieve the wider elevational beam divergences of 12.5°, 15°, 17.5°, and 20°, the effective number of elements in azimuth was reduced from 64 to 56, 47, 41, and 36 elements respectively. This azimuthal apodization was needed to maintain the 70° azimuth opening angle when bringing the virtual focus closer to the probe for more elevational divergence. The intensity ripples between two adjacent transmissions are listed in Table 2.2 for different elevational beam divergences as a measure of non-uniformity. The results show a lower intensity ripple for a broader beam which indicates a more uniform image intensity along the elevation direction.

#### **Elevational Beam Divergence and Grating Lobes**

In the next step, the effect of elevational beam divergence on the grating lobe level has been evaluated. The grating lobe directions for a micro-beamformer size of 2, 3, and 4 are  $53^{\circ}$ ,  $32^{\circ}$ , and  $24^{\circ}$  respectively, at the chosen frequency of



Figure 2.7: Transmit pressure at grating lobe directions, normalized to transmit pressure at main lobe for different beam divergences.

6 MHz. The acoustic pressure at grating lobe directions has been compared for elevational beam divergence of  $10.7^{\circ}$ ,  $12.5^{\circ}$ ,  $15^{\circ}$ ,  $17.5^{\circ}$ , and  $20^{\circ}$ . Fig. 2.7. shows that increasing the transmission elevational beam divergence intensifies the grating lobe levels. An elevational beam divergence of  $10.7^{\circ}$ , which is the narrowest possible divergence, with a beam separation of  $10^{\circ}$ , provides the lowest grating lobe level and highest SNR at the cost of some non-uniformity in image intensity. We chose the elevational beam divergence of  $10.7^{\circ}$  and used this value in the remainder of the simulations. With this elevational divergence, the set of steerable fan-shaped beams provides an imaging field of view of  $70^{\circ}$  (width of insonified region at -3 dB) in the elevation direction.

#### 2.3.2. Imaging Performance Evaluation

The angular weighted coherent compounding approach is evaluated for a microbeamformer of 2 - 4 elements by reconstructing images of a single point-scatterer. Applying the proposed angular weighting function reduces the grating lobe level to a value that we will refer to as the grating lobe residual (GLR). Fig. 2.8(a) illustrates the effect of using angular-weighted coherent compounding compared to uniform coherent compounding for a micro-beamformer size of 3 elements. This angular weighted coherent compounding reduces the grating lobe levels by 28, 16, and 13 dB for micro-beamformer sizes of 2, 3, and 4, respectively. Elevational point spread functions for angular-weighted compounding for different micro-beamformer sizes are illustrated in Fig. 2.8(b). The width of the point spread function at -6 dB is illustrated in Fig. 2.9 for different imaging depths in both azimuthal and elevational directions.  $\mathbf{2}$ 



Figure 2.8: (a) Point spread function (PSF) in the elevational direction, for a single point scatterer at 50 mm depth which is reconstructed by uniform coherent compounding and reconstructed by angularweighted coherent compounding. Grating lobe (GL) level and grating lobe residual level are shown for uniform coherent compounding and angular-weighted compounding respectively. (b) Point spread function for angular-weighted compounding for different micro-beamformer sizes and elevational beam divergence of 10.7°.



Figure 2.9: Lateral resolution as a function of depth in azimuth and elevation directions for a microbeamformer size of 3 elements. The horizontal dash line indicates the desired lateral resolution for EWI.

The spatial variance of the proposed imaging scheme in elevation direction is evaluated by a lateral shift variance plot. Fig. 2.10(a) depicts the lateral shift variance plot for micro-beamforming size 3. Although the effect of the 7 transmit beams is visible as non-uniform image intensity in this plot, the lateral resolution, side lobe level, and grating lobe levels are fairly diagonally uniform. In addition to lateral shift variability, LSV-plot was used to compare the proposed angular weighting function with the conventional triangular weighting function [35]. Since the proposed method compounds a high number of transmissions to reconstruct a single receive line, the LSV-plot is more diagonally uniform and has a narrower



Figure 2.10: Lateral shift-variance plot of a single point scatterer at different positions ranging from -35° to 35° in elevation direction which is reconstructed by (a) the proposed weighting function and (b) the triangular weighting function.

main lobe. However, it produces higher clutter around the main lobe as illustrated in Fig. 2.10.

The numerical tissue-mimicking phantom has been imaged to evaluate the image quality in terms of CNR for the various micro-beamformer sizes. The clutter level is increased in hypoechoic regions by increasing the micro-beamformer size (Fig. 2.11.(a-d)), which reduces the CNR in both elevation and azimuth planes as illustrated in Fig. 2.11(e). Narrow transmission beams with a large separation angle cause non-uniform image intensity along elevation direction, as expected from Fig. 2.6. Fig. 2.11(a) and (b) show that the axial resolution (estimated at 0.8 mm) is not affected by the micro-beamforming, as expected, and seems sufficient with respect to the atrial wall thickness.

#### 2.3.3. Imaging of a Beating Atrium Phantom

A rendered volumetric image of the beating left atrial numerical phantom has been generated with a micro-beamformer size of 3 elements and is shown in Fig. 2.12. A quarter of the top hemisphere has been removed to show the wall thickness in the azimuthal, elevational, and c-plane. We can clearly see the spherical object mimicking the left atrium. The atrial wall appeared thicker in the elevational direction which is caused by the lower elevational resolution. A combination of fan-shaped transmit beams and the proposed weighting function successfully suppressed the  $\mathbf{2}$ 



Figure 2.11: Reconstructed images of a tissue-mimicking phantom in (a) elevation direction for a fully sampled array; (b) in elevation direction for a micro-beamformer size of 3 elements; (c) in azimuth direction for a fully sampled array, and (d) in azimuth direction for a micro-beamformer size of 3 elements. Red circles show hypoechoic regions where the CNRs were calculated. (e) Contrast to noise ratio in azimuth and elevation for different micro-beamformer sizes. (f) Schematic representation of tissue-mimicking phantom, rotated to the elevation (a,b) or azimuth plane (c,d).

grating lobes and no obvious grating lobe is visible. However, non-uniform image intensity is rather visible in the elevation direction.

Fig. 2.12(a)-(c) illustrates the tissue displacement maps on 3 orthogonal planes, namely azimuthal, elevational, and c-plane, which were calculated at t = 1, 11, and 21 ms. The electromechanical wave has initiated from a point shown by the blue arrow in Fig. 2.12(a) and propagated omnidirectionally on the atrial surface.

 $\mathbf{2}$ 





Figure 2.12: Tissue frame-to-frame axial displacement map corresponding to electromechanical wave propagation at (a) t = 1 ms, (b) t = 11 ms, and (c) t = 21 ms plotted on orthogonal planes. The blue arrow on (a) shows the initial point of EW and the red arrow on (c) shows the location at which the displacement is perpendicular to the ultrasound axial direction and cannot be estimated. The probe is located left side of the atrium at a 50-mm distance. Please note that the color scale represents axial displacement over 1ms, so a displacement of 10 µm is equivalent to an axial velocity component of 10 mm/s.

Fig. 2.12(b) and (c) show the propagation of the electromechanical wave at 11 and 21 ms after the wave generation. The electromechanical wavefront is clearly visible and the color gradient indicates the radial strains. The electromechanical wave map was masked by an intensity-based mask derived from the B-mode images. There is no unwanted motion detection on other parts of the region of interest. The red arrow in Fig. 2.12(c) shows a region where the radial motion is purely perpendicular to the ultrasound wave axial direction. Therefore, the 1D axial cross-correlation technique could not detect this displacement.

### 2.4. Discussion and Conclusion

In this study, we have proposed a novel imaging scheme for high-frame-rate 3D ultrasound imaging with a matrix intracardiac catheter. The main goal is to enable volumetric imaging with a sufficient frame rate and image quality for electromechanical wave imaging while reducing the data rate to realistic values. The proposed method consists of three main parts: 1) implementing 1D microbeamforming in the elevation direction to reduce the data rate, 2) transmitting fan-shaped beams to suppress grating lobes in the elevation direction while providing a high frame rate, and 3) angular-weighted coherent compounding to further reduce the grating lobe artifacts and improve CNR.

The transducer consists of  $64 \times 18$  elements with a pitch of 160 µm, which is larger than half of the wavelength at 5 - 8 MHz, giving rise to spatial aliasing. Generating a diverging wave from an array with such pitch can lead to a secondary, trailing, pulse as a result of poor destructive interference of the signals from the adjacent elements – see Fig. 2.4. This secondary pulse will also interact with scatterers, leading to secondary reflections in the reconstructed images and thus increased axial clutter levels [36] as illustrated in Fig. 2.5(b). Such axial lobes were previously reported by Rodriguez-Molares *et al.* [39] for coherent plane-wave compounding. They theoretically analyzed secondary pulse formation and proved that late-arriving transmit waves lead to delayed echoes which render as axial lobes. They showed that in a densely sampled array (pitch  $< \lambda/2$ ), late-arriving signals will cancel each other out which reduces the axial lobe level. In this work, we reduce the secondary pulse level by carefully choosing the central frequency and azimuthal beam divergence for the given array configuration (Fig. 2.5(a)).

We intend to image the EW wavefront propagation in 3D with a sufficient spatial resolution by employing a high volume rate of 1 kHz. This is sufficient for local motion estimation over the whole volume, but previous research on EWI [19] suggests that while a local motion detection rate of 0.5 - 2 kHz is required, a slower rate might be sufficient for the motion sampling (the generation of the strain maps), leaving room for more lenient interleaved spatial interrogation. Konofagou and Provost [19] report that a mapping rate of 120 Hz would suffice since the highest frequency content of the electromechanical wave would be around 50 Hz. However, since we operate at a higher spatial resolution and the EW is a complex reaction-diffusion wave, we are not sure that this bound is valid in our situation. Therefore, we preferred to pursue the full 1 kHz volume rate. Furthermore, this also opens the way to other high-frame-rate applications such as shear wave elastography,

ultrafast Doppler, etc.

A high ultrasound frequency was preferred to achieve a sufficient lateral resolution for EWI. On the other hand, a large opening angle was needed to capture the entire left atrium. As shown in Fig. 2.5, a central frequency of 6 MHz and azimuth opening angle of 35° will result in a low secondary-pulse pressure level. Furthermore, at these values, small variations in the fabrication process and delayquantization frequency will not have large effects on the secondary pressure level.

The elevational beam divergence is subject to a trade-off between the imageintensity uniformity, SNR, and grating lobe levels. A wider transmission reduces image intensity non-uniformity, but also generates a higher acoustic pressure in the grating lobe direction. Since we use a single virtual source method, the effective aperture size in the azimuthal direction must be reduced to generate divergences larger than the smallest realizable divergence of 10.7°. Consequently, wider elevational divergence reduces SNR.

To achieve a high SNR and low grating lobe levels we selected a divergence of 10.7° in the elevation direction. This leads to a non-uniform image intensity in the elevation direction. This non-uniformity is not expected to influence the performance of the proposed method for EWI, since the tissue displacement is measured in the axial direction. Non-uniformity of the transmitted power in B-mode can be compensated for by applying a weighting function to the reconstructed images. However, this corrects the non-uniform image intensity at the cost of producing a non-uniform SNR.

By using a dual virtual source technique similar to that proposed by Chen *et al.* [31], the dependence between the azimuthal and elevational beam divergence can be avoided, such that all aperture sizes can be used. However, we found that using a dual source was an unsuitable solution for our design since beams appeared to curve in elevation direction when steering to large angles, which decreases the overlap between the transmit beam directions and the micro-beamforming presteering direction.

The proposed set of steerable fan-shaped beams can insonify a region of  $70^{\circ} \times 70^{\circ}$  which is sufficient to cover the entire left atrium from the center of the right atrium, as illustrated in Fig. 2.12.

Fig. 2.12 illustrates a rendered volumetric image of the left atrium. Since the lateral resolution is lower in elevation, the atrial wall seems to be thicker in this direction in comparison to the azimuthal direction. The image intensity non-uniformity caused by narrow transmit beams in elevation is visible as darker traces. In addition, there are some artifacts at lower depths that are caused by the high clutter level of the proposed method as shown in Fig. 2.10(a). The high clutter levels are mainly caused by implementing 1D micro-beamforming in elevation as illustrated in Fig. 2.8(b). This can reduce the tissue displacement estimation accuracy. Hence, more advanced beamforming techniques such as minimum variance and clutter suppression techniques might be further used to improve motion estimation and electromechanical wave tracking.

In this study, we evaluated the imaging performance of the proposed method in a series of simulations in which we tried to mimic the in-vivo situation. However, there are more practical challenges that could not be investigated in these simulations. For instance, signal-to-noise ratio or detecting EWI onset in the presence of noise, motion, and anisotropic tissue scattering are important in-vivo challenges that should be evaluated extensively in in-vitro, ex-vivo, and/or in-vivo studies.

The tissue motion has been estimated by 1D cross-correlation and illustrated in Fig. 2.12(a)-(c). The atrial wall thickening has been successfully shown as a radial gradient in the frame-to-frame displacement map. However, this is only a limited and highly simplified proof of principle of the electromechanical wave onset detectability. The electromechanical wave onset is usually tracked by finding a zero crossing on strain rate data. Since only three pairs of high-frame-rate volumes have been simulated in this study, a full map of an electromechanical wave is not generated. Nevertheless, the electromechanical wave onset is clearly visible on the tissue displacement maps. The electromechanical wave approximately propagated one-eighth of the sphere circumference at 10 ms which equals a velocity of 2 m/s. These intermediate results show the general imaging performance of the proposed method and its potential for EWI, although this remains to be confirmed by more realistic simulations and shown in experiments.

The proposed method provides an azimuthal lateral resolution smaller than 3.7 mm on the entire imaging depth (10 cm), which is likely sufficient for EWI [7]. The diameter of the targeted 10-Fr catheter limits the transducer size to 3 mm in the elevation direction, leading to a poorer lateral resolution in this direction. Therefore, at depths beyond 40 mm, the resolution in elevation direction exceeds 5 mm. The results show that the proposed method outperforms the state of the art on 3D ICE in terms of frame rate [22]. Moreover, previous works on pseudo-3D EWI showed that 2 to 4 planes already provided additional information for EWI of the atrium [14], [20] to reconstruct a volume by spatial interpolation. Our proposed method provides a volume consisting of 12 planes in elevation direction (natural

divergence of the transducer at 6 MHz is  $6^{\circ}$ ) which can thus provide even more 3D details. In addition, the proposed method acquires the 3D electro-mechanical wave at a single heartbeat which is more suitable than ECG-gated 3D EWI for mapping cardiac arrhythmia. With this increase in 3D detail, the proposed design has the potential to improve the accuracy of electromechanical activation mapping and detect smaller and non-periodic arrhythmogenic sources.

In this study, we have focused on EWI for mapping atrial fibrillation. However, recent studies showed that EWI can be useful for mapping other cardiac conduction system abnormalities and arrhythmias such as atrial flutters [18] or accessory pathways in patients with the Wolf-Parkinson-White syndrome [13].

Many aspects are still unclear regarding EWI. The relation between electrical activation and mechanical contraction is determined by a combination of biochemical and mechanical interactions of partially unknown properties. Especially in cases of nonuniformities or pathology, this relation will be very complex. Whether the mechanical response alone can be used for patient-specific diagnosis remains to be shown. Nevertheless, measuring the mechanical response with a high spatiotemporal resolution is important to gain insight here.

A practical probe implementation requires channel-count reduction since the number of elements far exceeds the number of cables that can be accommodated. A 1D micro-beamforming scheme reduces the data rate by a factor equal to the number of elements per sub-aperture. We have shown that a data rate reduction by a factor of 2 results in negligible image quality loss. A micro-beamformer size of 3 or 4 introduces some imaging artifacts. PSF analysis has shown that the elevational resolution changes negligibly by increasing the micro-beamformer size to 3, but increases from 6.1 mm to 7 mm by utilizing a micro-beamformer of 4 elements at 50 mm imaging depth. Fig. 2.8(b) shows an increase of 16 and 22 dB in residual grating lobe level for a micro-beamformer of 3 and 4 elements respectively.

The channel count is reduced from 1152 (the number of elements) by a factor corresponding to the micro-beamformer size. Since a 10-Fr intracardiac catheter with a diameter of 3.3 mm limits the number of data transferring channels to around 100 micro-coaxial cables [34], the total reduction factor should be at least 12 to account for additional control signals and power supplies. A further onchip reduction by at least a factor of 4 is thus required to reduce the channel count sufficiently. This can be achieved by methods such as analog time division multiplexing [40], in-probe digitization combined with digital multiplexing [30], [32] or coding of the digitized signal [41].



Figure 2.13: Generating a fan-shape transmitting beam with a divergence of  $\varphi_{el}$  in elevation and steered to  $\Theta$  using a virtual source behind the transducer.

In conclusion, the results show a feasible solution for intracardiac 3D EWI. The 1D micro-beamforming approach, in combination with on-chip channel-count reduction, can sufficiently reduce the data rate for transfer over a 10-Fr intracardiac catheter, while the proposed fan-shaped transmit beams and weighted averaging function preserve the image quality. The proposed method has the potential to offer single-beat 3D EWI of the left atrium.

# Appendix I

#### Fan-Shaped Transmit Beam Generation

To calculate a set of delays required for generating the desired beam, first, a virtual source location was determined based on beam divergence, transducer size, and steering direction in elevation, then the transducer effective size in azimuth was determined based on the virtual source location and azimuthal beam divergence (Fig. 2.13). The virtual source location was calculated by

$$X_{vs} = \begin{cases} -sign(\Theta) \frac{L_{el} \tan|\Theta|}{\tan|\Theta| - \tan(|\Theta| - \varphi_{el})} & , \Theta \neq 0\\ 0 & , \Theta = 0 \end{cases}$$
(2.3)

.

$$Z_{vs} = \begin{cases} \frac{X_{vs}}{\tan(\Theta)} & , \Theta \neq 0\\ \frac{L_{el}}{\tan(\varphi_{el})} & , \Theta = 0 \end{cases}$$
(2.4)

where  $\Theta$  is the steering angle with a counter-clockwise direction from the axial direction,  $\varphi_{el}$  is the beam divergence in elevation from a center line to a -6 dB contour and  $L_{el}$  is half of the transducer size in elevation. The Y-coordinate value of the virtual source was always zero because there was no transmit beam steering in the azimuth dimension. Having the virtual source location, the effective transducer size in azimuth was calculated by

$$L_{az} = R_{vs} \tan(\varphi_{az}) \tag{2.5}$$

$$R_{vs} = \sqrt{X_{vs}^2 + Z_{vs}^2} \tag{2.6}$$

where  $L_{az}$  is half of the effective aperture size in azimuth and  $\phi_{az}$  is the beam divergence in azimuth from a center line to a -6 dB contour.

# References

- J. Jalife, O. Berenfeld, and M. Mansour, "Mother rotors and fibrillatory conduction: a mechanism of atrial fibrillation," *Cardiovascular Research*, vol. 54, pp. 204–216, May 2002. DOI: 10.1016/s0008-6363(02)00223-7.
- [2] M. Haïssaguerre, D. C. Shah, P. Jaïs, et al., "Mapping-guided ablation of pulmonary veins to cure atrial fibrillation," The American Journal of Cardiology, vol. 86, K9–K19, Nov. 2000. DOI: https://doi.org/10.1016/S0002-9149(00)01186-3.
- [3] A. Yaksh, L. van der Does, C. Kik, et al., "A novel intra-operative, highresolution atrial mapping approach," *Journal of Interventional Cardiac Elec*trophysiology, vol. 44, no. 3, pp. 221–225, Oct. 2015. DOI: 10.1007/s10840-015-0061-x.

- [4] M. Allessie and N. de Groot, "CrossTalk opposing view: Rotors have not been demonstrated to be the drivers of atrial fibrillation," *The Journal of Physiology*, vol. 592, no. 15, pp. 3167–3170, Aug. 2014. DOI: https://doi. org/10.1113/jphysiol.2014.271809.
- J. Sra and M. Akhtar, "Mapping Techniques for Atrial Fibrillation Ablation," *Current Problems in Cardiology*, vol. 32, no. 12, pp. 669–767, Dec. 2007. DOI: https://doi.org/10.1016/j.cpcardiol.2007.09.002.
- [6] H. Ashikaga, B. A. Coppola, B. Hopenfeld, E. S. Leifer, E. R. McVeigh, and J. H. Omens, "Transmural Dispersion of Myofiber Mechanics: Implications for Electrical Heterogeneity In Vivo," *Journal of the American College of Cardiology*, vol. 49, no. 8, pp. 909–916, Feb. 2007. DOI: https://doi.org/ 10.1016/j.jacc.2006.07.074.
- [7] E. E. Konofagou and J. Provost, "Electromechanical wave imaging for noninvasive mapping of the 3D electrical activation sequence in canines and humans in vivo," *Journal of Biomechanics*, vol. 45, no. 5, pp. 856–864, Mar. 2012, Special Issue on Cardiovascular Solid Mechanics. DOI: https: //doi.org/10.1016/j.jbiomech.2011.11.027.
- [8] E. McVeigh, O. Faris, D. Ennis, P. Helm, and F. Evans, "Electromechanical mapping with MRI tagging and epicardial sock electrodes," *Journal of Electrocardiology*, vol. 35, no. 4, Part B, pp. 61–64, Oct. 2002. DOI: https: //doi.org/10.1054/jelc.2002.37156.
- [9] O. P. Faris, F. J. Evans, D. B. Ennis, et al., "Novel technique for cardiac electromechanical mapping with magnetic resonance imaging tagging and an epicardial electrode sock," Annals of biomedical engineering, vol. 31, no. 4, pp. 430–440, Apr. 2003. DOI: 10.1114/1.1560618.
- [10] B. T. Wyman, W. C. Hunter, F. W. Prinzen, and E. R. McVeigh, "Mapping propagation of mechanical activation in the paced heart with MRI tagging," *American Journal of Physiology-Heart and Circulatory Physiology*, vol. 276, no. 3, H881–H891, Mar. 1999. DOI: 10.1152/ajpheart.1999.276.3.H881.
- [11] F. R. Badke, P. Boinay, and J. W. Covell, "Effects of ventricular pacing on regional left ventricular performance in the dog," *American Journal of Physiology-Heart and Circulatory Physiology*, vol. 238, no. 6, H858–H867, Jun. 1980. DOI: 10.1152/ajpheart.1980.238.6.H858.

- M. Pernot and E. Konofagou, "Electromechanical imaging of the myocardium at normal and pathological states," in *IEEE Ultrasonics Symposium*, 2005., vol. 2, Mar. 2005, pp. 1091–1094. DOI: 10.1109/ULTSYM.2005.1603040.
- [13] L. Melki, C. S. Grubb, R. Weber, et al., "Localization of Accessory Pathways in Pediatric Patients With Wolff-Parkinson-White Syndrome Using 3D-Rendered Electromechanical Wave Imaging," JACC: Clinical Electrophysiology, vol. 5, no. 4, pp. 427–437, Apr. 2019. DOI: https://doi.org/10.1016/ j.jacep.2018.12.001.
- [14] P. Nauleau, L. Melki, E. Wan, and E. Konofagou, "Technical Note: A 3-D rendering algorithm for electromechanical wave imaging of a beating heart," *Medical Physics*, vol. 44, no. 9, pp. 4766–4772, Jun. 2017. DOI: https:// doi.org/10.1002/mp.12411.
- [15] J. Provost, A. Costet, E. Wan, et al., "Assessing the atrial electromechanical coupling during atrial focal tachycardia, flutter, and fibrillation using electromechanical wave imaging in humans," Computers in Biology and Medicine, vol. 65, pp. 161–167, Oct. 2015. DOI: https://doi.org/10.1016/j. compbiomed.2015.08.005.
- [16] J. Provost, V. Gurev, N. Trayanova, and E. E. Konofagou, "Mapping of cardiac electrical activation with electromechanical wave imaging: An in silicoin vivo reciprocity study," *Heart Rhythm*, vol. 8, no. 5, pp. 752–759, May 2011. DOI: https://doi.org/10.1016/j.hrthm.2010.12.034.
- [17] J. Provost, W.-N. Lee, K. Fujikura, and E. E. Konofagou, "Imaging the electromechanical activity of the heart in vivo," *Proceedings of the National Academy of Sciences*, vol. 108, no. 21, pp. 8565–8570, Apr. 2011. DOI: 10. 1073/pnas.1011688108.
- [18] J. Provost, V. T.-H. Nguyen, D. Legrand, et al., "Electromechanical wave imaging for arrhythmias," *Physics in Medicine and Biology*, vol. 56, no. 22, pp. L1–L11, Oct. 2011. DOI: 10.1088/0031-9155/56/22/f01.
- [19] J. Provost, S. Thiébaut, J. Luo, and E. E. Konofagou, "Single-heartbeat electromechanical wave imaging with optimal strain estimation using temporally unequispaced acquisition sequences," *Physics in Medicine and Biology*, vol. 57, no. 4, pp. 1095–1112, Feb. 2012. DOI: 10.1088/0031-9155/57/4/ 1095.

- [20] A. Costet, E. Wan, E. Bunting, J. Grondin, H. Garan, and E. Konofagou, "Electromechanical wave imaging (EWI) validation in all four cardiac chambers with 3D electroanatomic mapping in canines in vivo," *Physics in Medicine* and Biology, vol. 61, no. 22, pp. 8105–8119, Oct. 2016. DOI: 10.1088/0031– 9155/61/22/8105.
- [21] J. Grondin, D. Wang, C. S. Grubb, N. Trayanova, and E. E. Konofagou, "4D cardiac electromechanical activation imaging," *Computers in Biology and Medicine*, vol. 113, p. 103 382, Oct. 2019. DOI: https://doi.org/10.1016/ j.compbiomed.2019.103382.
- [22] D. Wildes, W. Lee, B. Haider, et al., "4-D ICE: A 2-D Array Transducer With Integrated ASIC in a 10-Fr Catheter for Real-Time 3-D Intracardiac Echocardiography," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 63, no. 12, pp. 2159–2173, Oct. 2016. DOI: 10.1109/ TUFFC.2016.2615602.
- [23] R. Fontes-Carvalho, F. Sampaio, J. Ribeiro, and V. Gama Ribeiro, "Threedimensional intracardiac echocardiography: a new promising imaging modality to potentially guide cardiovascular interventions," *European Heart Journal - Cardiovascular Imaging*, vol. 14, no. 10, p. 1028, Jun. 2013. DOI: 10. 1093/ehjci/jet047.
- [24] M. Alkhouli, T. Simard, A. M. Killu, P. A. Friedman, and R. Padang, "First-in-Human Use of a Novel Live 3D Intracardiac Echo Probe to Guide Left Atrial Appendage Closure," *JACC: Cardiovascular Interventions*, vol. 14, no. 21, pp. 2407–2409, Nov. 2021. DOI: https://doi.org/10.1016/j.jcin. 2021.07.024.
- [25] G. Montaldo, M. Tanter, J. Bercoff, N. Benech, and M. Fink, "Coherent plane-wave compounding for very high frame rate ultrasonography and transient elastography," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 56, no. 3, pp. 489–506, Mar. 2009. DOI: 10.1109/ TUFFC.2009.1067.
- [26] P. Santos, G. U. Haugen, L. Løvstakken, E. Samset, and J. D'hooge, "Diverging Wave Volumetric Imaging Using Subaperture Beamforming," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 63, no. 12, pp. 2114–2124, Dec. 2016. DOI: 10.1109/TUFFC.2016.2616172.

- [27] C. Chen, Z. Chen, D. Bera, et al., "A Pitch-Matched Front-End ASIC With Integrated Subarray Beamforming ADC for Miniature 3-D Ultrasound Probes," *IEEE Journal of Solid-State Circuits*, vol. 53, no. 11, pp. 3050–3064, Sep. 2018. DOI: 10.1109/JSSC.2018.2864295.
- [28] S. A. Scampini, "Microbeamforming transducer architecture," U.S. Patent US20080262351A1, Oct. 2008.
- [29] J. D. Larson, "2-D phased array ultrasound imaging system with distributed phasing," U.S. Patent US5229933A, Jul. 1993.
- [30] D. Bera, F. van den Adel, N. Radeljic-Jakic, et al., "Fast Volumetric Imaging Using a Matrix Transesophageal Echocardiography Probe with Partitioned Transmit-Receive Array," Ultrasound in Medicine & Biology, vol. 44, no. 9, pp. 2025–2042, Sep. 2018. DOI: https://doi.org/10.1016/j. ultrasmedbio.2018.05.017.
- [31] Y. Chen, L. Tong, A. Ortega, J. Luo, and J. D'hooge, "Feasibility of Multiplane-Transmit Beamforming for Real-Time Volumetric Cardiac Imaging: A Simulation Study," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 64, no. 4, pp. 648–659, Apr. 2017. DOI: 10.1109/TUFFC. 2017.2651498.
- [32] G. Gurun, C. Tekes, J. Zahorian, et al., "Single-chip CMUT-on-CMOS frontend system for real-time volumetric IVUS and ICE imaging," *IEEE Trans*actions on Ultrasonics, Ferroelectrics, and Frequency Control, vol. 61, no. 2, pp. 239–250, Feb. 2014. DOI: 10.1109/TUFFC.2014.6722610.
- [33] Y. M. Hopf, B. Ossenkoppele, M. Soozande, et al., "A Pitch-Matched ASIC with Integrated 65V TX and Shared Hybrid Beamforming ADC for Catheter-Based High-Frame-Rate 3D Ultrasound Probes," in 2022 IEEE International Solid- State Circuits Conference (ISSCC), vol. 65, Mar. 2022, pp. 494–496. DOI: 10.1109/ISSCC42614.2022.9731597.
- [34] W. Lee, S. Idriss, P. Wolf, and S. Smith, "A miniaturized catheter 2-D array for real-time, 3-D intracardiac echocardiography," *IEEE Transactions on Ul*trasonics, Ferroelectrics, and Frequency Control, vol. 51, no. 10, pp. 1334– 1346, Oct. 2004. DOI: 10.1109/TUFFC.2004.1350962.
- [35] T. Hergum, T. Bjastad, and H. Torp, "Parallel beamforming using synthetic transmit beams [biomedical ultrasound imaging]," in *IEEE Ultrasonics Symposium*, 2004, vol. 2, Apr. 2004, 1401–1404 Vol.2. DOI: 10.1109/ULTSYM. 2004.1418060.

- [36] Y. Zhang, Y. Guo, and W.-N. Lee, "Ultrafast Ultrasound Imaging Using Combined Transmissions With Cross-Coherence-Based Reconstruction," *IEEE Transactions on Medical Imaging*, vol. 37, no. 2, pp. 337–348, Feb. 2018. DOI: 10.1109/TMI.2017.2736423.
- [37] J. Jensen, "FIELD: A program for simulating ultrasound systems," Medical and Biological Engineering and Computing, vol. 34, pp. 351–352, Jan. 1996.
- [38] M. van Wijk and J. Thijssen, "Performance testing of medical ultrasound equipment: fundamental vs. harmonic mode," *Ultrasonics*, vol. 40, no. 1, pp. 585–591, May 2002. DOI: https://doi.org/10.1016/S0041-624X(02) 00177-4.
- [39] A. Rodriguez-Molares, J. Avdal, H. Torp, and L. Løvstakken, "Axial lobes in coherent plane-wave compounding," in 2016 IEEE International Ultrasonics Symposium (IUS), Nov. 2016, pp. 1–4. DOI: 10.1109/ULTSYM.2016. 7728520.
- [40] Q. Liu, C. Chen, Z.-y. Chang, C. Prins, and M. A. P. Pertijs, "A mixed-signal multiplexing system for cable-count reduction in ultrasound probes," in 2015 *IEEE International Ultrasonics Symposium (IUS)*, Oct. 2015, pp. 1–4. DOI: 10.1109/ULTSYM.2015.0141.
- [41] Z. Chen, M. Soozande, H. J. Vos, et al., "Impact of Bit Errors in Digitized RF Data on Ultrasound Image Quality," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 67, no. 1, pp. 13–24, Jan. 2020. DOI: 10.1109/TUFFC.2019.2937462.

# 3

# A COMPACT HIGH-VOLTAGE PULSER FOR MINIATURE ULTRASOUND PROBES

This chapter is based on the publication "A Compact Integrated High-Voltage Pulser Insensitive to Supply Transients for 3D Miniature Ultrasound Probes" in IEEE Journal of Solid-State Circuits Letters, vol. 5, pp. 166-169, June 2022.

## 3.1. Introduction

U ltrasonic imaging is widely used in medicine for its versatile and nondestructive nature. In recent years, integrated circuit research is working on enabling further applications and, at the example of catheter-based devices for intracardiac echocardiography (ICE), moving from 2D to 3D visualization [1]. A major challenge in this transition is posed by the necessity of 2D transducer arrays with around 1000 elements and small pitches in the order of 100 to 200  $\mu$ m to achieve sufficient image quality. To maximize the imaging aperture within probe tips with around 3 mm diameter, circuitry to transmit pressure waves and receive echoes on the same transducer need to be matched to the element area.

For transducer actuation, small circuits have been achieved through the use of



Figure 3.1: Common ultrasound system with level shifters, a pulser and transmit/receive (T/R) switch to receive electronics (RX).

standard CMOS devices [2], [3]. These, however, can not provide sufficient signalto-noise ratio (SNR) at relevant imaging depths due to limited output pressure and the attenuation of ultrasound in the medium. Therefore, high-voltage (HV) transistors are typically applied despite their large size, and transmit/receive (T/R) switches are used to protect the receive path from high potential differences [4]–[6]. A common topology is shown in Fig. 3.1. While a class-A pulser topology as in [7] is area saving, the associated large static power consumption and tissue heating are undesirable and push-pull architectures are more common. These can be unipolar [8], [9] or bipolar [4]–[6], leading to a trade-off between less out-of-band energy for bipolar pulses at the expense of several more HV transistors.

The pull-up path is typically implemented with an HV PMOS and an HV level shifter is required to drive from logic supply levels. The voltage gap can be bridged by HV devices such as transistors [4], [8] or capacitors [10], the latter being preferable in terms of area but more sensitive to HV supply transients preventing regular operation, their cause being high transient pulsing currents and inductive loops, especially in catheter-based probes with limited local supply decoupling.

This work presents a unipolar pulser with an embedded T/R switch and a capacitive HV level shifter. The architecture is designed for beneficial scaling with large arrays and employs a latch-based high-side topology that enables pulsing in the presence of significant HV supply transients. The system is designed for driving a 160- $\mu$ m×160- $\mu$ m PZT transducer element with a central frequency of 6 MHz but the presented techniques are also applicable to other transducers once scaled.

This chapter is organized as follows. Section 3.2 presents the implementation of the transmit circuit, including the pulser, the HV level shifter, and an on-chip



Figure 3.2: Structure and timing of the proposed unipolar pulser.

regulator. Section 3.3 details the fabricated prototype and discusses electrical and acoustical measurements. The chapter ends with a comparison to the prior art and a conclusion.

# **3.2.** Circuit Implementation

#### 3.2.1. High-Voltage Pulser

The proposed HV pulser design is shown in Fig. 3.2. The pull-up and pull-down paths are realized with a single HV transistor, M1 and M2, each, enabling compact unipolar pulsing. Similar to the bipolar design in [4], a T/R switch M3 is integrated with the pulser. This can be implemented as a smaller 5-V transistor given that the HV NMOS shields the RX node from rising above its breakdown voltage.

The transducer array is on one side directly bonded to the ASIC on nodes TD while on the other side an aluminum foil provides a common low impedance path to ground. A transmit event is initiated by closing M1 and keeping M2 open, pulling the transducer to the HV supply. To end the pulse, the transducer is then pulled to ground by closing M2 and opening M1. During this repeatable transmission phase, M3 remains closed to prevent any damage to it. In the receive phase, M3 and M1 are open while M2 conducts incoming signals to the receive electronics.

#### 3.2.2. High-Voltage Level Shifter

While HV transistors can sustain large source-to-drain potential differences, their gates typically still have a limited breakdown voltage. To enable logic-level control,



Figure 3.3: Level shifter architectures based on (a) HV transistors and (b) capacitors.



Figure 3.4: Proposed supply transient insensitive, capacitor-based HV level shifter.

the HV PMOS of the pulser, therefore, has to be driven with a level shifter. Fig. 3.3 shows two common ways to bridge such potential gaps with HV devices [4], [10].

The implementation with an HV transistor and a pull-up resistor provides a reliable and simple solution at the expense of using large HV devices. In addition, the pull-up resistor has to provide sufficient current to pull up the large pulser transistor with a sufficient slew rate, leading to static power consumption. These issues can be addressed with a capacitor-based design given that the bridging components are smaller and can be implemented in metal layers. This however has issues with transients on the HV supply. The transistors can be kept within their safe operating range with additional diodes but the capacitors will maintain their charge and cause an undefined overdrive on the HV PMOS of the pulser or in the worst case flipping of the high-side latch formed between M3 and M4. In area-limited catheter-based probes with substantial inductive supply loops, this can be particularly problematic as the highest current transients on the HV supply occur during pulsing, and dedicated transient-reduction techniques are difficult to integrate without area penalty or compromising the pulse shape.

Fig. 3.4 shows the proposed HV level shifter designed to enable a small footprint in large arrays without introducing supply transient sensitivity on the control of the high-side HV transistor. The HV gap is still bridged with capacitors C1 and C2 but as the high-side circuitry poses a much smaller capacitive load than the large HV PMOS, their size can be reduced while maintaining the same attenuation towards the top plates. The capture circuit accepts differential signals from the low side while rejecting common mode inputs. A differential input signal causes transistors M1/M3 and M2/M4 to act like inverters towards the attached resetset (RS) NAND latch. If the capacitor top plate potentials however get close to each other, as could happen in the event of ripple on the HV supply VDDHV, potentials at nodes R and S will remain high, leading the RS latch to maintain its output state. For a low common mode, transistors M1 and M2 directly connect the output of the capture circuit to VDDHV and a high common mode leads M3 and M4 to connect the common mode to the output. The high-side capture circuit serves the additional purpose of shielding the mismatch-sensitive latch with a less vulnerable structure, making the system more reliable than a direct high-side latch implementation. An output inverter is driving the HV PMOS transistor to present a small load to the latch and enable it to switch quickly.

Unlike its conventional counterpart, this capacitive level shifter maintains the overdrive on the pull-up device of the HV pulser through a driver in a separate supply domain provided by an on-chip regulator generating VSSHS. Capacitive coupling maintains the potential difference even in the event of HV supply transients and the HV PMOS conductivity, therefore, remains stable. The high-side transistors of the level shifter can be small low voltage devices as they are shielded from the substrate potential with an isolation pocket. This pocket can be shared among multiple level shifters, making the device beneficial for arrayed architectures. Should HV supply transients cause potential differences between the capacitor top plates and gate regions of the low voltage transistors to become close to their breakdown limit, parasitic diodes of transistors M3 to M6 will dissipate any excess charge.

#### 3.2.3. High-Side Regulator

To provide the low voltage pocket for driving the HV PMOS of the pulser, VSSHS, a supply 5 V lower than the HV supply, is generated on the periphery of the chip. The circuit is shared by the whole array and implemented as a basic HV PMOS



Figure 3.5: Regulator for high-side low supply.

source follower as shown in Fig. 3.5. The Zener diode Z1 provides an input 5.6 V lower than the HV supply and is biased with a low current from a current mirror. The HV NMOS M3 shields the current mirror transistor M2 by preventing its drain to pass the low voltage supply, VDDLV, enabling a small footprint with a defined mirror ratio. VSSHS is therefore regulated to VDDHV  $- 5.6 \text{ V} + \text{V}_{\text{th,p}}$ ,  $\text{V}_{\text{th,p}}$  being the HV PMOS threshold voltage.

The additional capacitor C1 provides a high-frequency path to protect the gate of HV PMOS M4 in the event of HV supply transients. The decoupling capacitor C2 is sized to accommodate transient currents from driving the HV PMOS pulser transistors and can be as low as 100 pF for at least 20 consecutive pulses of all elements in this design. While the source follower can provide sufficient sinking capabilities for the application even with a low quiescent current, an additional Zener diode Z2 is added for sourcing capabilities in case of an unexpected drop in the regulated output. The power consumption of the regulator is negligible compared to the level shifter and pulser operation.

# 3.3. Measurement Results

A prototype chip has been fabricated in TSMC 180 nm HV BCD technology. It features an array of 8 by 9 elements and contains a pulser as well as an analog frontend per element. The element pitch is 160  $\mu$ m by 160  $\mu$ m of which 27 % is occupied by transmit-related structures, leading to an area per element of 0.008 mm<sup>2</sup>. The HV level shifters of 6 elements are laid out together to share their isolation ring for the high-side circuitry. Next to a 65-V high-voltage supply, a 5-V



Figure 3.6: Electrical measurement of 3 and 30 pulses from the proposed pulser.



Figure 3.7: Comparison of pulsing with the proposed and reference HV level shifter in the presence of supply ripple with 27 nH added supply inductance.

and a 1.8-V supply are used in the transmit circuitry to provide level shifting and logic-level control options. The consumed power largely depends on the imaging mode and is dominated by the dynamic power consumption of driving the transducer. On-chip 160-pF VDDHV decoupling is added according to an estimation of how much could be integrated with a full device.

Electrical characterization has been performed with a 2-pF capacitive load emulating the transducer capacitance. Different pulse sequences are evaluated and shown in Fig. 3.6. Next to the regular array with transducers, two separate pulsers are installed at the periphery of the chip. One performs HV level shifting with the proposed design as the rest of the array while the other implements a capacitive level shifting scheme as shown in Fig. 3.3 with added Zener diodes to protect the high-side transistors against breakdown during measurement. This


Figure 3.8: Die micrograph with PZT transducer array and pulser inset as microscope image and layout view.



Figure 3.9: One transmitted pulse measured by a hydrophone at about 10 cm distance in (a) and reflected at about 5.5 cm from plate-reflector, received by the on-chip analog frontend and sampled at 24 MHz in (b).

additional setup is used to study the behavior of the two circuits when subjected to the same HV supply transients. Since this prototype, however, is more than ten times smaller than an actual imaging device and therefore also has a much lower transient pulsing current, a series inductor is inserted in the HV supply connection. Fig. 3.7 shows the supply variation caused by the 72-element array with an additional inductance of just 27 nH, implementing a low estimate of the supply ripple a full device could experience. The reference structure already shows a variation of the pull-up path conductivity up to complete opening, resulting in an irregular pulse shape and imaging artifacts. The proposed circuit in contrast shows little variation from the regular pulse shape even at the highest tested inductance of 220 nH.

Acoustic verification of the chip is performed with a PZT transducer matrix with a central frequency of 6 MHz. Fig. 3.8 shows a die photo with the array manufactured on top of the chip. Pulses with a peak amplitude of 65 V are

|                          | Integrated HV Transmitter |                       |                                 |                        |  |  |
|--------------------------|---------------------------|-----------------------|---------------------------------|------------------------|--|--|
|                          | This Work                 | JSSC'20 [4]           | JSSC'19 [6]                     | ESSCIRC'14 [9]         |  |  |
| Technology               | 180nm<br>HV BCD           | 180nm<br>HV BCD       | 180nm<br>HV SOI                 | 180nm<br>HV BCD-SOI    |  |  |
| Max output               | 65 Vpp                    | 60 Vpp                | 138 Vpp                         | 100 Vpp                |  |  |
| Pulse freq.              | 6 MHz                     | 9 MHz                 | 2 MHz                           | 10 MHz                 |  |  |
| Load                     | 2 pF                      | 18 pF                 | Not Stated                      | 23 pF                  |  |  |
| Area                     | 0.008 mm <sup>2</sup>     | 0.167 mm <sup>2</sup> | $0.04 \text{ mm}^{2 \text{ a}}$ | 0.12 mm <sup>2 a</sup> |  |  |
| # DMOS                   | 2                         | 10                    | 10                              | 3                      |  |  |
| # HV Diodes <sup>b</sup> | 0                         | 4                     | 2                               | 0                      |  |  |
| T/R embedded             | yes                       | yes                   | no                              | no                     |  |  |
| Level shifter            | capacitive                | HV MOS                | HV MOS                          | HV MOS                 |  |  |
| Pulser                   | unipolar                  | bipolar               | bipolar                         | unipolar               |  |  |

TABLE 3.1: COMPARISON WITH THE PRIOR ART

<sup>a</sup>Not Including T/R. <sup>b</sup>Including zener diodes.

driven from the on-chip pulsers to the bottom plate of the transducer elements posing a 2-pF capacitive load. The top plate of the transducer stack is in a final step connected to an aluminum ground foil that is shared by the whole array and connected to the transmit ground potential of the chip. Fig. 3.9(a) shows the pressure measurement obtained with a hydrophone in water, 10 cm in front of the surface of the assembly. A single pulse was fired from a row of nine elements and the waveform as well as a peak-to-peak pressure level of 34 kPa are in good agreement with our expectations. Verification of the ability to perform pulseecho measurements is obtained in a second experiment in which a plate reflector is installed in front of the array. After transducer actuation, the embedded T/R switch is opened, and received echoes are channeled to the included analog frontend [11]. The obtained results are subsequently sampled at 24 MHz and displayed in Fig. 3.9(b).

Table 3.1 compares the presented transmit circuitry to the prior art. It is expected that the HV level shifter can similarly be applied to bipolar pulsers and that realization of the transmit circuitry in an SOI technology as in [6] would lead to further area reduction due to smaller lateral HV device dimensions and enable higher driving voltages.

# **3.4.** Conclusion

An HV transmit circuit for catheter-based ultrasound probes is presented. A compact pulser with an integrated T/R switch is driven with a novel HV level shifter that is insensitive to HV supply transients. The level shifter does not require any HV transistors, has no static power consumption, and benefits from application in large 2D arrays as used for 3D ultrasonic imaging. Electrical and acoustical measurement results that verify the functionality and performance of the presented techniques have been presented.

# References

- D. Wildes, W. Lee, B. Haider, et al., "4-D ICE: A 2-D Array Transducer With Integrated ASIC in a 10-Fr Catheter for Real-Time 3-D Intracardiac Echocardiography," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 63, no. 12, pp. 2159–2173, Oct. 2016. DOI: 10.1109/ TUFFC.2016.2615602.
- [2] J. Lee, K.-R. Lee, B. E. Eovino, et al., "A 5.37mW/Channel Pitch-Matched Ultrasound ASIC with Dynamic-Bit-Shared SAR ADC and 13.2V Charge-Recycling TX in Standard CMOS for Intracardiac Echocardiography," in 2019 IEEE International Solid- State Circuits Conference - (ISSCC), Mar. 2019, pp. 190–192. DOI: 10.1109/ISSCC.2019.8662531.
- [3] H.-K. Cha, D. Zhao, J. H. Cheong, B. Guo, H. Yu, and M. Je, "A CMOS High-Voltage Transmitter IC for Ultrasound Medical Imaging Applications," *IEEE Transactions on Circuits and Systems II: Express Briefs*, vol. 60, no. 6, pp. 316–320, Jun. 2013. DOI: 10.1109/TCSII.2013.2258260.
- [4] M. Tan, E. Kang, J.-S. An, et al., "A 64-Channel Transmit Beamformer With ±30-V Bipolar High-Voltage Pulsers for Catheter-Based Ultrasound Probes," *IEEE Journal of Solid-State Circuits*, vol. 55, no. 7, pp. 1796–1806, Apr. 2020. DOI: 10.1109/JSSC.2020.2987719.

- [5] K. Chen, H.-S. Lee, A. P. Chandrakasan, and C. G. Sodini, "Ultrasonic Imaging Transceiver Design for CMUT: A Three-Level 30-Vpp Pulse-Shaping Pulser With Improved Efficiency and a Noise-Optimized Receiver," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 11, pp. 2734–2745, Nov. 2013. DOI: 10.1109/JSSC.2013.2274895.
- [6] Y. Igarashi, S. Kajiyama, Y. Katsube, et al., "Single-Chip 3072-Element-Channel Transceiver/128-Subarray-Channel 2-D Array IC With Analog RX and All-Digital TX Beamformer for Echocardiography," *IEEE Journal of Solid-State Circuits*, vol. 54, no. 9, pp. 2555–2567, Sep. 2019. DOI: 10.1109/ JSSC.2019.2921697.
- [7] G. Gurun, C. Tekes, J. Zahorian, et al., "Single-chip CMUT-on-CMOS frontend system for real-time volumetric IVUS and ICE imaging," *IEEE Trans*actions on Ultrasonics, Ferroelectrics, and Frequency Control, vol. 61, no. 2, pp. 239–250, Feb. 2014. DOI: 10.1109/TUFFC.2014.6722610.
- [8] G. Jung, C. Tekes, M. W. Rashid, et al., "A Reduced-Wire ICE Catheter ASIC With Tx Beamforming and Rx Time-Division Multiplexing," *IEEE Transactions on Biomedical Circuits and Systems*, vol. 12, no. 6, pp. 1246–1255, Dec. 2018. DOI: 10.1109/TBCAS.2018.2881909.
- [9] M. Sautto, D. Leone, A. Savoia, et al., "A CMUT transceiver front-end with 100-V TX driver and 1-mW low-noise capacitive feedback RX amplifier in BCD-SOI technology," in ESSCIRC 2014 - 40th European Solid State Circuits Conference (ESSCIRC), Nov. 2014, pp. 407–410. DOI: 10.1109/ESSCIRC. 2014.6942108.
- [10] Z. Liu, L. Cong, and H. Lee, "Design of On-Chip Gate Drivers With Power-Efficient High-Speed Level Shifting and Dynamic Timing Control for High-Voltage Synchronous Switching Power Converters," *IEEE Journal of Solid-State Circuits*, vol. 50, no. 6, pp. 1463–1477, Jun. 2015. DOI: 10.1109/JSSC. 2015.2422075.
- [11] Y. M. Hopf, B. Ossenkoppele, M. Soozande, et al., "A Pitch-Matched ASIC with Integrated 65V TX and Shared Hybrid Beamforming ADC for Catheter-Based High-Frame-Rate 3D Ultrasound Probes," in 2022 IEEE International Solid- State Circuits Conference (ISSCC), vol. 65, Mar. 2022, pp. 494–496. DOI: 10.1109/ISSCC42614.2022.9731597.

# 4

# AN ASIC WITH SHARED HYBRID BEAMFORMING ADC FOR HIGH-FRAME-RATE 3D ICE

This chapter is based on the publication "A Pitch-Matched Transceiver ASIC with Shared Hybrid Beamforming ADC for High-Frame-Rate 3D Intracardiac Echocardiography" in IEEE Journal of Solid-State Circuits, vol. 57, no. 11, pp. 3228 -3242, Nov. 2022.

# 4.1. Introduction

S everal cardiovascular conditions can be addressed using minimally invasive interventions, including the treatment of cardiac arrhythmia through electrophysiology and catheter ablation, trans-catheter valve replacement, closure of atrial septal defects and occlusion of the left atrial appendage [1], [2]. Real-time guidance of these interventions is traditionally achieved through fluoroscopy. However, this is associated with a low resolution of soft tissue and exposure of the patient and the physician to harmful ionizing radiation [3].

Ultrasound imaging can alleviate these disadvantages. But simply obtaining images of the heart from outside the body with a hand-held probe, in a so-



Figure 4.1: (a) Example of an intracardiac echocardiography (ICE) procedure with catheter entry from the inferior vena cava and imaging of the left ventricle from the right atrium. (b) Transducer matrix array integration overview. (c) Schematic of a  $1 \times 3$ -element subarray including transmit (TX) and receive circuitry.

called transthoracic echocardiogram (TTE), suffers from a limited acoustic window through the chest and requires a dedicated operator [4]. It is possible to obtain unobstructed, high-resolution ultrasound images from within the body by imaging from the esophagus in a transesophageal echocardiography (TEE) procedure or from within the heart in an intracardiac echocardiography (ICE) procedure [5]. The latter can often work on the same local anesthesia as the actual intervention as opposed to the, more risky, general anesthesia required for working from the esophagus. This has made ICE one of the most commonly applied ultrasound tools for minimally invasive cardiac interventions [4].

Until recently, a drawback of ICE probes was their limitation to 2D images. This was mostly due to the integration and wiring challenge posed by the larger

vascular system, as shown in an example with entry from the inferior vena cava in Fig. 4.1(a). Within this space, the transducer array has to be accommodated at the tip and all external connections in the shaft. To enhance the visualization for the physician, 3D images were initially created from either manual [7] or motorized [8] rotation of 1D transducer arrays, sacrificing real-time imaging capability or resulting in very low frame rates. In [9], a helical 1D array was applied instead, enabling 3D imaging at higher frame rates at the cost of a limited elevation opening angle of just 22°. The class of forward-looking, ring-shaped probes [10], [11] showed a similar issue in being able to provide 3D images but only for a small volume ahead of the device and therefore not being applicable for the full range of procedures. A way to achieve a sufficient field of view and imaging rate is provided with the advanced integration of 2D transducer arrays in the catheter tip. However, the problem is that for individual connection from each transducer element to an imaging system outside the body, the interconnect becomes limiting to the array size, resulting in insufficient image quality [12]–[14].

transducer matrix that is generally required for 3D imaging [6]. The probes are limited to a diameter of around 3 mm to enable accessing the heart through the

Subarray beamforming has recently been investigated as an approach to interface large arrays, such as the approximately 1000 elements of a typical 3D ICE device [6], [15]. The method shifts part of the receive beamforming, usually applied in the imaging system, into the catheter in the form of delay-and-sum operations on the received signals of a subarray [16]. This effectively reduces the number of connections needed inside the catheter shaft as only the combined signal is transmitted. However, it does not provide the raw data of the full array and introduces focusing errors [17]. These result in increased grating and side lobe levels as well as broadening of the main beam, all negatively impacting image quality. Narrower transmit beams can be used to mitigate these effects but require more acquisitions per volume, ultimately leading to a trade-off between subarray size, with related channel-count reduction, and achievable frame rate [18].

A reduction in frame rate leads to worse motion tracking and can prevent the use of upcoming imaging modes like high-frame-rate blood flow or electromechanical wave imaging [19], which offer more diagnostic potential to physicians but require about 1000 volumes/s. A method to manage the impact on frame rate while still offering cable-count reduction is provided by multiplexing multiple channels onto one cable in analog or digital form. Analog multiplexing has been demonstrated in the time [20], [21] and frequency [22] domain but is constrained

by the limited bandwidth across the commonly applied micro-coaxial cables and suffers from channel-to-channel crosstalk [23]. Digital time-domain multiplexing (TDM), on the other hand, has been shown to benefit from better tolerance to cross-talk, interference and noise [15]. The availability of digital receive signals in the catheter moreover opens the possibility for future co-integration with emerging image processing such as data reduction with machine-learned compression [24], [25] or adaptive beamforming [26]. A major benefit of multiplexing lies in the compatibility with subarray beamforming, as has been shown in digital beamforming of element-level signals [27]–[29] and analog beamforming with subsequent digitization and TDM [15], [30], [31]. While the former requires an analog-to-digital converter (ADC) per element, the subarray area available for the latter makes the scheme more feasible for large arrays.

Both ways, current digital ICE probe designs still suffer from large ADCs. This expresses itself in the reported designs not being able to match the element-level circuitry to the transducer pitch [28], [31], a requirement for a scalable system, or the associated silicon area requirement making co-integration with adequate transmit circuitry impossible [15], [32], [33]. These transmitters need to excite the transducer elements with high-voltage (HV) pulses to obtain sufficient signal-to-noise ratio (SNR) for around 10 cm imaging depth, requiring the use of HV transistors with large isolation rings [6].

In this chapter, a scalable ASIC with a co-integrated  $160 \text{-}\mu\text{m} \times 160 \text{-}\mu\text{m}$  transducer array is presented [34]. While [15] has already shown a large channelcount reduction through the combination of subarray beamforming and digital TDM with a beamforming ADC, the presented converter was too big to be able to include transmitters. Moreover, its associated subarray size of  $3 \times 3$  elements precludes frame rates on the order of 1000 volumes/s. To address these issues, this chapter presents a novel hybrid beamforming ADC, consisting of an efficient charge-sharing successive-approximation-register (SAR) first stage and a compact single-slope (SS) second stage. The ADC architecture achieves the smallest reported power consumption and area among miniature ultrasound probe ADCs, enabling a subarray size of only  $1 \times 3$  elements and thereby pushing the maximum frame rate to the targeted 1000 volumes/s. Moreover, it allows for the integration of per-element 65-V transmitters [35] and frontends, analog subarray beamformers and digitization in a pitch-matched fashion. The system has a 12-fold data-channel-count reduction, resulting in 96 data channels for the envisioned full system [17]. In a trade-off between the available circuit area and the achievable image quality, the transducer pitch was chosen just above the half-wavelength margin. The resulting grating lobe artifacts stay below an acceptable level [17] and can be further reduced through coherent compounding of sub-volume acquisitions [36] and potentially further mitigated through recently shown machine-learning-based image processing techniques [37], [38].

This chapter is organized as follows. Section 4.2 presents the system architecture, with the circuit implementation of the ADC being further detailed in Section 4.3. Section 4.4 covers the fabricated prototype and discusses electrical, acoustic and imaging experiments. The chapter ends with a comparison to the prior art and a conclusion.

# 4.2. System Design

#### 4.2.1. Overview

An overview of the conceptual system, showing common techniques in ultrasound imaging applied in this design, is given in Fig. 4.1(c). Each transducer element is used for the transmission (TX) of pressure waves as well as the subsequent reception (RX) of echoes, generated by reflectors in the imaged medium, in a pulse-echo (PE) cycle. The shared element usage between TX and RX maximizes the achievable aperture in size-constrained probes, as displayed in Fig. 4.1(c).

During the TX phase, HV pulsers are driving the transducer elements based on inputs from a TX controller. The controller applies delays to achieve TX beam steering, forming diverging waves to scan the whole volume with a small number of transmissions. In order to protect the low-voltage RX circuits from the HV operation of the transmitter, a transmit/receive (T/R) switch is placed between the two sections of the design.

Once the transmission is completed, the switch closes and a readout circuit can access the transducer. A common issue in ultrasound imaging systems is the large dynamic range (DR) accumulated by propagation in an exponentially attenuating medium. The attenuation A, in dB, is proportional to the distance z, the attenuation coefficient  $\alpha$  and, for heart tissue, approximately the frequency f[39]:

$$A = 2 \cdot z \cdot f \cdot \alpha \tag{4.1}$$



Figure 4.2: Conceptual sketch of the transducer stack directly manufactured on the surface of the ASIC.

The imaging frequency has to be determined based on a trade-off between axial resolution and imaging depth and is in this design set to 6 MHz with a depth of 10 cm [17]. Combined with an attenuation coefficient of about 0.5 dB/MHz/cm for heart tissue [39], the attenuation could reach 60 dB. Considering an instantaneous DR of 40 dB, the total DR can therefore reach 100 dB. As not all of that range is of interest throughout an RX period and the attenuation is time-dependent, the DR to be handled by the RX circuits can however be reduced by a procedure called time gain compensation (TGC). This is achieved by giving the analog frontend a variable gain that can be adjusted during one PE cycle, giving equally strong reflectors at different distances from the transmitter a similar output and easing the design requirements for the following circuitry [40].

The received signals from each channel in the device are combined in a beamforming operation to reconstruct the information of each voxel in the imaged volume. An SNR gain of  $\sqrt{N}$  for N elements in the beamformer is possible if the system SNR is limited by uncorrelated noise and can aid in the design for a large DR. To reduce the number of connections from the probe to a beamforming system, part of the reconstruction can already be performed in the probe per subarray. This subarray beamforming, also referred to as micro-beamforming ( $\mu$ BF), applies delay-and-sum operations to individual signals of a subarray, effectively steering the RX beam to target the previously insonified area [16]. The combined signals can then be digitized for further local processing and sent to an imaging system.

An overview of the co-integration of the ASIC and a bulk-piezoelectric PZT transducer matrix array is given in Fig. 4.1(b). The 160-µm-pitch stack is an optimized version of what has been presented in [41] and gets manufactured directly

on the surface of the ASIC. It connects to exposed top metal pads on one side and a shared aluminum ground foil on the other side, as shown in Fig. 4.2. The  $8 \times 9$ -element assembly serves as a prototype with an architecture and layout design suitable for scaling to a full array of  $64 \times 18$  elements [17]. While the prototype has a reduced aperture, it is already tailored towards the intended imaging scheme of acquiring a  $70^{\circ} \times 70^{\circ} \times 10$  cm volume in front of the transducer matrix at a frame rate of 1000 volumes/s. Given an imaging depth z of 10 cm and the speed of sound c of 1540 m/s in tissue, a maximum pulse-repetition-frequency (PRF) = c/(2z) of 7.7 kHz can be reached. This provides 7 PE cycles for the formation of each volumetric image at a rate of 1000 volumes/s. To optimize their usage towards the best image quality, the beam profile and subarray receive beamformer size are determined in a trade-off with the required channel-count reduction and necessary circuit area. This work applies seven transmit beams of 70° divergence in the azimuth direction and  $10.7^{\circ}$  divergence in the elevation direction in a sweep of the elevation steering angle to cover the full  $70^{\circ} \times 70^{\circ} \times 10$  cm volume. In receive, subarrays of 3 elements along the elevation direction are applied which pre-steer the received signal in the same direction as the transmit beams [17].

#### 4.2.2. Architecture

The implemented system architecture of the prototype is shown in Fig. 4.3. Each transducer element's bottom plate is connected to an individual pulser and a variable-gain analog front-end (AFE), implementing TGC. The TX part is realized as a unipolar, 65-V design with a pull-up and a pull-down path provided by DMOS transistors. To facilitate test modes and reduce the complexity, the TX control signals are provided externally per row, with an area reserved for an integrated TX controller in a future version. A single low-voltage (LV) transistor, protected against voltage breakdown by the HV NMOS of the pulser, enables T/Rswitching to the input of a low-noise amplifier (LNA)[42] in the receive phase [35]. The LNA is configurable to a voltage gain between -12 dB and 24 dB at a step size of 18 dB and, together with the second-stage programmable gain amplifier (PGA) that provides 6 dB to 24 dB at a step size of 6 dB, achieves a total variable gain between -6 dB and 48 dB at a step size of 6 dB [15]. In each PE cycle, the gain is gradually switched from the lowest to the highest gain setting to first grant sufficient linearity for the initially received strong echoes and later enough gain to detect strongly attenuated echoes out of the noise floor. The PGA is a voltage



Figure 4.3: System architecture including element-level pulsers and analog frontends, subarray beamformers, and a shared hybrid ADC structure per subgroup. On the periphery, each channel is connected to an FPGA via a datalink and an LVDS driver.

amplifier with a compact T-type capacitive feedback network [43], configured for quick settling after switching the gain during active operation. To provide rejection of common interference in the following signal chain, it converts the single-ended signal to a differential output.

The outputs of three element-level circuits are merged in a  $1\times3$ -element subarray beamformer, of which two are combined in a  $2\times3$ -element subgroup. Each subgroup shares a hybrid SAR/SS ADC structure that separately digitizes the two channels at 24 MS/s with a resolution of 10 bits each. The outputs of two subgroups are received by a periphery-level block, providing a datalink to process the received data and applying TDM onto a low-voltage differential signal (LVDS) driver to transfer the data to an off-chip field-programmable gate array (FPGA). Each periphery block serves a  $4\times3$  matrix of elements and 6 of these are arranged to form the prototype of  $8\times9$  elements. A total channel-count reduction of 12 is implemented by the subarray beamformer size of 3 and TDM of the digital outputs of 4 subarrays onto one channel.

#### 4.2.3. Hybrid ADC

To cover the discussed dynamic range of up to 100 dB, a 10-bit resolution is chosen to complement the 54-dB TGC and  $\sqrt{N}$  beamforming gain. The sampling



Figure 4.4: Architecture of the proposed hybrid analog-to-digital (ADC) converter formed between a successive-approximation-register (SAR) first stage and a single-slope (SS) second stage.

frequency should be at least 4 times higher than the frequency of the transducer to maintain an acceptable side lobe level[44], leading to a 24-MHz sampling rate for the 6-MHz device. For similar specifications, SAR ADCs have been shown to be an effective solution due to their high efficiency and Nyquist-rate sampling [45]. Similar to [15], the SAR conversion is performed in the charge domain to benefit from direct integration with a subarray beamformer into a beamforming ADC. However, this design employs a capacitive digital-to-analog converter (CDAC) with metal-oxid-semiconductor capacitors (MOSCAPs). While their voltage-dependent capacitance excludes them from use in charge-redistribution ADCs, in chargesharing topologies MOSCAPs reduce area through high integration density and improve comparator offset as well as noise tolerance [46].

One drawback of SAR ADCs in general is the exponential increase of the CDAC size per added bit of resolution in the binary search algorithm. This can be alleviated by means of combination with a slope ADC that digitizes the residual of a SAR first stage in a hybrid converter [33], [47]. A slope ADC can be compact but suffers in efficiency when converting with a high resolution and sampling rate. In the following, the architecture of a hybrid SAR/SS beamforming ADC operating in the charge domain is presented. An optimal split between the two stages



Figure 4.5: (a) Schematic overview of subarray beamformer. (b) Conceptual S/H-cell timing diagram with maximum positive and negative angles.

is found in a 6-bit SAR and a 5-bit SS conversion with one bit of redundancy between them. This considers the required size reduction of the SAR stage and added complexity to the slope stage of the design. In order to benefit from the reduction of the CDAC size, the gained area must outweigh that of the added SS blocks. As the unit capacitor of the CDAC is still sized for 10-bit linearity, this is mainly achieved by enabling a CDAC size that is actually linearity-limited, good MOSCAP matching and the overhead reduction due to a reduced number of unit capacitors.

The ADC is interfacing with the subarray beamformer as shown in Fig. 4.4, avoiding the need for an additional high-bandwidth buffer compared to converters with conventional sampling [33], [47]. The delaying of the received signals is implemented by means of capacitive sample-and-hold (S/H) cells, similar to [48]. The summation of the delayed signals can then be achieved passively by switching the right cells together at the input of the ADC, as in [15]. In contrast with [15], the control-logic of the subarray beamformer, shown in Fig. 4.5(a), can also provide delay quantization independent of the sampling frequency. This is enabled by separate read- and write-pointers,  $S_{RDx}$  and  $S_{WRx}$ . The former is advancing based on completed ADC conversion cycles,  $ADC_{RDYC}$ , while the latter advances based on an adjustable input trigger, in this case a locally generated 24-MHz signal and its 180°-phase-shifted version, TRG and TRG. While each ADC requires its own read-pointer, the write-pointers can be shared between multiple channels. The pointers are realized as ring-counters, and different steering angles, determined by ANG, can be realized by delaying the reset signal between the pointers. An



Figure 4.6: Conceptual timing diagram showing essential nodes of the hybrid ADC and how two subarray outputs are digitized by the shared structure.

area-efficient implementation of this delay is achieved by re-using the pointer of one AFE, AFE0, as the reference counter for the delay. Based on ANG, this can be used to implement the delay with a simple multiplexer and a compact reset-set (RS) latch. The resulting S/H-cell switching patterns are shown for a maximum positive and negative steering angle in Fig. 4.5(b). A delay resolution of 20.8 ns is achieved with a range up to 125.0 ns, offering steering capability of the axis of the beam in a  $\pm 30^{\circ}$ -window.

The conceptual timing of the converter operation is shown in Fig. 4.6. Following the subarray beamformer, the combined charge can be quantized through the positive or negative connection of binary-scaled units of a pre-charged CDAC and detection of the polarity with a discrete-time comparator (DTC). After this SAR conversion, the possible residual charge range between nodes  $V_{Px}$  and  $V_{Nx}$  has been reduced to the least-significant bits (LSBs) and can thereafter be digitized by applying a differential slope with compact current sources on the same nodes. To have the redundant range between the two stages work for positive and negative decision errors in the SAR conversion, part of the CDAC is switched after the first phase to implement a level shift, placing the SAR residual in the middle of the SS conversion range. As long as the residual is not contaminated and they are within the redundant range, all SAR decision errors, such as DTC noise, CDAC settling errors, and mismatch between the SAR and SS comparator offset, can be covered.

High area efficiency is achieved by hardware-sharing between two neighboring subarrays, forming one shared ADC structure. The two phases of the conversion are executed in two periods of a 48-MHz clock, twice as high as the ADC sample rate, and while the first subarray performs the SAR conversion, the second applies the slope and vice versa. In this fashion, the SAR reference pre-charger, as well as the slope generator, can be shared and a drawback that is typically introduced in the hybrid architecture can be mitigated: Usually, the combination of a SAR and a SS stage requires the introduction of an asynchronous DTC for an efficient SAR part and a continuous-time comparator (CTC) to avoid a high-frequency comparator clock for the SS[33], [47]. While this is still the case, they can be shared between the two subarrays with the addition of a small pre-amplifier that provides isolation and serves as part of both comparisons. Another benefit of hardwaresharing is that the SAR CDAC can be switched out and slowly pre-charged during slope conversion. Compared to designs with conventional references [33], [47], this allows for the use of a low-bandwidth reference with a constant current drawn from the supply. Moreover, it does not require an additional CDAC used for ping-pong operation as in[15]. To further facilitate application in large imager arrays, critical nodes in the circuit, including the SAR reference, the slope generator, and the DTC offset, are calibrated against mismatch.

As in ultrasound probes, the area underneath the transducer matrix is typically very limited to achieve a scalable, pitch-matched design but there is more space available at the periphery [32], [40], [48], the received signals are brought out of the core as soon as they are quantized. This applies to the return-to-zero (RZ), asynchronous SAR outputs, SAR<sub>P</sub> and SAR<sub>N</sub>, which are gated versions of the DTC outputs,  $D_P$  and  $D_N$ , as well as the SS outputs, SS<sub>P</sub> and SS<sub>N</sub>. The SAR and SS outputs of both converted channels are each carried on one connection to the periphery and correctly merged there to minimize interconnect space.

#### 4.2.4. Datalink

An overview of the datalink is shown following Fig. 4.7. The SAR and SS outputs of two shared ADC structures, serving 12 elements in total, are captured on the periphery by separate receivers. The SAR receiver recovers the SAR output data, as well as a clock that can be used to sample the data in the following stage, directly



Figure 4.7: Datalink block diagram showing processing of the outputs of 4 ADCs and the clocking scheme shared across the ASIC.

from the asynchronous RZ SAR output [15]. The SS receiver, on the other hand, converts the differential time-domain output of the SS into a short pulse. This pulse is used to latch the output of a delay-locked-loop (DLL) that divides the 48-MHz clock period into 32 segments, corresponding to a 5-bit word. As the SS comparison in the core is based on the same time reference, this word is an accurate representation of the second-stage ADC output. The resulting 32-bit word is then converted from a thermal to a binary code to simplify further processing and can be sampled with the same pulse as the latch array. The asynchronous SAR and SS data can subsequently be synchronized to the periphery clock by dual-clock first-in, first-out (FIFO) memories. To reduce the final data bandwidth to the imaging system, the recombination block merges the 6-bit SAR and 5-bit SS codes into the actual 10-bit output code of the ADC. This is done by summing both parts and removing the redundancy introduced between the two stages in the following manner [47]:

| SAR      |   | 1 | 0 | 0 | 1 | 0 | 1 | -1 |   |   |   |
|----------|---|---|---|---|---|---|---|----|---|---|---|
| SS       | + |   |   |   |   |   | 1 | 0  | 1 | 0 | 1 |
| Combined |   | 1 | 0 | 0 | 1 | 0 | 1 | 1  | 1 | 0 | 1 |

The subtraction of the redundancy is implemented with the two's complement. The outputs of two recombiners are subsequently merged through TDM, realized by arranging two 10-bit words at 48 MHz to 8-bit words at 120 MHz. The following structures are based on [15] and first apply 8b10b-encoding to enable a standard transmission protocol [49]. The 10-bit output words are then serialized to a 1.2-Gbps output stream to conventional LVDS drivers with timing provided by a DLL. Both DLLs are based on [50], shared among the whole ASIC and operate on a single 240-MHz system clock provided from an FPGA. A 120-MHz clock is generated locally at the periphery for the encoder and TDM while a 48-MHz clock provides the timing for all core circuitry as well as the data reconstruction on the periphery. While a balanced clock distribution in the core is required as a time reference for the generation of the intended RX and TX beams, the 48-MHz clock phase of the core does not need to be precisely matched to that of the periphery as only the clock period is relevant for clock-data recovery (CDR).

# 4.3. Circuit Implementation

#### 4.3.1. Timing

As the key enabling block of the system, the implementation of the shared hybrid ADC structure is further detailed in this section. The timing can be followed in Fig. 4.8(c). A non-overlapping (NOV) clock generator, shown in Fig. 4.8(a), creates a division of the 48-MHz system clock,  $CK_{SYS}$ . The resulting synchronous NOV signals are used especially in the operation of the SAR reference as well as the SS circuitry and prevent interference from sharing blocks. The SAR reference, shown in Fig. 4.9(a), is configured as a voltage-controlled current source with a calibrated input  $V_{CAL-SAR}$  and source degeneration  $R_S$  to increase its linearity and lower its transconductance. The current in the pre-charger is always running and steered between the two DACs of the converter in fixed time intervals, with a short reset period applied at the beginning of each charging. The slope generator in Fig. 4.9(b) is also implemented as a steered voltage-controlled current source between the two parts of the shared ADC structure and calibrated by  $V_{CAL-SS}$ .



Figure 4.8: (a) Synchronous timing generator of the shared converter. (b) Loop determining the asynchronous ADC timing. (c) Conceptual timing diagram.

very short NOV period  $SS_x$  is generated locally to prevent connection of the preamplifier input nodes of the two involved ADCs,  $V_{Px}$  and  $V_{Nx}$ , while still having the slope period closely matched to that of  $CK_{SYS}$ . If  $SS_{RDYx}$  indicates a SS conversion has finished, the input of the corresponding pre-amplifier is connected together to a mid-rail node  $V_{CM}$ , setting up the next conversion.

While the reference circuits require determined timing, particularly the comparators benefit from more flexible control to make most efficient use of the  $CK_{SYS}$ period. Feedback loop (1) in Fig. 4.8(b) offers this by applying control signals not based on a clock but based on finished ADC conversion parts. Two area-efficient SR-latches determine a completed conversion from the CTC outputs  $C_{P/N}$  and the RDY[0] signal of the last bit in the SAR logic. If both parts are ready, an internal flip-flop (FF) is in regular operation directly toggled with ADC<sub>RDY</sub>, indicating that the state of the converter structure shifts and the CTC and DTC can be switched,



Figure 4.9: (a) Schematic of the CDAC reset and pre-charger. (b) Slope generator schematic with reset and switching blocks.

enabling a longer settling period for the pre-amplifier before the next part of the conversion starts. The reset of the RS latches for the next conversion part is determined by the combination of the  $ADC_{RDY}$  and the ACT<sub>S</sub> signal, the second being a latched version of the active signal indicating the passing of an NOV period. The signal needs to be latched as the SS conversion can extend over the period in which the slope generator is applied because of the delay of the CTC. To still enable successful conversion, the comparators can be connected longer and, while the slope generator is already shifting to the other converter part, still retrieve the output. In order to prevent loss of synchronization with the periphery in the case of a fault in the SAR residual, a stop signal STP is also issued by  $ACT_{S}$ . This signal can force a stop of the slope conversion by setting the RS latch and has a delay designed to be just above the delay of the CTC. The following SAR conversion typically doesn't suffer from an over-ranging slope conversion as it is timed asynchronously with sufficient margin. The following SS conversion can also still complete successfully as the over-ranging period would usually fall into the conversion range covering the redundant range.

#### 4.3.2. Comparator

Each ADC has its own pre-amplifier to shield the signal to be quantized from the interference of the switching and operation of the following stages as displayed in Fig. 4.10(a). An efficient current-reuse topology is employed and complemented with cascoding transistors to increase the output impedance. Part of the tail current source is split off to enable common-mode regulation of the differential



Figure 4.10: (a) Schematic of the pre-amplifier with switches to second stages. (b) Schematic of the continuous-time comparator. (c) Schematic of the discrete-time comparator.

amplifier with a small auxiliary amplifier sensing the output nodes. The structure can be configured to be part of the SAR and SS comparison by adjusting the impedance at the output nodes,  $O_{Px}$  and  $O_{Nx}$ . During slope conversion, a high impedance is set to provide a first gain stage for the CTC while during SAR conversion a low impedance is implemented to enhance the amplifier bandwidth and SAR timing.

A second gain stage is used in the SS phase to amplify small slope residuals to logic levels within the conversion time. The differential amplifier has a passive common-mode feedback architecture and tail cascodes to increase the output impedance as shown in Fig. 4.10(b). The DTC in Fig. 4.10(c) is based on [51] and, despite the pre-amplifier, still employs an additional dynamic pre-amplifier as the benefit from shielding the continuous pre-amplifier outputs from kick-back during slope conversion outweighs the added cost. It can, moreover, be used to calibrate the offset of the DTC similar to [52], first-order matching the SAR and SS stage offsets given that the large pre-amplifier gain during SS conversion significantly reduces the impact of the second stage continuous amplifier offset.



Figure 4.11: Asynchronous ADC logic core determining the CDAC switching, SAR output code and discrete-time comparator calibration.

# 4.3.3. Logic & CDAC

Fig. 4.11 shows the logic controlling SAR switching. An overview of the CDAC is given in Fig. 4.12(a) and example conversions of one positive and one negative

4



Figure 4.12: (a) Overview of the SAR DAC with unit cell details. (b) Exemplary conversion diagram showing the residual on the sampling capacitors.

input are illustrated in Fig. 4.12(b). The DAC has a binary scaled section,  $C_{5.1}$ , for the 6-bit SAR conversion and is controlled with  $TOP_x$  and  $BOT_x$  to switch the unit in either to raise or lower the difference between  $V_{Px}$  and  $V_{Nx}$ . The structure is laid out symmetrically as multiples of a unit cell to mitigate effects such as mismatch and charge injection. The unit capacitance in the pre-charged state is about 7 fF and five times lower when depleted. Charge-redistribution to create binary charge levels without an exponentially growing CDAC, as in [53], is only applied to generate a level half of  $C_1$ . This is used to shift the SAR residual to the middle of the redundant range for slope conversion because matching of that level has less impact on the converter linearity. The process is realized by first including  $C_{CR}$ during the pre-charging of the CDAC with  $Px_{NOV}$  and then, controlled by  $RD_x$ , sharing its charge with the unit used for level-shifting,  $C_{LS}$ , through the shared node CR. This is done between the first and third SAR comparison unless SAR or SS calibration is performed. Regardless of the outcome of the last SAR bit,  $C_{LS}$  is always switched in to lower the residual with  $BOT_{xLS}$ . The last SAR bit decision does however influence the switching of the two remaining DAC units, the regularly pre-charged  $C_{PLS}$  and the empty unit  $C_{NLS}$ , as shown in Fig. 4.13(b). In regular operation, C<sub>PLS</sub> is switched in for a positive SAR residual, effectively



Figure 4.13: Asynchronous dynamic comparator clock as well as reset generator schematics in (a) and level shifter state diagram in (b).

mapping it onto the negative range as it is similar to  $C_1$  in charge.  $C_{NLS}$ , on the other hand, is switched in with the same polarity but only for negative SAR residuals with the purpose of closer matching of the voltage attenuation posed by the CDAC for both cases. The process is illustrated in Fig. 4.12(b) and enables the slope to always run in the same direction, reducing the SS complexity.

The SAR logic is based on latches due to their area efficiency and works together with the asynchronous DTC timer in Fig. 4.13(a). To ensure locking of the result of a DTC decision and settling of the CDAC before the next comparison, an additional delay is introduced in the SAR clock generator. To also avoid sampling more than one bit per comparator decision, delay cells could be added between the stages of the SAR logic. But in this design, it is instead chosen to lock the stages with  $DC_{RDY}$  to save power and area. In order to still function well in the case of an over-ranging SS conversion, the whole latch array is cleared with ACT and the SAR clock  $CK_{SAR}$  is halted by the set active signal ACT<sub>S</sub>. This makes sure that there is no switching of the CDAC before the correct sample is applied and that the new SAR conversion only begins after the correct comparator is switched in.

#### 4.3.4. Calibration

The last stage of the SAR logic is used to calibrate the offset of the DTC. After the last SAR comparison, the pre-amplifier is disconnected and the input of the DTC is shorted together with  $RST_{DTC}$ , such that the following comparison samples the polarity of the offset [15]. To optimize the precision, the next comparison is not triggered regularly but delayed by halting the seventh  $CK_{SAR}$  pulse of the SAR cycle, effectively enhancing settling. This is achieved with a delayed SAR<sub>RDY</sub> and  $FB_{CAL}$  signal, as shown in Fig. 4.11. The sampled offset polarity is then used to



Figure 4.14: Conceptual diagram of the calibration charge pumps in (a), the related SAR and SS controls in (b) and calibration switching overview in (c).

switch the control inputs of a regulator for the  $V_{CAL-DTC}$  node of the DTC,  $UP_{DTC}$  and  $DN_{DTC}$ . All calibration circuits in this design work in a negative feedback loop and with charge pumps as conceptually displayed in Fig. 4.14(a). The comparator calibration can, except for during SAR or Slope calibration, determined with CAL, always run in the background but isn't required to complete each conversion cycle. To minimize the chance of losing synchronization with the periphery, and therefore one PE cycle, the calibration step is therefore not required for setting ADC<sub>RDY</sub> and is skipped without harming the operation if there is a seventh SAR decision that takes a long time to converge.

The SAR and SS references are additionally calibrated to adjust for inter-stage gain errors, process mismatch and temperature variation. The procedure first follows regular SAR operation with a negative full-scale reference input provided through the PGA and then makes use of the  $C_{PLS}$  DAC unit cell to assess the current status. As listed in Fig. 4.13(b) and illustrated in Fig. 4.14(c), it is used to either fully reduce the residual to zero for SAR calibration or, for slope calibration, to get a residual equal to the maximum SS input, including the redundant range. Using the same cell, just with different polarity, enables accurate slope calibration to the maximum SAR residual range irrespective of a pre-amplifier offset. However, as the offsets of the two pre-amps can be different, calibration is only obtained with one DAC. The other part of the shared ADC structure still performs DTC offset cancellation in parallel. While this one-sided calibration generally introduces an



Figure 4.15: (a) Micrograph of the chip including transducer matrix array with magnifications of the core and subgroup areas. (b) Distribution of receive power and area per subgroup of  $3 \times 2$  elements.

offset between the two converters due to pre-amplifier offset differences, there are no implications from DAC mismatch beyond regular linearity investigation and no significant performance difference could be measured between shared structures.

The modified residuals are ultimately used to first adjust the SAR reference,  $V_{CAL-SAR}$ , based on detecting whether there was a zero-crossing in the seventh comparator decision. This is realized by checking the three least significant bits (LSBs) of the SAR ADC, as shown in Fig. 4.14(b). The additional delay for improved settling in the seventh comparator cycle is still applied while the comparators are not switched early to ensure the calibration completes. The slope control voltage,  $V_{CAL-SS}$ , on the other hand, is adjusted based on letting the full residual run in a regular SS operation. It is then detected if it was completed early with  $SS_{RDY1}$ , or needed to be stopped with STP.

# 4.4. Experimental Results

The reported design has been fabricated in a 0.18- $\mu$ m HV BCD process with an active area of about 1.8 mm<sup>2</sup> for the 8×9 element core and about 0.8 mm<sup>2</sup> for the periphery circuitry. A chip with a prototype transducer array manufactured on

89

its surface is shown in Fig. 4.15(a). It measures  $5 \times 5 \text{ mm}^2$  to ease the prototype transducer manufacturing and includes two rings of dummy transducers to reduce edge effects of the small array. The array is designed to have a center frequency of 6 MHz and a fractional bandwidth in the order of 50%. A magnification shows the pitch-matched floor plan of a  $3 \times 2$  element subgroup consisting of elementlevel pulsers, two  $3 \times 1$  element subarray beamformers and two hardware-sharing ADCs. The transmit circuits operate from a 65-V HV supply, a 5-V supply to drive the pulser transistors and a 1.8-V supply for logic-level control. All receive circuitry is powered from a 1.8-V supply with the addition of a 1.2-V supply for the DLL delay cells and a 2.3-V supply for the SAR reference. An overview of the area and receive power distribution per subgroup is given in Fig. 4.15(b). Due to area-intense HV isolation, the pulser circuits (TX) occupy the largest part of the subgroup area with 31 %, while the hybrid ADCs cover only 13 % of the space and consume about as much power as the analog frontend. As the TX operation has a very small duty cycle, the total power consumption is dominated by the receiver, with 0.65 mW/element in the core and 1.23 mW/element including the datalink and LVDS drivers.

Electrical and acoustic experiments have been conducted. The chips are wirebonded to separate daughter boards and samples for electrical characterization have additional wire-bonds to connect transducer pads to the PCB while acoustic samples have a co-integrated transducer array. All daughter boards are mounted on a common, custom mother board that is connected to a commercial FPGA [54]. Measurements are conducted by observing the ADC outputs as provided through the datalink, received by the FPGA and forwarded to a PC. Characterization of the TX part has previously been shown in [35] and is not covered in this section.

#### 4.4.1. Electrical Measurements

The characterization of the receive transfer function shown in Fig. 4.16 is obtained by providing sinusoidal inputs of varying frequency to the transducer pads from a waveform generator and referring the ADC output to the output of the AFE. The circuit shows the expected mid-band gain range of 54 dB in 10 steps from -6 dB to 48 dB with an 8.1-MHz cut-off frequency in the highest gain setting. The difference among the gain steps in low-frequency roll-off is caused by the feedback configuration and does not affect the imaging as the output is filtered around the center frequency of the transducer. Fig. 4.17(b) shows the time-domain recording



Figure 4.16: Receive transfer function measured across frequency for all gain settings of the analog frontend.



Figure 4.17: (a) TX/RX control, TX voltage and RX output code recorded with the intended pulse-repetition-interval (PRI), with zoom-in of the switching period. (b) Recording of one RX output with an exponentially decaying sinusoidal signal at the analog frontend input and active time gain compensation.

of an ADC output code with an exponentially decaying 6-MHz sinusoidal signal provided at the input of the analog frontend. By switching through the 10 gain settings of the analog frontend, the output amplitude can be kept within the observable range, verifying functional time-gain compensation within one receive period, with a small settling time after gain switching. Combined with the ability to operate at the intended PRF of 7.7 kHz, corresponding to a pulse-repetition-interval (PRI) of 130 µs, as demonstrated in Fig. 4.17(a), this verifies that the chip



Figure 4.18: Power spectral density plot showing the peak SNR of the entire receive signal path with a 5.95-MHz sinusoidal input to the analog frontend.



Figure 4.19: (a) Example of the SAR and SS calibration procedures of two different ADCs. (b) Heatmap showing the relative difference in the total calibration settling time from a common starting point across the ASIC.

enables recording of the full imaging depth at the targeted frame rate.

The peak ADC output power spectral density is shown in Fig. 4.18, measured with a 5.95-MHz sinusoidal signal from a waveform generator. As the waveform is provided at the analog frontend input, the whole receive path is characterized to a peak SNR of 52.3 dB, with a folded -35-dB third and -62-dB fifth harmonic from the PGA output stage in an 80 % bandwidth around the transducer center frequency.

A test mode in which the ADC output of one channel can be read without recombination of the SAR and slope output data is employed to follow the ADC



Figure 4.20: Interference test showing the power spectral densities of both outputs of one shared ADC structure with concurrent 1-MHz and 2.25-MHz inputs.

calibration procedures. The global starting potentials for the SAR and slope reference can be individually set in a reset state and are prior to this test placed away from the final settling points. Fig. 4.19(a) shows how the SAR and slope local calibration nodes converge to the final values at the example of the bottom-right and bottom-left output codes of the SAR and slope stage. The SAR algorithm tries to fully reduce a full-scale input, leading to a settled value at an output code of zero. The slope algorithm, on the other hand, adjusts the slope current to detect the zero-crossing of a full-scale slope stage input just at the end of the associated time interval and therefore settles around the maximum slope output code. An overview of the combined SAR and slope calibration time until a settled state is reached from the same starting point is shown per shared ADC structure in Fig. 4.19(b), giving an indication of the channel-to-channel mismatch.

To investigate how the high integration density, as well as hardware-sharing within each subgroup and at the periphery, affect the cross-talk between channels, an electrical measurement, studying the outputs of the two co-integrated subgroup ADCs, is performed. Their inputs can be accessed separately and are concurrently driven by two outputs of a waveform generator, in this example at 1.00 MHz and 2.25 MHz. The corresponding output spectra of the two ADCs are both plotted in Fig. 4.20 and show no significant tones at the fundamental or harmonic frequencies of the other channel. This implies that a circuit cross-talk better than -75 dBc is achieved and that other sources like mechanical cross-talk in the transducer array should be dominant.

Table 4.1 summarizes the performance of the hybrid ADC and gives a comparison to the state-of-the-art. In contrast with general-purpose converters, this Mother Board

|                       | This Work             | JSSC'21 [32]           | VLSI'19 [33]          | JSSC'18 [15]          |  |  |  |
|-----------------------|-----------------------|------------------------|-----------------------|-----------------------|--|--|--|
| Technology            | 180 nm BCD            | 180 nm                 | 180 nm                | 180 nm                |  |  |  |
| Architecture          | SAR & SS              | SAR                    | SAR & SS              | SAR                   |  |  |  |
| Resolution            | 10 bit                | 10 bit                 | 10 bit                | 10 bit                |  |  |  |
| Sample Rate           | 24 MHz                | 20 MHz                 | 30 MHz                | 30 MHz                |  |  |  |
| Low BW<br>Driver      | $\checkmark$          | ×                      | ×                     | $\checkmark$          |  |  |  |
| Reference<br>Included | $\checkmark$          | ×                      | ×                     | $\checkmark$          |  |  |  |
| Area *                | 0.010 mm <sup>2</sup> | 0.026 mm <sup>2</sup>  | 0.013 mm <sup>2</sup> | 0.046 mm <sup>2</sup> |  |  |  |
| Power *               | 0.82 mW               | 1.23 mW                | 1.14 mW               | 1.42 mW               |  |  |  |
| * Per Subarray        |                       |                        |                       |                       |  |  |  |
| b)<br>a)              |                       | XYZ<br>Stage           |                       |                       |  |  |  |
|                       |                       | External<br>Transducer |                       | hantom                |  |  |  |

 TABLE 4.1: PRIOR ART COMPARISON IN MINIATURE ULTRASOUND PROBE ADCs

Figure 4.21: (a) Overview of acoustic measurements setup. (b) Inset showing watertank with external transducer. (c) Imaging experiment details.

C)

ASIC

Watertank

Board

design implements several application-oriented features like direct integration with a subarray beamformer, allowing for a low-bandwidth ADC driver, and including a low-bandwidth SAR reference. Therefore, Table 4.1 compares the design with prior ADC designs targeting miniature ultrasound probes. Set against the most comparable design [15], the ADC occupies more than 4x less area and consumes more than 1.5x less power. Even in comparison with the more general designs [32], [33], the efficient architecture enables the lowest reported area and power consumption.

#### 4.4.2. Acoustic Measurements

An overview of the acoustic measurement setup is given in Fig. 4.21(a). All acoustic measurements are obtained with the daughter-board and ASIC with transducer array directly interfacing with a watertank. Fig. 4.22 shows the input-referred voltage noise spectral density in the highest gain setting with the transducer ele-



Figure 4.22: Input-referred spectral voltage noise density with and without calibration of the static mismatch pattern of the subarray beamformer.



Figure 4.23: ADC output measured during sweep of the transducer surface pressure in all gain settings, showing a 91-dB dynamic range.

ments loaded by water. Two tones appear in the spectrum at  $f_s/6$  and its second harmonic,  $f_s$  being the sampling frequency of the ADC. These are generated by mismatch of the subarray beamforming cells and can be removed by recording and subtracting the static pattern as shown. The input-referred voltage noise density is 12.7 nV/ $\sqrt{Hz}$  at 6 MHz, around 5 nV/ $\sqrt{Hz}$  higher than what the electronics are designed for, with the difference being attributed to the thermal noise of the transducer.

Fig. 4.23 displays a sweep of the transducer surface peak pressure with measured ADC output for all gain settings, used to characterize the dynamic range of the receive path. The pressure waves are generated from an external commercial transducer [55] connected to a waveform generator that transmits sinusoidal waves at 5.5 MHz as shown in Fig. 4.21(b). The displayed surface pressure is calibrated with a commercial hydrophone [56] with known sensitivity in the place of the ASIC



Figure 4.24: (a) Elevation plane image of the 3-needle phantom recorded at 1000 volumes/s with the prototype. (b) Rendered 3D image of the same recording.

before the measurement. A total dynamic range of 91 dB is recorded between the 0-dB SNR point of the highest and the 1-dB compression point of the lowest gain setting.

An overview of the measurement setup used for imaging experiments is given in Fig. 4.21(c). A phantom of three needles with a total spacing of 7 mm is immersed in a watertank about 13 mm from an acoustic window used to interface with a daughter board carrying a chip with co-integrated transducers. Cables with a length of 1 m connect the PCB assembly to an FPGA board that receives the high-bandwidth LVDS data via a motherboard and forwards it to a measurement PC that performs the image reconstruction based on conventional delay-and-sum operations. In seven transmit/receive cycles, the ASIC first excites 6-MHz pressure waves with 65-V pulses and then records generated echoes from the phantom with the seven subarray beamformer settings. Fig. 4.24 shows the resulting image once in an elevation plane in (a) and once as a rendered 3D image in (b). While the aperture is too small to provide the resolution of a full array, the needle heads can clearly be distinguished in 3D space, demonstrating the functionality of the prototype.

A summary of the system characteristics and comparison to the prior art in catheter-based ultrasound systems [6], [15], [32], [33], [40] is provided in Table 4.2. This work describes the first design to integrate element-level HV transmitters and analog frontends, subarray beamforming, and in-probe digitization in a scalable fashion for 3D imaging. A dedicated architecture enables the highest reported

|                          | This Work                        | JSSC'21 [32]          | JSSC'20 [40]          | VLSI'19 [33]          | JSSC'18 [15]                     | TUFFC'16 [6]       |
|--------------------------|----------------------------------|-----------------------|-----------------------|-----------------------|----------------------------------|--------------------|
| Technology               | 180nm BCD                        | 180nm                 | 180nm BCD             | 180nm                 | 180nm                            | N/A                |
| Transducer               | 2D PZT                           | 2D PMUT               | 1D CMUT               | 2D PZT                | 2D PZT                           | 2D PZT             |
| Array Size               | 8 × 9                            | 6 × 6                 | 64                    | $4 \times 4$          | 6 × 24                           | 60 × 14            |
| Integrated<br>Transducer | $\checkmark$                     | ×                     | $\checkmark$          | $\checkmark$          | $\checkmark$                     | $\checkmark$       |
| Center<br>Frequency      | 6 MHz                            | 5 MHz                 | 7 MHz                 | 5 MHz                 | 5 MHz                            | 5.6 MHz            |
| Pitch-<br>matched        | $\checkmark$                     | ×                     | $\checkmark$          | $\checkmark$          | $\checkmark$                     | $\checkmark$       |
| Element<br>Pitch         | 160 μm ×<br>160 μm               | 250 μm ×<br>250 μm    | 205 μm ×<br>1800 μm   | 150 μm ×<br>150 μm    | 150 μm ×<br>150 μm               | 110 μm ×<br>180 μm |
| Integrated<br>TX         | $\checkmark$                     | $\checkmark$          | $\checkmark$          | x                     | x                                | $\checkmark$       |
| Max. TX<br>Voltage       | 65 V                             | 13.2 V                | 60 V                  | N/A                   | N/A                              | 40 V               |
| Digitization             | $\checkmark$                     | $\checkmark$          | X                     | $\checkmark$          | $\checkmark$                     | X                  |
| Receiver<br>Architecture | AFE + µBF<br>+ ADC +<br>Datalink | AFE + ADC             | AFE                   | AFE + ADC             | AFE + μBF<br>+ ADC +<br>Datalink | AFE + µBF          |
| Channel Reduction        | 12-fold                          | N/A                   | N/A                   | N/A                   | 36-fold                          | 15 to 20-fold      |
| Supported<br>Frame Rate  | 1000 vol/s                       | N/A                   | N/A                   | N/A                   | 200 vol/s                        | 50 vol/s           |
| Active Area<br>/ Element | 0.032 mm <sup>2 §</sup>          | 0.063 mm <sup>2</sup> | 0.464 mm <sup>2</sup> | 0.023 mm <sup>2</sup> | 0.026 mm <sup>2 §</sup>          | N/A                |
| RX power<br>/ Element    | 1.23 mW §                        | 1.14 mW               | 5.2 mW                | 1.54 mW               | 0.91 mW §                        | < 0.12 mW          |
| Input DR                 | 91 dB                            | N/A                   | 82 dB                 | N/A                   | 85 dB                            | N/A                |
| Peak SNR                 | 52.3 dB                          | 57.8 dB ‡             | N/A                   | 49.8 dB               | 52.8 dB                          | N/A                |

TABLE 4.2: COMPARISON WITH THE PRIOR ART IN CATHETER-BASED ULTRASOUND IMAGERS

\* Transducers on separate board connected with wires.

<sup>†</sup> Scalability limited by transducer connection outside of pitch.

§ Including the Datalink and LVDS drivers.

<sup>‡</sup> ADC only, excluding AFE.

frame rate with a channel-count reduction sufficient to enable an array with less than 100 data channels in the catheter when scaled to the full size.

# 4.5. Conclusion

A transceiver ASIC combining HV transmission with subarray beamforming and in-probe digitization for catheter-based 3D ultrasound probes has been presented. A pitch-matched design is facilitated by an area- and power-efficient hybrid beamforming ADC tailored to the application in large imaging arrays. The novel architecture enables a high frame rate of 1000 volumes/s while also providing sufficient data channel reduction through subarray beamforming and time division multiplexing. A prototype with a co-integrated transducer matrix has been manufactured and successfully applied in a 3D imaging experiment. Together with competitive power consumption and large dynamic range, the system is a promising solution for future miniature ultrasound probes.

# References

- Z. M. Hijazi, K. Shivkumar, and D. J. Sahn, "Intracardiac Echocardiography (ICE) During Interventional & Electrophysiological Cardiac Catheterization," *Circulation*, vol. 119, no. 4, pp. 587–596, Feb. 2009. DOI: 10.1161/ CIRCULATIONAHA.107.753046.
- [2] G. Perk, R. M. Lang, M. A. Garcia-Fernandez, et al., "Use of Real Time Three-Dimensional Transesophageal Echocardiography in Intracardiac Catheter Based Interventions," *Journal of the American Society of Echocardiography*, vol. 22, no. 8, pp. 865–882, Aug. 2009. DOI: https://doi.org/10.1016/j. echo.2009.04.031.
- [3] B. Luani, T. Rauwolf, C. Genz, A. Schmeißer, M. Wiemer, and R. Braun-Dullaeus, "Intracardiac echocardiography versus fluoroscopy for endovascular and endocardial catheter navigation during cryo-ablation of the slow pathway in AVNRT patients," *Cardiovasc Ultrasound*, vol. 17, no. 12, Jun. 2019. DOI: 10.1186/s12947-019-0162-2.
- [4] S. S. Kim, Z. M. Hijazi, R. M. Lang, and B. P. Knight, "The Use of Intracardiac Echocardiography and Other Intracardiac Imaging Tools to Guide
Noncoronary Cardiac Interventions," Journal of the American College of Cardiology, vol. 53, no. 23, pp. 2117–2128, Jun. 2009. DOI: https://doi.org/10.1016/j.jacc.2009.01.071.

- [5] A. Enriquez, L. C. Saenz, R. Rosso, et al., "Use of Intracardiac Echocardiography in Interventional Cardiology: Working With the Anatomy Rather Than Fighting It," *Circulation*, vol. 137, no. 21, pp. 2278–2294, May 2018. DOI: 10.1161/CIRCULATIONAHA.117.031343.
- [6] D. Wildes, W. Lee, B. Haider, et al., "4-D ICE: A 2-D Array Transducer With Integrated ASIC in a 10-Fr Catheter for Real-Time 3-D Intracardiac Echocardiography," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 63, no. 12, pp. 2159–2173, Oct. 2016. DOI: 10.1109/ TUFFC.2016.2615602.
- [7] C. Knackstedt, A. Franke, K. Mischke, et al., "Semi-automated 3-dimensional intracardiac echocardiography: development and initial clinical experience of a new system to guide ablation procedures," *Heart Rhythm*, vol. 3, no. 12, pp. 1453–1459, Dec. 2006. DOI: 10.1016/j.hrthm.2006.05.026.
- [8] W. Lee, W. Griffin, D. Wildes, et al., "A 10-Fr ultrasound catheter with integrated micromotor for4-D intracardiac echocardiography," *IEEE Trans*actions on Ultrasonics, Ferroelectrics, and Frequency Control, vol. 58, no. 7, pp. 1478–1491, Jul. 2011. DOI: 10.1109/TUFFC.2011.1967.
- [9] W. T. Wilser, S. R. Barnes, and L. Garbini, "Helical acoustic array for medical ultrasound," U.S. Patent US8449467B2, May 2013.
- [10] G. Gurun, C. Tekes, J. Zahorian, et al., "Single-chip CMUT-on-CMOS frontend system for real-time volumetric IVUS and ICE imaging," *IEEE Trans*actions on Ultrasonics, Ferroelectrics, and Frequency Control, vol. 61, no. 2, pp. 239–250, Feb. 2014. DOI: 10.1109/TUFFC.2014.6722610.
- [11] J. W. Choe, O. Oralkan, A. Nikoozadeh, et al., "Volumetric real-time imaging using a CMUT ring array," *IEEE Transactions on Ultrasonics, Ferroelectrics,* and Frequency Control, vol. 59, no. 6, pp. 1201–1211, Jun. 2012. DOI: 10. 1109/TUFFC.2012.2310.
- [12] E. D. Light, S. F. Idriss, P. D. Wolf, and S. W. Smith, "Real-time threedimensional intracardiac echocardiography," Ultrasound in Medicine & Biology, vol. 27, no. 9, pp. 1177–1183, Sep. 2001. DOI: https://doi.org/10. 1016/S0301-5629(01)00421-5.

- [13] W. Lee, S. Idriss, P. Wolf, and S. Smith, "A miniaturized catheter 2-D array for real-time, 3-D intracardiac echocardiography," *IEEE Transactions on Ul*trasonics, Ferroelectrics, and Frequency Control, vol. 51, no. 10, pp. 1334– 1346, Oct. 2004. DOI: 10.1109/TUFFC.2004.1350962.
- [14] D. E. Dausch, K. H. Gilchrist, J. B. Carlson, S. D. Hall, J. B. Castellucci, and O. T. von Ramm, "In vivo real-time 3-D intracardiac echo using PMUT arrays," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 61, no. 10, pp. 1754–1764, Oct. 2014. DOI: 10.1109/TUFFC. 2014.006452.
- [15] C. Chen, Z. Chen, D. Bera, et al., "A Pitch-Matched Front-End ASIC With Integrated Subarray Beamforming ADC for Miniature 3-D Ultrasound Probes," *IEEE Journal of Solid-State Circuits*, vol. 53, no. 11, pp. 3050–3064, Sep. 2018. DOI: 10.1109/JSSC.2018.2864295.
- B. Savord and R. Solomon, "Fully sampled matrix transducer for real time 3D ultrasonic imaging," in *IEEE Symposium on Ultrasonics*, 2003, Oct. 2003, pp. 945–953. DOI: 10.1109/ULTSYM.2003.1293556.
- [17] M. Soozande, B. W. Ossenkoppele, Y. Hopf, et al., "Imaging Scheme for 3-D High-Frame-Rate Intracardiac Echography: A Simulation Study," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 69, no. 10, pp. 2862–2874, Oct. 2022. DOI: 10.1109/TUFFC.2022.3186487.
- [18] P. Santos, G. U. Haugen, L. Løvstakken, E. Samset, and J. D'hooge, "Diverging Wave Volumetric Imaging Using Subaperture Beamforming," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 63, no. 12, pp. 2114–2124, Dec. 2016. DOI: 10.1109/TUFFC.2016.2616172.
- [19] O. Villemain, J. Baranger, M. K. Friedberg, et al., "Ultrafast Ultrasound Imaging in Pediatric and Adult Cardiology: Techniques, Applications, and Perspectives," JACC: Cardiovascular Imaging, vol. 13, no. 8, pp. 1771–1791, Aug. 2008. DOI: https://doi.org/10.1016/j.jcmg.2019.09.019.
- [20] C. Daft, P. Wagner, S. Panda, and I. Ladabaum, "Two Approaches to Electronically Scanned 3D Imaging Using cMUTs," in 2006 IEEE Ultrasonics Symposium, Oct. 2006, pp. 685–688. DOI: 10.1109/ULTSYM.2006.185.
- [21] T. M. Carpenter, M. W. Rashid, M. Ghovanloo, D. M. J. Cowell, S. Freear, and F. L. Degertekin, "Direct Digital Demultiplexing of Analog TDM Signals for Cable Reduction in Ultrasound Imaging Catheters," *IEEE Transac-*

tions on Ultrasonics, Ferroelectrics, and Frequency Control, vol. 63, no. 8, pp. 1078–1085, Aug. 2016. DOI: 10.1109/TUFFC.2016.2557622.

- [22] M. W. Rashid, C. Tekes, M. Ghovanloo, and F. L. Degertekin, "Design of frequency-division multiplexing front-end receiver electronics for CMUT-on-CMOS based intracardiac echocardiography," in 2014 IEEE International Ultrasonics Symposium, Sep. 2014, pp. 1540–1543. DOI: 10.1109/ULTSYM. 2014.0381.
- [23] Q. Liu, C. Chen, Z.-y. Chang, C. Prins, and M. A. P. Pertijs, "A mixed-signal multiplexing system for cable-count reduction in ultrasound probes," in 2015 *IEEE International Ultrasonics Symposium (IUS)*, Oct. 2015, pp. 1–4. DOI: 10.1109/ULTSYM.2015.0141.
- [24] G. Pilikos, L. Horchens, J. Batenburg, T. van Leeuwen, and F. Lucka, "Deep data compression for approximate ultrasonic image formation," in 2020 IEEE International Ultrasonics Symposium, Sep. 2020. DOI: 10.1109/IUS46767. 2020.9251753.
- [25] N. Shlezinger, A. Amar, B. Luijten, R. J. G. van Sloun, and Y. C. Eldar, Deep Task-Based Analog-to-Digital Conversion, Jan. 2022. DOI: 10.48550/ ARXIV.2201.12634.
- [26] B. Lam, M. Price, and A. P. Chandrakasan, "An ASIC for Energy-Scalable, Low-Power Digital Ultrasound Beamforming," in 2016 IEEE International Workshop on Signal Processing Systems (SiPS), Dec. 2016, pp. 57–62. DOI: 10.1109/SiPS.2016.18.
- [27] M.-C. Chen, A. Peña Perez, S.-R. Kothapalli, et al., "A Pixel Pitch-Matched Ultrasound Receiver for 3-D Photoacoustic Imaging With Integrated Delta-Sigma Beamformer in 28-nm UTBB FD-SOI," *IEEE Journal of Solid-State Circuits*, vol. 52, no. 11, pp. 2843–2856, Nov. 2017. DOI: 10.1109/JSSC. 2017.2749425.
- [28] Y.-J. Kim, S.-E. Cho, J.-Y. Um, et al., "A Single-Chip 64-Channel Ultrasound RX-Beamformer Including Analog Front-End and an LUT for Non-Uniform ADC-Sample-Clock Generation," *IEEE Transactions on Biomedical Circuits and Systems*, vol. 11, no. 1, pp. 87–97, Feb. 2017. DOI: 10.1109/ TBCAS.2016.2571739.

- [29] M. D'Urbino, C. Chen, Z. Chen, et al., "An Element-Matched Electromechanical ΔΣ ADC for Ultrasound Imaging," *IEEE Journal of Solid-State Circuits*, vol. 53, no. 10, pp. 2795–2805, Oct. 2018. DOI: 10.1109/JSSC. 2018.2859961.
- [30] J.-Y. Um, Y.-J. Kim, S.-E. Cho, et al., "An Analog-Digital Hybrid RX Beamformer Chip With Non-Uniform Sampling for Ultrasound Medical Imaging With 2D CMUT Array," *IEEE Transactions on Biomedical Circuits and* Systems, vol. 8, no. 6, pp. 799–809, Dec. 2014. DOI: 10.1109/TBCAS.2014. 2375958.
- [31] T. Kim, S. Shin, and S. Kim, "An 80.2 dB DR 23.25 mW/Channel 8-Channel Ultrasound Receiver With a Beamforming Embedded SAR ADC," *IEEE Transactions on Circuits and Systems II: Express Briefs*, vol. 66, no. 9, pp. 1487–1491, Sep. 2019. DOI: 10.1109/TCSII.2018.2889810.
- [32] J. Lee, K.-R. Lee, B. E. Eovino, et al., "A 36-Channel Auto-Calibrated Front-End ASIC for a pMUT-Based Miniaturized 3-D Ultrasound System," *IEEE Journal of Solid-State Circuits*, vol. 56, no. 6, pp. 1910–1923, Jan. 2021. DOI: 10.1109/JSSC.2021.3049560.
- [33] J. Li, Z. Chen, M. Tan, et al., "A 1.54mW/Element 150µm-Pitch-Matched Receiver ASIC with Element-Level SAR/Shared-Single-Slope Hybrid ADCs for Miniature 3D Ultrasound Probes," in 2019 Symposium on VLSI Circuits, Aug. 2019, pp. C220–C221. DOI: 10.23919/VLSIC.2019.8778200.
- [34] Y. M. Hopf, B. Ossenkoppele, M. Soozande, et al., "A Pitch-Matched ASIC with Integrated 65V TX and Shared Hybrid Beamforming ADC for Catheter-Based High-Frame-Rate 3D Ultrasound Probes," in 2022 IEEE International Solid- State Circuits Conference (ISSCC), vol. 65, Mar. 2022, pp. 494–496. DOI: 10.1109/ISSCC42614.2022.9731597.
- [35] Y. M. Hopf, B. Ossenkoppele, M. Soozande, et al., "A Compact Integrated High-Voltage Pulser Insensitive to Supply Transients for 3-D Miniature Ultrasound Probes," *IEEE Solid-State Circuits Letters*, vol. 5, pp. 166–169, Jun. 2022. DOI: 10.1109/LSSC.2022.3180071.
- [36] S. Bae and T.-K. Song, "Methods for Grating Lobe Suppression in Ultrasound Plane Wave Imaging," *Applied Sciences*, vol. 8, no. 10, p. 1881, Oct. 2018. DOI: 10.3390/app8101881.

- [37] V. Kumar, P. Lee, B. Kim, M. Fatemi, and A. Alizad, "Gap-filling method for suppressing grating lobes in ultrasound imaging: experimental study with deep-learning approach," *IEEE Access*, pp. 76276–76286, Apr. 2020. DOI: 10.1109/access.2020.2989337.
- [38] D. Perdios, M. Vonlanthen, F. Martinez, M. Arditi, and J.-P. Thiran, "CNN-Based Ultrasound Image Reconstruction for Ultrafast Displacement Tracking," *IEEE Transactions on Medical Imaging*, vol. 40, no. 3, pp. 1078–1089, Mar. 2021. DOI: 10.1109/TMI.2020.3046700.
- [39] T. Szabo, Diagnostic ultrasound imaging: inside out, 2nd ed. Oxford, UK: Academic Press, 2014.
- [40] E. Kang, M. Tan, J.-S. An, et al., "A Variable-Gain Low-Noise Transimpedance Amplifier for Miniature Ultrasound Probes," *IEEE Journal of Solid-State Circuits*, vol. 55, no. 12, pp. 3157–3168, Dec. 2020. DOI: 10.1109/JSSC. 2020.3023618.
- [41] C. Chen, S. B. Raghunathan, Z. Yu, et al., "A Prototype PZT Matrix Transducer With Low-Power Integrated Receive ASIC for 3-D Transesophageal Echocardiography," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 63, no. 1, pp. 47–59, Jan. 2016. DOI: 10.1109/TUFFC. 2015.2496580.
- [42] C. Chen, Z. Chen, Z.-y. Chang, and M. A. P. Pertijs, "A compact 0.135mW/channel LNA array for piezoelectric ultrasound transducers," in ESS-CIRC Conference 2015 - 41st European Solid-State Circuits Conference (ES-SCIRC), Nov. 2015, pp. 404–407. DOI: 10.1109/ESSCIRC.2015.7313913.
- [43] K. A. Ng and Y. P. Xu, "A Compact, Low Input Capacitance Neural Recording Amplifier," *IEEE Transactions on Biomedical Circuits and Systems*, vol. 7, no. 5, pp. 610–620, Oct. 2013. DOI: 10.1109/TBCAS.2013.2280066.
- [44] B. Steinberg, "Digital beamforming in ultrasound," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 39, no. 6, pp. 716–721, Nov. 1992. DOI: 10.1109/58.165556.
- [45] B. Murmann. "ADC Performance Survey 1997-2021." (2022), [Online]. Available: http://web.stanford.edu/~murmann/adcsurvey.html (visited on 02/06/2022).

- [46] T. Rabuske and J. Fernandes, "A SAR ADC With a MOSCAP-DAC," *IEEE Journal of Solid-State Circuits*, vol. 51, no. 6, pp. 1410–1422, Jun. 2016. DOI: 10.1109/JSSC.2016.2548486.
- [47] C.-C. Liu, M.-C. Huang, and Y.-H. Tu, "A 12 bit 100 MS/s SAR-Assisted Digital-Slope ADC," *IEEE Journal of Solid-State Circuits*, vol. 51, no. 12, pp. 2941–2950, Dec. 2016. DOI: 10.1109/JSSC.2016.2591822.
- [48] C. Chen, Z. Chen, D. Bera, et al., "A Front-End ASIC With Receive Subarray Beamforming Integrated With a 32×32 PZT Matrix Transducer for 3-D Transesophageal Echocardiography," *IEEE Journal of Solid-State Circuits*, vol. 52, no. 4, pp. 994–1006, Apr. 2017. DOI: 10.1109/JSSC.2016.2638433.
- [49] A. X. Widmer and P. A. Franaszek, "A DC-Balanced, Partitioned-Block, 8B/10B Transmission Code," *IBM Journal of Research and Development*, vol. 27, no. 5, pp. 440–451, Sep. 1983. DOI: 10.1147/rd.275.0440.
- [50] H.-H. Chang, J.-W. Lin, C.-Y. Yang, and S.-I. Liu, "A wide-range delaylocked loop with a fixed latency of one clock cycle," *IEEE Journal of Solid-State Circuits*, vol. 37, no. 8, pp. 1021–1027, Aug. 2002. DOI: 10.1109/JSSC. 2002.800922.
- [51] M. van Elzakker, E. van Tuijl, P. Geraedts, D. Schinkel, E. A. M. Klumperink, and B. Nauta, "A 10-bit Charge-Redistribution ADC Consuming 1.9 μW at 1 MS/s," *IEEE Journal of Solid-State Circuits*, vol. 45, no. 5, pp. 1007–1015, May 2010. DOI: 10.1109/JSSC.2010.2043893.
- [52] M. Miyahara, Y. Asada, D. Paik, and A. Matsuzawa, "A low-noise selfcalibrating dynamic comparator for high-speed ADCs," in 2008 IEEE Asian Solid-State Circuits Conference, Dec. 2008, pp. 269–272. DOI: 10.1109/ ASSCC.2008.4708780.
- [53] B. Malki, T. Yamamoto, B. Verbruggen, P. Wambacq, and J. Craninckx, "A 70 dB DR 10 b 0-to-80 MS/s Current-Integrating SAR ADC With Adaptive Dynamic Range," *IEEE Journal of Solid-State Circuits*, vol. 49, no. 5, pp. 1173–1183, May 2014. DOI: 10.1109/JSSC.2014.2309086.
- [54] Cyclone V device datasheet, Intel, Santa Clara, CA, USA, 2019.
- [55] V309 datasheet, Olympus Corporation, Tokyo, Japan, 2014.
- [56] Cyclone V device datasheet, Precision Acoustics Ltd., Dorchester, United Kingdom, 2013.

# 5

# A HIGH-FRAME-RATE ULTRASOUND ASIC WITH MULTI-LEVEL LOAD MODULATION DATALINK

This chapter is based on the publication "A Pitch-Matched High-Frame-Rate Ultrasound Imaging ASIC for Catheter-Based 3D Probes" in preparation for publication in the IEEE Journal of Solid-State Circuits.

## 5.1. Introduction

U ltrasound imaging is a popular tool in medical treatments due to its relatively safe nature, cost-effectiveness, and compatibility with minimally invasive interventions [1], [2]. A special class of imaging devices used for the latter are catheter-based probes. These enable high-resolution images taken directly next to the area of interest in the body. The probes are disposable and purpose-built for their application in procedures like intravascular ultrasound (IVUS) [3] or intracardiac echocardiography (ICE) [4], [5]. While IVUS is commonly used in procedures such as plaque detection in the vascular system and can typically work with a

smaller imaging array [6]–[8], ICE is applied in a variety of cardiovascular interventions with more demanding requirements on the imaging depth and resolution [9], [10].

Particularly in upcoming 3D probes with 2D transducer arrays, these requirements lead to significant challenges in the design of ICE catheters. While 2D imagers with 1D transducer arrays commonly apply passive probes with direct element connection to the imaging system [11], [12], the higher cable count, crosstalk and attenuation across thinner cables for 2D transducer matrices lead to the application of ASICs in ultrasound imaging catheters [9], [13]. The basic functionality of the electronics includes ultrasound transmission (TX) and reception (RX) on each array element to maximize the imaging aperture in the limited space of the catheter, amplification of the received echoes for robust signal transmission, and communication with an imaging system outside the body [14], [15]. But as for 3D imaging, the transducer pitch in the azimuthal and elevation direction has to be designed for low impact on image quality [10], there is little space for circuitry that is matched to the pitch of the transducer. A pitch-matched design is a requirement for scalability to the about 1000 elements of a full 3D ICE design [9], [10] but leads to a high requirement on the integration density of the circuits. Prior 3D ICE designs have thus been limited to a subset of the desirable functionality as they could not integrate transmit beamforming [15], only low-voltage transmission [16], no transmit functionality at all [17], [18], or had to strongly limit the achievable volume acquisition rate [9].

The biggest challenge in the transition to 3D probes is posed by the communication with the imaging system. Catheters with a diameter of around 3 mm have to accommodate all TX and RX channel signals next to common connections like power and additional controls [9]. To reduce the number of TX cables, pulses are often generated on the ASIC. The configuration data can be provided via a serial link into local registers [19]–[21] or an efficient shift register (SR) [14], [22] before the next TX phase and delays for TX beamforming can be generated with local counters. However, this leads to large registers and counter cells for a dense array with a large number of delay steps. An alternative is to implement TX control in a row/column (R/C) approach [23]–[26]. This minimizes the amount of associated circuitry underneath the element but, in turn, limits the amount of possible TX patterns for application in a 3D ICE probe. A similar challenge is faced when externally generated pulses are only passed or blocked on the chip instead of local pulse generation [8], [25], [27]. Reduction of RX cables in the catheter has previously been achieved by multiplexing of transducer signals on fewer connections over multiple transmit/receive (T/R) cycles [27], [28]. To avoid the associated loss in frame rate, other designs have made use of the wider bandwidth of the channel compared to the imaging frequency. In these, several RX signals are multiplexed on a single connection within one T/R cycle [29]–[31]. While if applied for sensitive analog signals, the channel-to-channel crosstalk can become an issue [32], particularly for digital signals, this can be done with a low impact on the signal integrity [15]. Nevertheless, the limited bandwidth across thin cables in the catheter restricts the maximum cable-count reduction achievable with this method in both cases.

Another approach is to apply subarray beamforming, also known as microbeamforming  $(\mu BF)$  [33]. This shifts part of the RX beamforming into the catheter by delaying and summing the received signals of a subarray of elements. However, it comes at the cost of less raw data being available in the final image reconstruction and reduced frame rate as multiple acquisitions are required per reconstructed volume [34]. While the former limits the effectiveness of imaging algorithms, the latter can preclude the use of modes like high-frame-rate blood flow or electromechanical wave imaging [35]. These disadvantages can be mitigated while achieving significant cable-count reduction by a combination of subarray beamforming and digital time-division multiplexing. This has been shown with element-level digitization with subsequent digital beamforming [36], [37], digitization of the output of analog beamformers [15], [18], [38], and mixed schemes [39]. However, even more can be gained from the digital transmission as ultrasound imaging can tolerate higher bit error rates (BER) than most communication links [40]. While current digital probe designs commonly rely on conventional low-voltage differential signaling (LVDS) [15], [17], [18], [41] with relatively low BER, BER could potentially be traded to reduce the circuit area and power consumption. Moreover, it could allow for a lower cable count by dividing the total output data bandwidth across fewer channels with a higher transmission rate.

An additional aspect to consider in the design of the amplification for large transducer arrays is the common-mode interference across the channels. Each individual frontend amplifier actually has a reduced noise requirement before it limits the global performance as the transducer noise increases with decreasing size [42], [43]. The uncorrelated noise can then still reveal weaker signals after beamforming in the imaging system with an SNR gain of  $\sqrt{N}$  for N combined signals [44]. As more elements are summed for a smaller pitch, the final signal-to-noise ratio

is, to first order, equal for different element sizes in a transducer array of a given size [45]. But correlated noise between channels effectively reduces the maximum gain achievable through beamforming, meaning it should remain well below the noise floor despite the larger number of channels. This challenging requirement of ultrasound frontend design has received little attention in the scientific literature.

In this chapter, a pitch-matched ASIC with a co-integrated transducer array with a pitch of 160  $\mu$ m × 160  $\mu$ m is presented. The design builds on the foundation laid in [15] with a 4x larger array and novel circuitry to form a comprehensive ultrasound imager architecture. On-chip TX control is implemented by a compact combination of an R/C and an SR approach that offers all required beam patterns. Additional area savings are achieved by encoding the delay of the TX beamformer as the difference from its neighbor, similar to [21], and re-using the shift-register cells as counters for delay generation. Local biasing schemes for the transducer and analog frontend (AFE) amplifiers enable quick settling of the input after the TX phase and a lower impact of correlated noise on the final image. Additionally, a multi-level pulse amplitude modulation (PAM) channel combined with TDM and  $\mu$ BF achieves an RX cable-count reduction of 18 while still enabling a high frame rate of 1000 volumes/s. To reduce the heating of tissue in the patient, the driver is implemented as a load modulation (LM) architecture, reducing the power consumption of the channel at the tip of the catheter.

This chapter is organized as follows. The architecture and system design considerations are described in Section 5.2. Section 5.3. provides circuit implementation details on the TX beamformer, analog frontend, and LM data transmission. The fabricated prototype, measurement setup, and results are discussed in Section 5.4. The chapter concludes with a comparison to the prior art and a conclusion.

### 5.2. System Design

#### 5.2.1. Overview

Fig. 5.1(a) provides an overview of the developed system. A matrix of  $16 \times 18$  transducer elements is designed for directly interfacing the presented ASIC as shown in Fig. 5.1(b). The array is 4x larger than that presented in chapter 4 and equivalent to a quarter of the aperture of the envisioned final ICE probe. The 160µm-pitch transducer stack is a revised version of the concept shown in [46]. The circuitry interfaces with the PZT elements through individual connections made



Figure 5.1: (a) Conceptional drawing of transducer stack. (b) Overview of chip assembly with details on the matrix organization.

with a gold contact on the bottom side and a common aluminum ground foil on the top.

On the side of the chip, implementation in a 0.18  $\mu$ m BCD technology offers a trade-off between the ability to integrate high-voltage (HV) TX-related structures and low-voltage (LV) logic as well as RX-related structures. Each element is connected to individual TX and RX paths to efficiently use the space inside the catheter. An isolating T/R switch between the two paths protects the receiver from HV breakdown during transmission. The receiver also implements time gain compensation (TGC) to manage the dynamic range (DR) requirement of ICE. While imaging is often done with a range of only 40 dB, ultrasound attenuation inside the human body can lead to a total DR in the order of 100 dB within one T/R cycle [15]. But as the attenuation is time-dependent, TGC can still reduce the DR to the following components by complementing it with a time-varying gain [13]. The matrix is further grouped into subarrays of three elements each and subgroups of two subarrays each. To enable a frontend layout matched to

the pitch of the transducer element, all backend circuitry, such as the datalinks and data output drivers, are pushed to the periphery. This enables the design of subgroups as unit cells that can be replicated to create a larger aperture. The periphery-level circuitry is designed in units as well and only occupies two sides, the top serving the upper and the bottom serving the lower half of the matrix, to allow scaling along one direction.

This design serves as a prototype to evaluate the architecture and shown techniques for application in a full ICE probe. While all of the functionality required for the realization of the  $64 \times 18$ -element imaging scheme presented in chapter 2 is included, only one-fourth of the aperture in the azimuth direction is accommodated in this step. To characterize the performance of the 1D subarray beamformer in the elevation direction and the imaging approach, the full aperture is implemented along that axis. The beamformer of just three elements enables a high frame rate as only few acquisitions are required per volume. The full probe targets a 10-cm imaging depth and thus operates at a pulse-repetition-frequency (PRF) of 7.7 kHz considering the speed of sound in human tissue. Being able to image with just seven fan-shaped beams per frame, this results in a total frame rate of up to 1000 volumes/s. The seven steps employ a TX beam with 10.7° divergence in the elevation and 70° divergence in the azimuth direction accordingly to achieve a field of view of  $70^{\circ} \times 70^{\circ} \times 10$  cm [10].

#### 5.2.2. Architecture

Fig. 5.2 shows the architecture implemented in the presented design. The topology is building on what has been discussed in chapter 4 and the TX part is based on the unipolar pulser with embedded T/R switch introduced in chapter 3. In this design, control of the TX beamforming (TX BF) is realized with per-element digital delay cells that are configured in a mixed scheme: While global steering information is provided on the level of rows or columns of elements to save area, individual control is supplied by a shift register spanning the whole matrix. To minimize the number of connections inside the catheter, the shift register is also used to load the row and column data as well as all global configuration settings. The entire shift register content of about 1.5 kb can be loaded in around 15  $\mu$ s at a clock frequency of 100 MHz. The SR is updated during the RX period and thus forms an upper limit to the PRF at about 66 kHz, irrespective of the imaging depth.



Figure 5.2: Architecture overview showing how element-level circuitry, subarray beamformers, and a shared ADC structure per subgroup are combined. The arrangement of the shift register and row/column approach is sketched and a global view of the data management and clocking is provided.

On the receiver side, each element is individually connected to a low-noise amplifier (LNA), followed by a second-stage programmable gain amplifier (PGA) that

also acts as a single-ended-to-differential converter. The LNA can be discretely switched in steps of 18 dB from -12 to 24 dB and the second stage can be configured in steps of 6 dB from 6 dB to 24 dB. Together, they implement TGC with a range of 54 dB in 10 steps of 6 dB from -6 dB to 48 dB. The LNA architecture is a modified version of the design in [47] to enable operation with HV transmitters and seamless gain switching during echo reception. As the received signal is most sensitive at this point, a local regulator for the LNA supplies is installed on the subgroup level here as well to reduce the effect of common interference across the whole array. The PGA is based on [48] and [18], again adapted to get full TGC range within one T/R cycle and thus full frame rate.

The digitization from chapter 4 is adopted with the same parameters in this design since both implementations target the same application. Three elements are combined in the elevation direction for analog subarray beamforming in the charge domain. Two neighboring subarrays are combined for layout with a shared hybrid beamforming analog-to-digital converter (ADC) in a subgroup. The ADC combines a successive approximation register (SAR) first stage and single-slope (SS) second stage and operates at 24 MSps per channel with a resolution of 10 bits. To allocate all hardware in the core, the SAR and SS outputs are individually transferred to the periphery.

On the periphery, two datalink configurations have been included. The regular one features recombination of the SAR and SS output to 10-bit words, 8b10b encoding [49], and serialization to conventional LVDS drivers. In the process, 4-fold time division multiplexing (TDM) is applied, leading to a rate of 1.2 Gbps per channel and, together with 3-fold subarray beamforming and considering the differential nature of the signals, a total cable count reduction factor of 6. The second setup only serves one channel and provides a parallel path from the recombination to a multi-level encoder and an LM driver. This enables conventional access via field-programmable gate arrays (FPGAs) [50] to the whole array for convenient imaging while providing an evaluation platform for the novel datalinks discussed in Section 5.3.4. The periphery and core are timed by shared dividers and delaylocked loops (DLLs) working on a 240-MHz clock provided by the FPGA.



Figure 5.3: (a)-(c) Examples of TX beamformer delay patterns with arrows indicating the delay propagation for a  $6 \times 6$  array: Centered diverging wave, diagonal plane wave & focused wave. (d) Decision table for the delay propagation based on shift register and row/column input.

# 5.3. Circuit Implementation

#### 5.3.1. TX Beamformer

To achieve a compact TX BF implementation, the system operates by passing a single trigger signal through the entire array based on relative delays between neighboring elements. For further area savings while still enabling the creation of all required beam patterns, control of the propagation direction is done from the R/C level but also with one local bit from the SR,  $CTRL_{loc}$ . Fig. 5.3 displays examples of TX patterns that can be generated with the presented method. The targeted imaging scheme [10] relies on diverging waves that propagate delay from the center outwards as shown in (a). But it is also possible to create other common patterns, such as the angled plane waves in (b) or focused waves in (c). The graphs also demonstrate how these patterns are created by indicating the starting position and row/column control with arrows along the side of the matrix. Arrows in the matrix are pointing from the element that the trigger signal was received from and Fig. 5.3(d) presents the complete element-level logic table.

The circuit of an SR BF cell is shown in Fig. 5.4. The local delay value is



Figure 5.4: Circuit details of a TX beamformer cell with additional local pulse disabling in dashed lines.

loaded into three flip-flops (FFs) as part of the SR with the SR input,  $IN_{SR}$ , and clock,  $CLK_{SR}$ . As the delay information is only required during the TX phase, loading of the delay cells in the SR is done during RX without additional hold cells. Moreover, an area-efficient implementation is reached by repurposing the unused SR cells to counters during TX based on the T/R control signal,  $EN_{RX}$ . The reconfiguration can be done by switching the FF data inputs to an inverting feedback loop around the cell and using the data output of lower-order bits as the clock input of higher-order bits, leading to only a few switches as overhead instead of more FFs. The number of delay steps equals  $2^N$ , N being the number of counter cells, and the actual delay is determined by the frequency of CLK<sub>SR</sub> during TX as it remains connected to the least-significant bit (LSB), b<sub>2</sub>. In this design, the 100-MHz SR clock is maintained during TX and there are three counter cells per element, leading to a total of eight steps from 0 ns to 70 ns.

The counter rotation is converted to an enable signal for a latch,  $EN_L$ , by detecting its highest value with a negative-AND (NAND) gate. Including an inverted version of  $CLK_{SR}$  can block any glitches in the asynchronous counter and clearing the latch during RX with  $EN_{RX}$  prevents any TX triggers in the wrong phase. Selection of the latch input is made with a multiplexer (MUX) controlled by the R/C signals and the output of the last element-level SR cell that serves as  $CTRL_{loc}$  but also as the output to the SR of the next element,  $OUT_{SR}$ . Five different input triggers can be selected, LEFT to BOT indicating the output triggers. Fig 5.5 shows a timing diagram with an example of loading the element SR during RX



Figure 5.5: Timing diagram showing the SR loading of a TX beamformer cell, subsequent pulsing based on the delay from the START trigger, and further propagation to the cell on the left.

before delaying the external trigger by two steps and even later having the left neighbor pass it further. In the transition between RX and TX,  $CLK_{SR}$  is held high such that the counter maintains its state. If it were low, the FFs would toggle depending on their current status. As the states are still unique in that case, the circuit can still be operated but an extra mapping between codes is required.

The dashed circuitry is optional and has been included in this ASIC to support additional debugging by excluding single elements from the entire imaging operation. The implementation adds one bit in the local SR to receive a disable signal, DIS, and latches it such that it is also available during the RX phase. By not directly using  $OUT_{TRIG}$  to control the pulser but manipulating it with a negative-OR (NOR) gate to  $OUT_{TX}$ , elements can be permanently connected to the HV supply. At the same time, the rest of the array can function as usual as the trigger is still being passed.

Next to a compact design, TX beamforming by shifting an LV trigger signal through the system also has the benefit of being able to dynamically change the number of pulses and even pulse frequency if required. One potential drawback for some applications is that only delays smaller than the pulse width can be generated as  $\text{EN}_{\text{L}}$  needs to latch during the active interval. However, this is commonly no issue for ultrasound imaging as is analyzed in the following. Typical ultrasound imager arrays remain at a pitch of half of the center transmit wavelength,  $\lambda$ , as that moves grating lobes out of the picture [51]. As the relative difference in distance, D, from a virtual source between neighboring elements can at most be equal to the element pitch, it can be shown that the maximum possible delay for these devices is half of the pulse period and thus in the coverable range:



Figure 5.6: Drawing showing the difference D in excitation of two elements with a certain pitch in order to create a steered diverging wave.

$$DEL = \frac{D}{c} = \frac{\frac{\lambda}{2}}{\lambda \cdot f} = \frac{T}{2}$$
(5.1)

where DEL is the required delay between elements, c is the speed of sound, f is the center frequency, and T is the pulse period. A way to determine the delay between neighbors for any pitch is shown using the diagram in Fig. 5.6. The relative distance from the virtual source, VS, can be calculated with

$$D = \frac{\sin(\alpha)}{\cos(\beta)} \cdot pitch \tag{5.2}$$

where  $\alpha$  is a known angle determining beam steering and  $\beta$  is a known angle determining beam divergence. The equation is found by applying the law of sines on the lower triangle based on angles  $\delta$  and  $\varepsilon$ , found with trigonometric equations, and can be used to derive the required delay by DEL = D/c. Although the pitch is slightly above half of  $\lambda$ , this results in maximum delays in the order of 70% of the 80-ns pulse width for the shown design with the intended 10° divergence and maximum steering angle of 30° [10].

#### 5.3.2. LNA

The LNA architecture is based on the design presented in [34] but has been extended to enable co-integration with TX and high-frame-rate operation. Fig. 5.7(a) shows a simplified schematic of the prior design with a direct connection of the input node,  $V_{IN}$ , to the transducer element and input biasing to  $V_{ref}$  through a high-impedance path,  $R_{IN}$ . A compact implementation is reached by a split capacitor feedback network [52], an efficient inverter-based amplifier core, and dynamic



Figure 5.7: (a) Simplified schematic of an inverter-based amplifier with split capacitor feedback and dynamic biasing. (b) Illustration of how the initial transducer state and forward biasing diodes due to large inputs can lead to offset issues with dynamic biasing.

biasing to potentials  $V_{biasx}$ . Discrete TGC is achieved by reconfiguration of the capacitor network comprised of  $C_{IN}$  and  $C_{FB}$  between receive cycles.

While the shown techniques have led to a receiver with state-of-the-art power and area efficiency, they also lead to issues in the transition to a high-frame-rate imaging system with TX and RX on each element. To illustrate this, Fig. 5.7(b) shows two common problems that can be experienced in a transducer frontend with incoming pressure waves, P, and dynamic biasing. To achieve quick settling following TX, the dynamic biasing is synchronized to the T/R cycle and active until shortly after TX is completed. The first issue is that, since echoes start returning immediately after the ultrasound transmission, the switching can never be guaranteed to take place at a moment in which no signal is present. As pulsing and dynamic biasing in a short time window require a reference with a relatively low impedance, the final DC operating point will be signal-dependent and can lead to a large offset. The second issue is that strong incoming signals, which often occur shortly after TX due to low initial attenuation in the medium, can lead to clipping due to the electrostatic discharge (ESD) protection diodes or the internal rails. This would change the DC operating point even during active RX, with no possibility to recover in a pure dynamic biasing implementation.

Irrespective of these external effects, the limitation to biasing in a short time interval before reception also leads to issues if one would switch through all gain levels of the discrete TGC scheme due to non-ideal switching procedures. Finally, the transducer needs to be biased to a reference,  $V_{ref}$ , being e.g. just the direct current (DC) level of a bipolar pulser [14] or a mid-rail reference if the pulsing goes all the way to the negative rail [53]. Since this would need to happen in a short



Figure 5.8: Conceptual schematic of the inverter-based low noise amplifier (LNA) showing gain switching, transducer biasing, and supply regulators.

time interval after TX, it cannot be covered by the high-impedance connection with  $R_{IN}$  for a design with TX and RX on the same element.

Fig. 5.8 shows the implemented design with proposed changes. The LNA is still applying dynamic biasing to quickly settle all critical nodes in a short time window after the possibly high disturbance of HV TX. To target the issues discussed above, this is assisted by the connection of the transducer to a mid-rail potential,  $V_{CM}$ , via resistors. The switches are controlled with RST<sub>LNA</sub> and RST<sub>LNA-DEL</sub> to always close at the beginning of an RX period. R1 is in the order of 50 k $\Omega$  and used to raise the transducer top plate from its initial state of 0 V after pulsing to  $V_{CM}$ in a time frame of 100s of ns to not lose too much RX time but also not cause a second transmission. R2 is around 1 M $\Omega$  and connected longer, in the order of  $\mu$ s, to settle any offset due to large inputs received shortly after TX but then also disconnected as it is a potential source of noise and interference from other channels.

Next to the added resistive components, the biasing has also been modified compared to [34] by adjusting the way in which the inverter transistors are operated. During initial dynamic biasing, the upper transistor is still directly connected to a biasing voltage,  $V_{biasp}$ , and the lower transistor is still driven by an amplifier in a feedback loop to force the LNA output voltage,  $V_{OUT}$ , to a local mid-rail potential,  $V_{MID}$ . However, this amplifier is not switched out and shorted after that but instead connected to two capacitors in parallel with the feedback capacitors of the LNA. This gives the structure a low-frequency path to correct for effects like disturbances from gain switching or residual issues in the transducer biasing. Due to the low bandwidth of the additional amplifier and the capacitive attenuation towards the output, its noise contribution in the signal bandwidth as well as the power and area are negligible compared to the entire LNA.

The capacitor ratios and unit cells, C, of 45 fF are as in [34]. The input capacitors are held at the same potential the transducer is biased to,  $V_{CM}$ , before being switched in to minimize introduced disturbance in the RX period. Similarly, the feedback capacitors are connected to the same potential  $V_{out}$  is driven to via the low-bandwidth loop after being switched out. After these capacitors are settled, there is also the option of removing them from  $V_{MID}$  in order to remove the noise from that source.

To mitigate the low power supply rejection ratio (PSRR) of the inverter-based topology [54], the LNA employs the local regulators [55] as in [34]. In this design, they are shared on the subgroup level and it can be chosen to switch to the analog ground, VSSA, instead of the negative low-dropout regulator (LDO) to investigate differences. Additionally, the dynamically set references of both LDOs are capacitively coupled to the ground foil node of the transducer,  $V_{GF}$ . This enables the architecture to also reject variations of the ground foil node that would otherwise be amplified with the signal. Combined with the improved PSRR, this aims to reduce the common mode interference across channels and thus maximize the possible gain from RX beamforming.

#### 5.3.3. PGA

The circuit implementation of the PGA is based on the compact architecture shown in [18] but modified as the original structure can't support high-frame-rate operation and has a limited output swing. The main issue with volume acquisition at a high rate is presented following Fig. 5.9. The prior design is similar to Fig. 5.9(a) and achieves gain switching by adding a capacitor to the virtual ground of an amplifier in a capacitive feedback configuration. While the constant load presented



Figure 5.9: Impact of switching the gain of an amplifier in a dynamically biased capacitive feedback in active operation from: (a) the back; (b) the front.

to the driving stage of this setup is beneficial, this switching alters the operating point of the setup and leads to an offset if done during active operation. As the system is dynamically biased with  $V_{\rm bias}$ , it can not recover over time and even with a low-bandwidth correction path, the possibly large step could lead to a significant loss of image information.

With the approach illustrated in Fig. 5.9(b), on the other hand, this can be avoided since the virtual ground node is not affected when the capacitor is switched from the input. The input experiences a short settling depending on the driver but then continues operation without an offset. In the proposed design, the configuration of Fig. 5.9(b) is adopted to enable gain switching during echo reception and thus high-frame-rate. Fig. 5.10(a) shows the schematic implementation with a compact T-type feedback [48] and Fig. 5.10(b) presents the associated logic table to determine the gain depending on the switch settings. To minimize the disturbance from gain switching, the input capacitors are connected to the output DC level of the LNA and the feedback capacitors are connected to a mid-rail potential,  $V_{CM}$ , when not in the loop. The LNA has sufficient phase margin and bandwidth to operate with the switching capacitance at its output and achieve quick settling.

Similar to [18], the structure is held in reset during TX to enable quick settling of the operating point after HV pulsing, but in addition to that, a low-bandwidth feedback path has been introduced. This enables the correction of any remaining disturbance caused by switching during one T/R cycle while adding little area and power consumption compared to that of the entire PGA. Particularly the switching of the T-type network can lead to non-ideal behavior as the capacitors can not be switched to a known reference, as is the case for the input capacitors. The feedback amplifier is implemented as a differential gain stage with a bandwidth of



Figure 5.10: Sketch of the programmable gain amplifier (PGA) displaying gain switching and the implementation of a low-bandwidth feedback amplifier.

around 1 MHz. As it is not part of determining the gain, the capacitors of the lowbandwidth path can be adjusted in a trade-off between attenuation of additional noise and achievable correction range. In this design, the unit capacitance, C, is 33 fF and the additional capacitors are 9 fF to arrive at a noise contribution more than 3 times lower than that of the main amplifier in the signal range.

The configuration as a single-ended-to-differential converter from the prior design is kept to enable differential operation of the ADC and dummy switches are placed in the bottom branch of the feedback network as well to balance the design. However, instead of connecting to a mid-rail supply, the bottom branch is driven relative to the reference of the single-ended input. This ensures that no interferers are unintentionally amplified with the received signal and is realized by mirroring the negative rail connection of the LNA. The telescopic amplifier has moreover been replaced by a two-stage miller-compensated differential amplifier with an auxiliary amplifier to provide common-mode feedback. The division in a first gain stage and a high-swing output stage can show bandwidth and power consumption similar to [18] while providing a larger output swing for the ADC and not leading to gain deviation.



Figure 5.11: (a) Block diagram summarizing the idea of load modulation. (b) Example of a singleended load modulation implementation. (c) Example of a differential load modulation implementation.

#### 5.3.4. Datalink

To explore possibilities to reduce the RX-related cables in the catheter, this design leverages the relatively relaxed BER requirement of ultrasound imaging [40]. By applying a multi-level PAM driver, several bits are transmitted in one symbol, allowing for a higher throughput without increasing the symbol rate which would cause limitations due to the low bandwidth of the thin cables [56]. To reduce the power consumption associated with data transmission at the tip of the catheter, where heating must be limited to ensure safe operation in the human body [57], the driver has moreover been implemented based on load modulation (LM) [27]. Instead of direct driving, this only switches a variable load impedance,  $Z_{var}$ , on the chip with a driver on the other end of the cable as shown in Fig. 5.11(a). This enables measurements at an evaluation point outside the body with part of the power consumption shifted to the system side where heating is not an issue.

In this design, single-ended and differential multi-level LM implementations are investigated. Fig. 5.11(b) gives an overview of the single-ended setup with the variable load being implemented by  $2^N - 1$  transistors with scaled widths  $M_x$ , Nbeing the number of bits per symbol. The array is controlled with a thermometer code in SE<sub>x</sub> and forms a divider with the termination resistance to a power supply,  $V_{LM}$ , on the system side. The differential configuration in Fig. 5.11(c) relies on



Figure 5.12: (a) Overview of the data and control configuration for the multi-level signaling. (b) Details of a single-ended implementation with two bits per symbol.

the same mechanism but introduces a second, inverted path with the same scaling via a second cable. While a differential channel takes an additional cable, more power consumption, and circuit area, the gained rejection of common interference compared to the single-ended structure can be beneficial in a narrow catheter with multiple parallel connections. Additionally, the shared return current in the ASIC will be constant to first order, reducing possible cross-talk between channels, and the differential signal swing is twice as high for the same supply. This potentially enables a higher bandwidth per channel at the same BER as the single-ended setup and makes differential signaling an interesting part of this study.

Fig. 5.12(a) gives an overview of the realized test setups in this design and their control. The 60 recombined data bits from six ADC pairs operating at 48 MHz are multiplexed into 12-bit words at 240 MHz. Based on this input, three different configurations are being analyzed in parallel: a differential driver with 2 bits per symbol at 1.44 GSps, a differential driver with 4 bits per symbol at 0.72 GSps, and two single-ended drivers with 2 bits per symbol at 0.72 GSps that share the total data input. All of these achieve the targeted RX data cable reduction



Figure 5.13: Micrograph of a chip with transducer stack on its surface and an inset showing the global floorplan below.

of 18 to ultimately arrive at a total of 64 data connections for the full probe [10] and enable the direct comparison of all schemes. The timing is provided by a DLL that creates six phases out of the input clock to accommodate the highest transmit frequency.

The internal structure of the control blocks is shown by the example of a singleended cell in Fig. 5.12(b). After synchronization of the recombined and multiplexed ADC outputs to the input clock, an array of buffers is used to drive two outputs, b1 and b2. The buffers can be disabled to have a high output impedance and effectively multiplex onto one output based on the overlap of phases from the DLL. Local generation of the overlap avoids the need for distributing a high-frequency clock and is done with one layer of logic gates. For the single-ended setup, only every second DLL output phase is required to realize the 0.72-GSps output stream. In the last stage, the binary signal is converted to a thermometer representation for the LM driver based on logic gates.

The system also provides a second path with a pseudo-random bit sequence (PRBS) generator to be able to investigate the BER. It is implemented as a linear-feedback shift register (LFSR) of 20 cells with 12-bit parallel output [58].

# 5.4. Experimental Results

#### 5.4.1. Electrical Measurements

An overview of the manufactured device is given in Fig. 5.13. The design implements one-fourth of the final intended aperture in the azimuth direction and the

5

124



Figure 5.14: Measurement of the transmit delay between two neighboring elements with configurations from the minimum to the maximum setting.

full aperture in the elevation direction. The ASIC is fabricated in a 180-nm HV BCD technology and a transducer stack with 160-µm pitch is manufactured on its surface. In the inset, the global floorplan is showing a separation in two halves with datalinks on the top and bottom, enabling further scaling of the unit-cell-based structure along the horizontal axis.

Fig. 5.14 shows the electrical characterization of the transmit beamformer. Two neighboring elements are each connected with a capacitor to mimic the capacitive load of a single transducer element at about 2 pF. The delay between them is swept through all the configurations from zero to seven unit time periods, T. The interval is determined by the clock frequency of the time reference and in this case 10 is ns.

To investigate the implemented scheme for common-mode interference rejection around the first-stage amplifier, an experiment with an intentionally injected interferer is carried out. Fig. 5.15 displays the root mean square (RMS) ADC output caused by an amplitude sweep of a 6-MHz sinusoidal signal driven onto the ground foil node with the LNA input being connected to it in the highest gain setting. To study the effects, the two configurations of the negative supply connection of the LNA, either to the shared analog ground node or the local negative supply regulator, are compared. The larger ground foil interference required to elevate the output code from the noise floor in the case of LNA connection to the local regulator shows the effectiveness of the suppression.

The settling of biasing points in the presence of input signals is studied with



Figure 5.15: Measured ADC output relative to the amplitude of an interfering signal applied to the ground foil node in the case of low-noise amplifier connection to the shared analog ground or local negative supply regulator.



Figure 5.16: (a) Initial converter output code settling for an exponentially decaying sinusoidal input with 3 different phases with indicated amplifier biasing times. (b) Converter output code showing settling during a full transmit/receive cycle with time gain compensation and an exponentially decaying sinusoidal input.

a 2-pF capacitor, modeling the capacitive component of the transducer element, in series with an arbitrary waveform generator (AWG). Fig. 5.16(a) shows the ADC output at the beginning of a receive phase with three different phases of an exponentially decaying 1-MHz sinusoidal input in the lowest gain setting. The signal is active before the dynamic biasing is completed and shows the response



Figure 5.17: Eye diagrams of all load modulation datalink configurations with a  $V_{LM}$  of 0.7 V, data rates of 2.88 Gbps, and a 1-m assembly of 42-AWG micro-coax cables.

to different initial states of the high-impedance input node when actual reception is started. Typically,  $RST_{PGA}$  would be asserted longer than  $RST_{LNA}$  to optimize settling but it is in this case disabled earlier as it would otherwise mask the output signals needed to study the behavior. In this example,  $RST_{PGA}$  is active 500 ns after the start of input signaling with  $RST_{LNA}$  and  $RST_{LNA-DEL}$  being asserted 500 ns and 5 µs longer respectively. The recorded outputs show a signal-dependent offset introduced after the completion of dynamic biasing and subsequent recovery provided by the resistive input biasing and the low-bandwidth amplifier feedback paths. Fig. 5.16(b) additionally shows the ADC output during a full T/R cycle with an exponentially decaying 6-MHz input from an AWG. TGC is used from the lowest to the highest gain setting to maintain the output code level and it can be observed how non-ideal steps in the gain are compensated by the implemented biasing techniques.

Fig. 5.17 shows the eye diagrams of all load modulation datalink configurations as measured based on the output of the PRBS generator. The data is acquired at a supply voltage,  $V_{LM}$ , of 0.7 V with a 1-m assembly of 42-AWG micro-coax cables. While the single-ended version has a clear opening, the differential topologies show more disturbance and even complete closure for the channel with 4 bits per symbol.



Figure 5.18: Display of the bit error rate (BER) vs. the supply voltage for the two load modulation configurations with 2 bits per symbol, offering a data rate of 2.88 Gbps.

The degradation in differential signaling can be explained by the transmission across two micro-coax cables and a significant improvement is expected if the arrangement is replaced by a twinax cable or a twisted pair.

A study of the BER in relation to the applied  $V_{LM}$  on the system side is shown in Fig. 5.18. The test data is generated by the on-chip PRBS generator, measured with an oscilloscope, and compared to the ideal values after reception at the measurement computer. While the differential channel with 4 bits per symbol is excluded due to insufficient performance, both channels with two bits per symbol show an improvement of the BER with increased supply voltage. The single-ended case clearly performs best with the applied cable assembly and can achieve a BER of  $10^{-6}$  at a V<sub>LM</sub> of about 0.7 V. As the degradation of B-mode ultrasound images is unnoticeable at a BER of  $10^{-6}$  [40], this implies that imaging is possible at a total measured power consumption of about 3.6 mW for the single-ended channel. Of that total, less than 2.1 mW is spent in the chip due to dissipation in the receiver termination of 50  $\Omega$  and the cable resistance of 9  $\Omega$ , resulting in an onchip energy consumption of about 1.5 pJ/bit. While an extended setup with dedicated receivers and cable assemblies in an arrayed fashion is required to study the approach in detail, this proof-of-concept shows the high potential of multi-level load modulation datalinks in ultrasound imaging probes.

#### 5.4.2. Acoustic Measurements

An experiment investigating the noise behavior of the array is presented in Fig. 5.19. The figure shows the RMS outputs of a sample with transducer elements while be-



Figure 5.19: Investigation of the noise output of the array relative to the number of combined channels with the low-noise amplifiers being connected to the shared analog ground or local negative supply regulators.

ing loaded by water and recording in the highest gain setting without an input signal. To study the effect on the noise floor, the experiment is conducted with the negative rail of the LNA being connected to the shared analog ground or local supply regulators. In both cases, there first is an exponential reduction in the amplitude before the two curves flatten off at different effective array sizes. The initial decline is close to the anticipated  $\sqrt{N}$  improvement of uncorrelated interference for N combined signals [44] but gets limited by the presence of correlated interference between channels. The reduction in common interference and thus improvement of potential beamforming gain for the LNA connection to the local regulator compared to the direct analog ground connection can be explained with an enhanced PSRR. Noise on the shared power rails appears common to all elements and can, even in this reduced-size prototype and without active input, pose a limit to the beamforming gain.

Characterization of the TX beamforming functionality is shown in Fig. 5.20, following the example patterns demonstrated in Fig. 5.3. The measurements are obtained with the chip assembly being submerged in water and facing a commercial hydrophone [59]. By translation of the hydrophone with an xyz-stage during repeated transmission from the prototype, C-planes in parallel to the transducer surface are captured. These evaluate the maximum peak pressure at each position and are compared to simulations [60] based on the same set of delays. Fig. 5.20(a) shows the simulated pressure profile at a 5-cm distance from the array for the transmission of a diverging wave on the left and the corresponding hydrophone measurement on the right. Fig. 5.20(b) similarly shows the results for a plane



Figure 5.20: Comparison of simulated C-planes on the left and corresponding measured C-planes on the right for: (a) A diverging beam at 5 cm from the array; (b) A plane wave steered to the north-east at 5 cm from the array; (c) A focused beam at 2 cm from the array.

wave angled at 10° to the north-east at a distance of 5 cm from the array and Fig. 5.20(c) shows a plane 2 cm from the array with the TX beamforming focusing at its center. Good agreement between the measured and simulated profiles in all cases verifies the capability of creating the most common patterns applied in ultrasound imaging.

To verify imaging functionality, a commercial 3D wire test phantom [61] has



Figure 5.21: (a) Measurement setup used for imaging verification with details of the 3D wire phantom. (b) Reconstructed image plane in the elevation direction with 3D render. (c) Reconstructed image plane in the azimuth direction with 3D render.

been studied with the setup shown in Fig. 5.21(a). The ASIC is mounted on a custom daughterboard that faces the surface of the phantom. The daughterboard provides local decoupling and connects to a motherboard with active components for supply regulation and signal conditioning via 1-m cables. The acquired receive data from the 24 LVDS channels is captured and stored by commercially-available FPGA boards [50] in real-time. Each FPGA board offers 8 high-speed transceivers and three boards are thus used in parallel, while one of them additionally acts as

a control host for the ASIC configuration per T/R cycle. The acquired data is downloaded to a computer that initiates measurement routines and performs image processing.

Fig. 5.21(b) shows a reconstructed image with the wires aligned in the azimuthal direction of the array, so that they show up as point scatterers in the elevation plane. The upper seven gain settings of the TGC have been applied in each T/R cycle to maintain a suitable dynamic range in the signal path across the attenuating phantom. A switching artifact can be observed when the LNA transitions to its highest gain setting while PGA gain steps are barely noticeable. The artifact is caused by non-ideal switching resulting in a step in the RX signal and appears at a 0° azimuth and elevation angle after RX beamforming as the entire array is switching simultaneously. Given its deterministic nature, the artifact doesn't inherently harm the imaging and can potentially be compensated for during data processing.

As off-axis reflections don't propagate back to the ASIC, the wire targets appear almost as points rather than lines in the 3D plot. The behavior along the other dimension is thus evaluated by rotating the chip to align with the azimuth direction as displayed in Fig. 5.21(c). The lower opening angle in the azimuth render is caused by the application of the ultimately targeted imaging scheme described in [10]. This scans a volume of  $70^{\circ} \times 70^{\circ} \times 10$  cm at 1000 volumes/s with seven fanshaped beams with little divergence in the elevation direction and large divergence in the azimuth direction. As this prototype implements the full intended aperture in the elevation direction, the functionality of the 1D subarray beamformer and TX beamforming can be fully verified. But as the aperture in the azimuth direction is 4x smaller, the prototype with an approximately square array only shows an opening angle close to that of one elevation fan beam of approximately 20° with this imaging scheme.

Table 5.1 summarizes the system and gives a comparison to the prior art in catheter-based ultrasound imagers. The system complements an earlier design described in chapter 4 with an integrated TX beamformer and a novel multi-level load modulation while still providing the highest reported frame rate. This completes the architecture and shows good performance in a 4x larger array. The transmit voltage was lowered to investigate if relatively rapid transducer degradation in the prior design could be caused by pulsers malfunctioning in acoustic samples due to latchup-like phenomena. The theory that a resonant transducer could cause sufficiently high substrate charge injection is supported by the effect

|                          | This work                        | JSSC'22 [15]                     | JSSC'21 [16]                     | VLSI'19 [17]          | JSSC'18 [18]                     | TUFFC'16 [9]       |
|--------------------------|----------------------------------|----------------------------------|----------------------------------|-----------------------|----------------------------------|--------------------|
| Technology               | 180nm BCD                        | 180nm BCD                        | 180nm                            | 180nm                 | 180nm                            | N/A                |
| Transducer               | 2D PZT                           | 2D PZT                           | 2D PMUT                          | 2D PZT                | 2D PZT                           | 2D PZT             |
| Array Size               | 16 x 18                          | 8 x 9                            | 6 x 6                            | 4 x 4                 | 6 x 24                           | 60 x 14            |
| Center<br>Frequency      | 6 MHz                            | 6 MHz                            | 5 MHz                            | 5 MHz                 | 5 MHz                            | 5.6 MHz            |
| Pitch-<br>Matched        | $\checkmark$                     | $\checkmark$                     | $oldsymbol{\varkappa}^{\dagger}$ | $\checkmark$          | $\checkmark$                     | $\checkmark$       |
| Element<br>Pitch         | 160 μm<br>x 160 μm               | 160 μm<br>x 160 μm               | 250 μm<br>x 250 μm               | 150 μm<br>x 150 μm    | 150 μm<br>x 150 μm               | 110 μm<br>x 180 μm |
| Included<br>Transmit     | 30 V                             | 65 V                             | 13.2 V                           | ×                     | ×                                | 40 V               |
| Transmit<br>Beamforming  | internal                         | external                         | internal                         | ×                     | ×                                | internal           |
| Receiver<br>Architecture | AFE + μBF<br>+ ADC +<br>Datalink | AFE + µBF<br>+ ADC +<br>Datalink | AFE + ADC                        | AFE + ADC             | AFE + µBF<br>+ ADC +<br>Datalink | $AFE + \mu BF$     |
| Digitization             | $\checkmark$                     | $\checkmark$                     | $\checkmark$                     | $\checkmark$          | $\checkmark$                     | X                  |
| Elements /<br>RX cable   | 18 <sup>§</sup>                  | 6                                | 0.5                              | 0.5                   | 18                               | 15 to 20           |
| Data<br>Transmission     | Multi-Level<br>LM §              | LVDS                             | N/A                              | LVDS                  | LVDS                             | analog             |
| Supported<br>Frame-Rate  | 1000 vol/s                       | 1000 vol/s                       | N/A                              | N/A                   | 200 vol/s                        | 50 vol/s           |
| Active Area /<br>Element | 0.030 mm <sup>2 §</sup>          | 0.032 mm <sup>2 §</sup>          | 0.063 mm <sup>2</sup>            | 0.023 mm <sup>2</sup> | 0.026 mm <sup>2 §</sup>          | N/A                |
| RX Power /<br>Element    | ~1.1 mW §                        | 1.23 mW §                        | 1.14 mW                          | 1.54 mW               | 0.91 mW §                        | < 0.12 mW          |
| Peak SNR                 | 52.2 dB                          | 52.3 dB                          | 57.8 dB ‡                        | 49.8 dB               | 52.8 dB                          | N/A                |

TABLE 5.1: COMPARISON WITH THE PRIOR ART IN CATHETER-BASED ULTRASOUND IMAGERS

<sup>†</sup> Scalability limited by a non-integrated transducer array and its connection outside of the pitch. <sup>§</sup> With the proposed Datalink and drivers. <sup>‡</sup> ADC only, excluding AFE.

not being visible in the presented second design, which trades maximum voltage for additional lateral isolation structures. A better trade-off could be achieved in a silicon-on-insulator (SOI) technology that does not rely on lateral spacing as much as the applied junction-isolated process used in this project.
## 5.5. Conclusion

A transceiver ASIC implementing a comprehensive architecture for catheter-based high-frame-rate 3D ultrasound imaging probes has been presented. A compact on-chip TX beamformer design that can provide all required beam patterns is achieved by the combination of an area-efficient row/column and a flexible shift register approach. The impact of common interference and settling in the analog frontend have been investigated and mitigation methods have been presented. The application of time-division multiplexing, subarray beamforming, and a load modulation, multi-level data transmission channel has led to a high cable-count reduction while reducing the power consumed on the chip and still offering a high frame rate. A prototype chip has been manufactured and successfully applied in a high-frame-rate, 3D imaging experiment to verify the functionality.

A more detailed characterization of the design is ongoing at the time of writing but outside the scope of this thesis.

## References

- S. M. Bierig and A. Jones, "Accuracy and Cost Comparison of Ultrasound Versus Alternative Imaging Modalities, Including CT, MR, PET, and Angiography," *Journal of Diagnostic Medical Sonography*, vol. 25, no. 3, pp. 138– 144, May 2009. DOI: 10.1177/8756479309336240.
- [2] Z. M. Hijazi, K. Shivkumar, and D. J. Sahn, "Intracardiac Echocardiography (ICE) During Interventional & Electrophysiological Cardiac Catheterization," *Circulation*, vol. 119, no. 4, pp. 587–596, Feb. 2009. DOI: 10.1161/ CIRCULATIONAHA.107.753046.
- [3] H. M. Garcia-Garcia, M. A. Costa, and P. W. Serruys, "Imaging of coronary atherosclerosis: intravascular ultrasound," *European Heart Journal*, vol. 31, no. 20, pp. 2456–2469, Sep. 2010. DOI: 10.1093/eurheartj/ehq280.

- [4] S. S. Kim, Z. M. Hijazi, R. M. Lang, and B. P. Knight, "The Use of Intracardiac Echocardiography and Other Intracardiac Imaging Tools to Guide Noncoronary Cardiac Interventions," *Journal of the American College of Cardiology*, vol. 53, no. 23, pp. 2117–2128, Jun. 2009. DOI: https://doi. org/10.1016/j.jacc.2009.01.071.
- [5] A. Enriquez, L. C. Saenz, R. Rosso, et al., "Use of Intracardiac Echocardiography in Interventional Cardiology: Working With the Anatomy Rather Than Fighting It," *Circulation*, vol. 137, no. 21, pp. 2278–2294, May 2018. DOI: 10.1161/CIRCULATIONAHA.117.031343.
- [6] G. Gurun, C. Tekes, J. Zahorian, et al., "Single-chip CMUT-on-CMOS frontend system for real-time volumetric IVUS and ICE imaging," *IEEE Trans*actions on Ultrasonics, Ferroelectrics, and Frequency Control, vol. 61, no. 2, pp. 239–250, Feb. 2014. DOI: 10.1109/TUFFC.2014.6722610.
- [7] J. Lim, C. Tekes, E. F. Arkan, A. Rezvanitabar, F. L. Degertekin, and M. Ghovanloo, "Highly Integrated Guidewire Ultrasound Imaging System-on-a-Chip," *IEEE Journal of Solid-State Circuits*, vol. 55, no. 5, pp. 1310–1323, Jan. 2020. DOI: 10.1109/JSSC.2020.2967551.
- [8] D. M. van Willigen, E. Kang, J. Janjic, et al., "A Transceiver ASIC for a Single-Cable 64-Element Intra-Vascular Ultrasound Probe," *IEEE Journal* of Solid-State Circuits, vol. 56, no. 10, pp. 3157–3166, Jun. 2021. DOI: 10. 1109/JSSC.2021.3083217.
- [9] D. Wildes, W. Lee, B. Haider, et al., "4-D ICE: A 2-D Array Transducer With Integrated ASIC in a 10-Fr Catheter for Real-Time 3-D Intracardiac Echocardiography," *IEEE Transactions on Ultrasonics, Ferroelectrics, and* Frequency Control, vol. 63, no. 12, pp. 2159–2173, Oct. 2016. DOI: 10.1109/ TUFFC.2016.2615602.
- [10] M. Soozande, B. W. Ossenkoppele, Y. Hopf, et al., "Imaging Scheme for 3-D High-Frame-Rate Intracardiac Echography: A Simulation Study," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 69, no. 10, pp. 2862–2874, Oct. 2022. DOI: 10.1109/TUFFC.2022.3186487.
- [11] ViewFlex Xtra ICE Brochure, St. Jude Medical, St. Paul, MN, USA, 2012.
- [12] T. Proulx, D. Tasker, and J. Bartlett-Roberto, "Advances in catheter-based ultrasound imaging Intracardiac Echocardiography and the ACUSON Acu-NavTM Ultrasound Catheter," in *IEEE Ultrasonics Symposium*, 2005., vol. 1, Mar. 2005, pp. 669–678. DOI: 10.1109/ULTSYM.2005.1602941.

- [13] E. Kang, M. Tan, J.-S. An, et al., "A Variable-Gain Low-Noise Transimpedance Amplifier for Miniature Ultrasound Probes," *IEEE Journal of Solid-State Circuits*, vol. 55, no. 12, pp. 3157–3168, Dec. 2020. DOI: 10.1109/JSSC. 2020.3023618.
- [14] M. Tan, E. Kang, J.-S. An, et al., "A 64-Channel Transmit Beamformer With ±30-V Bipolar High-Voltage Pulsers for Catheter-Based Ultrasound Probes," *IEEE Journal of Solid-State Circuits*, vol. 55, no. 7, pp. 1796–1806, Apr. 2020. DOI: 10.1109/JSSC.2020.2987719.
- [15] Y. M. Hopf, O. Boudewine, S. Mehdi, et al., "A Pitch-Matched Transceiver ASIC with Shared Hybrid Beamforming ADC for High-Frame-Rate 3D Intracardiac Echocardiography," *IEEE Journal of Solid-State Circuits*, vol. 57, no. 11, pp. 3228–3242, Nov. 2022. DOI: 10.1109/JSSC.2022.3201758.
- [16] J. Lee, K.-R. Lee, B. E. Eovino, et al., "A 36-Channel Auto-Calibrated Front-End ASIC for a pMUT-Based Miniaturized 3-D Ultrasound System," *IEEE Journal of Solid-State Circuits*, vol. 56, no. 6, pp. 1910–1923, Jan. 2021. DOI: 10.1109/JSSC.2021.3049560.
- [17] J. Li, Z. Chen, M. Tan, et al., "A 1.54mW/Element 150µm-Pitch-Matched Receiver ASIC with Element-Level SAR/Shared-Single-Slope Hybrid ADCs for Miniature 3D Ultrasound Probes," in 2019 Symposium on VLSI Circuits, Aug. 2019, pp. C220–C221. DOI: 10.23919/VLSIC.2019.8778200.
- [18] C. Chen, Z. Chen, D. Bera, et al., "A Pitch-Matched Front-End ASIC With Integrated Subarray Beamforming ADC for Miniature 3-D Ultrasound Probes," *IEEE Journal of Solid-State Circuits*, vol. 53, no. 11, pp. 3050–3064, Sep. 2018. DOI: 10.1109/JSSC.2018.2864295.
- [19] J. Kang, C. Yoon, J. Lee, et al., "A System-on-Chip Solution for Pointof-Care Ultrasound Imaging Systems: Architecture and ASIC Implementation," *IEEE Transactions on Biomedical Circuits and Systems*, vol. 10, no. 2, pp. 412–423, Apr. 2016. DOI: 10.1109/TBCAS.2015.2431272.
- [20] Y. Igarashi, S. Kajiyama, Y. Katsube, et al., "Single-Chip 3072-Element-Channel Transceiver/128-Subarray-Channel 2-D Array IC With Analog RX and All-Digital TX Beamformer for Echocardiography," *IEEE Journal of Solid-State Circuits*, vol. 54, no. 9, pp. 2555–2567, Sep. 2019. DOI: 10.1109/ JSSC.2019.2921697.

- [21] J. M. Rothberg, T. S. Ralston, A. G. Rothberg, et al., "Ultrasound-on-chip platform for medical imaging, analysis, and collective intelligence," Proceedings of the National Academy of Sciences, vol. 118, no. 27, Jul. 2021. DOI: 10.1073/pnas.2019339118.
- [22] G. Jung, C. Tekes, M. W. Rashid, et al., "A Reduced-Wire ICE Catheter ASIC With Tx Beamforming and Rx Time-Division Multiplexing," *IEEE Transactions on Biomedical Circuits and Systems*, vol. 12, no. 6, pp. 1246–1255, Dec. 2018. DOI: 10.1109/TECAS.2018.2881909.
- [23] K. Chen, H.-S. Lee, and C. G. Sodini, "A Column-Row-Parallel ASIC Architecture for 3-D Portable Medical Ultrasonic Imaging," *IEEE Journal of Solid-State Circuits*, vol. 51, no. 3, pp. 738–751, Mar. 2016. DOI: 10.1109/ JSSC.2015.2505714.
- [24] K. Chen, B. C. Lee, K. E. Thomenius, B. T. Khuri-Yakub, H.-S. Lee, and C. G. Sodini, "A Column-Row-Parallel Ultrasound Imaging Architecture for 3-D Plane-Wave Imaging and Tx Second-Order Harmonic Distortion Reduction," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 65, no. 5, pp. 828–843, May 2018. DOI: 10.1109/TUFFC.2018. 2811393.
- [25] E. Kang, Q. Ding, M. Shabanimotlagh, et al., "A Reconfigurable Ultrasound Transceiver ASIC With 24 × 40 Elements for 3-D Carotid Artery Imaging," *IEEE Journal of Solid-State Circuits*, vol. 53, no. 7, pp. 2065–2075, Jul. 2018. DOI: 10.1109/JSSC.2018.2820156.
- [26] P. Guo, F. Fool, E. Noothout, et al., "A 1.2mW/channel 100µm-Pitch-Matched Transceiver ASIC with Boxcar-Integration-Based RX Micro-Beamformer for High-Resolution 3D Ultrasound Imaging," in 2022 IEEE International Solid-State Circuits Conference (ISSCC), vol. 65, Feb. 2022, pp. 496–498. DOI: 10.1109/ISSCC42614.2022.9731784.
- [27] M. Tan, C. Chen, Z. Chen, et al., "A Front-End ASIC With High-Voltage Transmit Switching and Receive Digitization for 3-D Forward-Looking Intravascular Ultrasound Imaging," *IEEE Journal of Solid-State Circuits*, vol. 53, no. 8, pp. 2284–2297, Aug. 2018. DOI: 10.1109/JSSC.2018.2828826.
- [28] I. O. Wygant, X. Zhuang, D. T. Yeh, et al., "Integration of 2D CMUT arrays with front-end electronics for volumetric ultrasound imaging," *IEEE Trans*actions on Ultrasonics, Ferroelectrics, and Frequency Control, vol. 55, no. 2, pp. 327–342, Feb. 2008. DOI: 10.1109/TUFFC.2008.652.

- [29] C. Daft, P. Wagner, S. Panda, and I. Ladabaum, "Two Approaches to Electronically Scanned 3D Imaging Using cMUTs," in 2006 IEEE Ultrasonics Symposium, Oct. 2006, pp. 685–688. DOI: 10.1109/ULTSYM.2006.185.
- [30] M. W. Rashid, C. Tekes, M. Ghovanloo, and F. L. Degertekin, "Design of frequency-division multiplexing front-end receiver electronics for CMUT-on-CMOS based intracardiac echocardiography," in 2014 IEEE International Ultrasonics Symposium, Sep. 2014, pp. 1540–1543. DOI: 10.1109/ULTSYM. 2014.0381.
- [31] T. M. Carpenter, M. W. Rashid, M. Ghovanloo, D. M. J. Cowell, S. Freear, and F. L. Degertekin, "Direct Digital Demultiplexing of Analog TDM Signals for Cable Reduction in Ultrasound Imaging Catheters," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 63, no. 8, pp. 1078–1085, Aug. 2016. DOI: 10.1109/TUFFC.2016.2557622.
- [32] Q. Liu, C. Chen, Z.-y. Chang, C. Prins, and M. A. P. Pertijs, "A mixed-signal multiplexing system for cable-count reduction in ultrasound probes," in 2015 *IEEE International Ultrasonics Symposium (IUS)*, Oct. 2015, pp. 1–4. DOI: 10.1109/ULTSYM.2015.0141.
- B. Savord and R. Solomon, "Fully sampled matrix transducer for real time 3D ultrasonic imaging," in *IEEE Symposium on Ultrasonics*, 2003, Oct. 2003, pp. 945–953. DOI: 10.1109/ULTSYM.2003.1293556.
- [34] C. Chen, Z. Chen, D. Bera, et al., "A Front-End ASIC With Receive Subarray Beamforming Integrated With a 32×32 PZT Matrix Transducer for 3-D Transesophageal Echocardiography," *IEEE Journal of Solid-State Circuits*, vol. 52, no. 4, pp. 994–1006, Apr. 2017. DOI: 10.1109/JSSC.2016.2638433.
- [35] O. Villemain, J. Baranger, M. K. Friedberg, et al., "Ultrafast Ultrasound Imaging in Pediatric and Adult Cardiology: Techniques, Applications, and Perspectives," JACC: Cardiovascular Imaging, vol. 13, no. 8, pp. 1771–1791, Aug. 2008. DOI: https://doi.org/10.1016/j.jcmg.2019.09.019.
- [36] Y.-J. Kim, S.-E. Cho, J.-Y. Um, et al., "A Single-Chip 64-Channel Ultrasound RX-Beamformer Including Analog Front-End and an LUT for Non-Uniform ADC-Sample-Clock Generation," *IEEE Transactions on Biomedical Circuits and Systems*, vol. 11, no. 1, pp. 87–97, Feb. 2017. DOI: 10.1109/ TBCAS.2016.2571739.

- [37] M.-C. Chen, A. Peña Perez, S.-R. Kothapalli, et al., "A Pixel Pitch-Matched Ultrasound Receiver for 3-D Photoacoustic Imaging With Integrated Delta-Sigma Beamformer in 28-nm UTBB FD-SOI," *IEEE Journal of Solid-State Circuits*, vol. 52, no. 11, pp. 2843–2856, Nov. 2017. DOI: 10.1109/JSSC. 2017.2749425.
- [38] T. Kim, S. Shin, and S. Kim, "An 80.2 dB DR 23.25 mW/Channel 8-Channel Ultrasound Receiver With a Beamforming Embedded SAR ADC," *IEEE Transactions on Circuits and Systems II: Express Briefs*, vol. 66, no. 9, pp. 1487–1491, Sep. 2019. DOI: 10.1109/TCSII.2018.2889810.
- [39] J.-Y. Um, Y.-J. Kim, S.-E. Cho, et al., "An Analog-Digital Hybrid RX Beamformer Chip With Non-Uniform Sampling for Ultrasound Medical Imaging With 2D CMUT Array," *IEEE Transactions on Biomedical Circuits and* Systems, vol. 8, no. 6, pp. 799–809, Dec. 2014. DOI: 10.1109/TBCAS.2014. 2375958.
- [40] Z. Chen, M. Soozande, H. J. Vos, et al., "Impact of Bit Errors in Digitized RF Data on Ultrasound Image Quality," *IEEE Transactions on Ultrasonics*, *Ferroelectrics, and Frequency Control*, vol. 67, no. 1, pp. 13–24, Jan. 2020. DOI: 10.1109/TUFFC.2019.2937462.
- [41] M. D'Urbino, C. Chen, Z. Chen, et al., "An Element-Matched Electromechanical ΔΣ ADC for Ultrasound Imaging," *IEEE Journal of Solid-State Circuits*, vol. 53, no. 10, pp. 2795–2805, Oct. 2018. DOI: 10.1109/JSSC. 2018.2859961.
- [42] I. Ladabaum, X. Jin, H. Soh, A. Atalar, and B. Khuri-Yakub, "Surface micromachined capacitive ultrasonic transducers," *IEEE Transactions on Ul*trasonics, Ferroelectrics, and Frequency Control, vol. 45, no. 3, pp. 678–690, May 1998. DOI: 10.1109/58.677612.
- [43] A. A. Vives, "Fundamentals of Piezoelectricity," in *Piezoelectric Transducers and Applications*, 2nd ed., Springer Berlin, Heidelberg, 2009, ch. 1, pp. 1–38.
  DOI: https://doi.org/10.1007/978-3-540-77508-9.
- [44] K. Thomenius, "Evolution of ultrasound beamformers," in 1996 IEEE Ultrasonics Symposium. Proceedings, vol. 2, Nov. 1996, pp. 1615–1622. DOI: 10.1109/ULTSYM.1996.584398.

- [45] C. Oakley, "Calculation of ultrasonic transducer signal-to-noise ratios using the KLM model," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 44, no. 5, pp. 1018–1026, Sep. 1997. DOI: 10.1109/ 58.655627.
- [46] C. Chen, S. B. Raghunathan, Z. Yu, et al., "A Prototype PZT Matrix Transducer With Low-Power Integrated Receive ASIC for 3-D Transesophageal Echocardiography," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 63, no. 1, pp. 47–59, Jan. 2016. DOI: 10.1109/TUFFC. 2015.2496580.
- [47] C. Chen, Z. Chen, Z.-y. Chang, and M. A. P. Pertijs, "A compact 0.135mW/channel LNA array for piezoelectric ultrasound transducers," in ESS-CIRC Conference 2015 - 41st European Solid-State Circuits Conference (ES-SCIRC), Nov. 2015, pp. 404–407. DOI: 10.1109/ESSCIRC.2015.7313913.
- [48] K. A. Ng and Y. P. Xu, "A Compact, Low Input Capacitance Neural Recording Amplifier," *IEEE Transactions on Biomedical Circuits and Systems*, vol. 7, no. 5, pp. 610–620, Oct. 2013. DOI: 10.1109/TBCAS.2013.2280066.
- [49] A. X. Widmer and P. A. Franaszek, "A DC-Balanced, Partitioned-Block, 8B/10B Transmission Code," *IBM Journal of Research and Development*, vol. 27, no. 5, pp. 440–451, Sep. 1983. DOI: 10.1147/rd.275.0440.
- [50] Cyclone V device datasheet, Intel, Santa Clara, CA, USA, 2019.
- [51] T. Szabo, Diagnostic ultrasound imaging: inside out, 2nd ed. Oxford, UK: Academic Press, 2014.
- [52] M. S. Akter, R. Sehgal, F. van der Goes, K. A. A. Makinwa, and K. Bult, "A 66-dB SNDR Pipelined Split-ADC in 40-nm CMOS Using a Class-AB Residue Amplifier," *IEEE Journal of Solid-State Circuits*, vol. 53, no. 10, pp. 2939–2950, Oct. 2018. DOI: 10.1109/JSSC.2018.2859415.
- [53] M. Sautto, A. S. Savoia, F. Quaglia, G. Caliano, and A. Mazzanti, "A Comparative Analysis of CMUT Receiving Architectures for the Design Optimization of Integrated Transceiver Front Ends," *IEEE Transactions on Ul*trasonics, Ferroelectrics, and Frequency Control, vol. 64, no. 5, pp. 826–838, Feb. 2017. DOI: 10.1109/TUFFC.2017.2668769.
- [54] Y. Chae and G. Han, "Low Voltage, Low Power, Inverter-Based Switched-Capacitor Delta-Sigma Modulator," *IEEE Journal of Solid-State Circuits*, vol. 44, no. 2, pp. 458–472, Feb. 2009. DOI: 10.1109/JSSC.2008.2010973.

- [55] T. Christen, "A 15-bit 140-μ W Scalable-Bandwidth Inverter-Based ΔΣ Modulator for a MEMS Microphone With Digital Output," *IEEE Jour*nal of Solid-State Circuits, vol. 48, no. 7, pp. 1605–1614, Jul. 2013. DOI: 10.1109/JSSC.2013.2253232.
- [56] J. Stonick, G.-Y. Wei, J. Sonntag, and D. Weinlader, "An adaptive PAM-4 5-Gb/s backplane transceiver in 0.25-µm CMOS," *IEEE Journal of Solid-State Circuits*, vol. 38, no. 3, pp. 436–443, Mar. 2003. DOI: 10.1109/JSSC. 2002.808282.
- [57] Marketing Clearance of Diagnostic Ultrasound Systems and Transducers, U.S. Department of Health, Human Services, Food, and Drug Administration, Silver Spring, MD, USA, Jun. 2019.
- J. J. O'Reilly, "Series-parallel generation of m-sequences," Radio and Electronic Engineer, vol. 45, pp. 171–176, Apr. 1975. DOI: 10.1049/REE.1975. 0033.
- [59] Cyclone V device datasheet, Precision Acoustics Ltd., Dorchester, United Kingdom, 2013.
- [60] Michigan State University. "FOCUS." (), [Online]. Available: https://www. egr.msu.edu/~fultras-web/ (visited on 01/18/2023).
- [61] User Guide Three-Dimensional Wire Test Object Model 055A, Computerized Imaging Reference Systems, Inc., Norfolk, VA, USA, 2013.

# 6

# CONCLUSION

This thesis has presented the development of a high-frame-rate 3D intracardiac echocardiography (ICE) probe concept from the evaluation of the idea to a functional prototype. In this chapter, the main contributions of the work are summarized, followed by a compilation of general findings gained in the process. The chapter concludes with an outlook on future opportunities in the design of 3D ICE catheters but also in the development of ultrasound probes in general.

## 6.1. Main Contributions

Proposition of an imaging scheme and probe design for high-frame-rate 3D ICE catheters (chapter 2)

We have investigated the potential for 3D high-frame-rate intracardiac echocardiography and proposed an imaging scheme with an associated probe design that enables it. Restricting subarray beamforming to a small subarray in elevation direction allows for a volume acquisition with only 7 fan-shape beams for an imaging rate of 1000 volumes/s and a sufficient cable count reduction for the realization of the electronics. Essential parameters such as beam divergence, center frequency, and matrix dimensions have been optimized based on system-level simulations. Implementation of the HV ultrasound transmitter with the lowest HV transistor count (chapter 3)

It has been shown how a compact unipolar pulser with an embedded transmit/receive switch and a capacitive high-voltage (HV) level shifter can realize the ultrasonic HV transmit circuit with the lowest reported HV transistor count and area while not having any static power consumption. The topology is specifically optimized to be insensitive to transients on the HV supply as in probes with limited local supply decoupling, such as imaging catheters, pulsing can lead to disturbances on the supply rails. Pulsing with up to 65-V has been evaluated in electrical and acoustic tests with a 6-MHz center frequency PZT transducer array.

#### Implementation of an area- and power-efficient hybrid beamforming ADC for massively arrayed applications (chapter 4)

A hybrid analog-to-digital converter (ADC), combining an efficient charge-sharing successive-approximation-register (SAR) first stage and a compact single-slope (SS) second stage, has been developed to address power and area requirements in catheter-based probes. Application in large ultrasound imaging arrays is facilitated by directly interfacing the ADC with a charge-domain subarray beamformer and generating the SAR reference using a power-efficient local reference generator. Additional hardware-sharing between neighboring channels ultimately has led to the lowest reported area and power consumption across miniature ultrasound probe ADCs. The functionality and performance within the receive signal chain have been demonstrated in electrical experiments.

Realization of the first architecture to combine element-level HV pulsers and AFEs,  $\mu$ BF, ADCs, and datalinks for ultrasound imaging (chapter 4)

The first architecture to combine element-level HV transmitters and analog frontends (AFEs), subarray beamforming ( $\mu$ BF), in-probe digitization, and datalinks in a scalable fashion for catheter-based probes has been presented. A pitchmatched design is enabled by an efficient split between the core circuitry and a periphery block, the latter including a datalink performing clock-data-recovery (CDR) and time-division multiplexing (TDM), which leads to a 12-fold total channel-count reduction. The feasibility of the scheme has been demonstrated with an 8×9-element prototype applied in electrical, acoustic, and imaging experiments with up to 1000 volumes/s. Implementation of a fully functional pitch-matched transceiver ASIC for interconnect-limited ultrasound probes (chapter 5)

A  $16 \times 18$ -element ultrasound imager with a fully functional transceiver ASIC for catheter-based probes has been realized. Transmit beamforming is implemented on the chip with a combination of a shift register and a row/column approach to address the common interconnect challenge. Additional cable count and power reduction in the receiver part of the design have been explored with a multi-level load modulation datalink. It has further been discussed how common-mode interference can limit beamforming gain and a strategy to reduce its impact with local regulators has been shown. The setup has been characterized in electrical measurements and imaging experiments to verify high-frame-rate imaging capability.

## 6.2. General Findings

While pursuing the core target of a high-frame-rate 3D ICE concept, several interesting findings on the investigated subjects but also bordering matters have been made. In the following, the main findings are summarized:

- The main challenges in the design of high-frame-rate 3D ICE ASICs are power consumption, circuit area, and output data bandwidth. The first is mainly restricted due to the maximum permissible generation of heat in the human body while the second and third are strongly constrained by the available space inside intracardiac catheters. (chapter 1)
- In catheter-based ultrasound probes with limited local decoupling, large transient currents during high-voltage pulsing on an ASIC can lead to a significant disturbance on the HV supply due to inductance in the loop. The disturbance can cause capacitive HV level shifters without a high-side latch to lose their state and result in indeterministic pulser behavior. (chapter 3)
- With architectures and building blocks having been extensively explored and described, the application environment usually provides the main challenge for ADC design. In catheter-based ultrasound probes, the main complications are the HV co-design, massively arrayed structure, and power as well as area limitations. (chapter 4)
- A hybrid ADC combining a SAR and a SS stage can provide an efficient implementation and particularly lead to a small area for arrayed charge-sharing-

based topologies as no second digital-to-analog converter (DAC) bank is required. (chapter 4)

- Using a transducer during the transmit and the receive phase can maximize the aperture in area-constraint applications but requires a T/R isolation switch as well as a transducer biasing network during reception. (chapter 3 & 5)
- Dynamic biasing synchronized to the transmit/receive (T/R) cycles of a system can be an efficient approach for the design of sensitive circuitry in the presence of HV pulsing. The disturbance from nonideal switching during discrete time-gain compensation within one T/R cycle can be compensated for with an additional low-bandwidth biasing path. (chapter 4 & 5)
- To first order, the signal-to-noise ratio (SNR) of a single large transducer is equal to that of an array of multiple smaller transducers with the same total area after receive beamforming. While this means that in a receiver array, the power spent for reduction of the noise per path can be reduced relative to the single-element case, correlated noise between channels has to be carefully designed for. (chapter 4 & 5)
- Power consumption in the output data transmission is a significant part of the total budget in digital 3D ICE probes. As limited reduction can be achieved in the energy per transmitted bit, reduction in the transmitted data is the most effective method to limit its share. (chapter 4 & 5)
- The resonant behavior of transducers can lead to the forward biasing of p-n junctions in the DMOS devices of BCD technologies if connected on the transducer side. To prevent latch-up from occurring, guard rings and sufficient spacing should be considered. (chapter 5)
- The combination of a shift register and a row/column approach allows for a transmit beamformer implementation that is compact and reconfigurable between individual T/R cycles while still being flexible in the generation of transmit patterns. (chapter 5)
- In the design of a 3D ICE catheter essentially only the power spent at the probe tip is of interest to limit heating inside the patient. This makes load modulation an interesting technique for data transmission as part of the power consumption can be shifted to the system. Multi-level signaling can

additionally provide a way to trade a relatively relaxed bit error rate (BER) for more power savings. (chapter 5)

• The design of a system with the demonstrated complexity relies heavily on good models and computer-aided design (CAD) tools. These are quite mature for the bipolar-CMOS-DMOS (BCD) technology applied in this project but not as advanced for the transducer design and especially the transducer-to-ASIC interface. The project planning should take this into account.

### 6.3. Future Work

Transmit Integration and Functionality

Although the high-voltage transmitter discussed in chapter 3 features the smallest reported design with the least amount of HV transistors, it still occupies about a third of the total element area. This is mostly due to the large lateral dimensions of the junction isolation in the applied BCD technology. An alternative would be the implementation with deep trench isolation (DTI) or even in a silicon-on-insulator (SOI) technology. The dielectric applied in DTI achieves lateral isolation with much lower spacing requirements and combined with the vertical isolation of the insulator in an SOI wafer, full dielectric isolation can be provided [1].

This can either enable higher transmit voltages in the same area due to more relaxed voltage breakdown conditions or provide more space at the same transmit voltage to realize additional functionality. One possibility would be to adjust the pulser for bipolar operation to increase the energy efficiency [2]. Another option would be to reduce the power consumption and increase flexibility in the pulse shape by opting for a multi-level topology [3].

#### Impact of Image Processing on Probe Design

Despite the strong application-specific nature of integrated circuits designed for co-integration with transducers in ultrasound imaging probes, design choices in the application-specific integrated circuit (ASIC) are regularly still made on the block level. While this simplifies the procedure, the implications for the actual imaging are sometimes hard to estimate and wrong decisions or over-design can occur. In chapter 5, it was e.g. shown how, based on the results of a recent study on the impact of BER on ultrasound imaging quality [4], the number of cables and power consumption in an ICE probe could be significantly reduced compared to prior conventional data transmission approaches. Similar studies on a system or circuit model combined with an ultrasound simulation tool like Field II [5] could enable additional findings and assist in the architecture design.

An aspect that would be quite interesting in this context is the investigation of ADC design for ultrasound probes. One could e.g. study how parameters like its linearity affect the image quality in different operating modes. Another relevant topic is how a sparsity of ultrasound signals can be exploited to reduce output data bandwidth. This is already explored in processing [6], [7] but is yet to be efficiently integrated with existing technologies.

#### Full-aperture high-frame-rate 3D ICE

The ASIC shown in chapter 5 could be extended to the full aperture described in chapter 2. The current version implements the full size in elevation but only a fourth in azimuth direction although the core circuitry is matched to the pitch of the transducer elements and all circuitry is in principle designed in unit cells that can be replicated with a low threshold. The main reason for this is that the combined output data bandwidth in the order of 100 Gb/s would exceed the capabilities of the current measurement setup based on discrete components and general-purpose evaluation boards. In addition to that, the shown multi-level load modulation datalink cannot work with standard receivers available as e.g. interfaces of an FPGA or discrete components.

One option to handle this in a future design would be by co-design of an applicationspecific receiver chip. An array of high-speed, low-resolution ADCs could digitize the multi-level signals and the data could be parallelized into multiple conventional datalinks to an external memory at a lower speed.

#### Advanced Ultrasound Imaging System Assemblies

A large part of the area and power concerns in current ultrasound imaging catheter ASICs are caused by the limitation to planar structures and suboptimal technology options. Transmit-related circuitry would ideally be implemented in a dedicated process with voltage capabilities in the 100s of volts while especially the digital parts of the receive chain could be significantly more area and power efficient in more miniaturized technology nodes. But since ultrasound imaging requires HV transmission as well as subsequent echo reception and designs commonly combine all parts on one chip, both sides currently have to make compromises. One path to investigate for future designs would thus be the possibility of co-integration of multiple chips, each dedicated to its specific function, in one probe. The issues of high interconnect density for large 2D transducer matrices and lateral space limitation in narrow catheters could potentially be relieved by 3D stacking of ICs and vertical connection with through-silicon-vias (TSVs) [8].

In line with this, the design could be modified to be compatible with transducers based on microelectromechanical systems (MEMS), such as piezoelectric micromachined ultrasonic transducers (PMUTs) [9] or capacitive micromachined ultrasonic transducers (CMUTs) [10]. While the current assemblies with bulk-piezoelectric material are also vertically assembled, the process is quite labor-intensive [11] and MEMS-based technologies could provide a cheaper, more reliable alternative that would suit the application of disposable ICE catheters.

#### Optical Data Transmission Links

There could be an opportunity in the switch from conventionally applied microcoaxial cables to fewer optical fibers with a much larger associated bandwidth. In [12], a 25.8 Gb/s optical datalink for future application in IVUS probes has recently been demonstrated by Li *et al.* While there is still an integration challenge with the electronics and probe, it has thus been shown that the size constraints for the inclusion of a vertical-cavity surface-emitting laser (VCSEL) in catheterbased imaging probes are feasible. This could reduce the data cable count for the imaging scheme presented in chapter 2 to only four. Additionally, one could explore if instead of integrating a VCSEL in the catheter, it is also possible to just modulate an externally provided optical signal to save power at the tip of the probe [13], similar to the idea of electrical load modulation discussed in chapter 5.

## References

 P. Wessels, M. Swanenberg, H. van Zwol, et al., "Advanced BCD technology for automotive, audio and power applications," Solid-State Electronics, vol. 51, no. 2, pp. 195–211, Feb. 2007. DOI: https://doi.org/10.1016/j. sse.2007.01.019.

- [2] W. Qiu, Y. Yu, F. K. Tsang, and L. Sun, "A multifunctional, reconfigurable pulse generator for high-frequency ultrasound imaging," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 59, no. 7, pp. 1558–1567, Jul. 2012. DOI: 10.1109/TUFFC.2012.2355.
- [3] K. Chen, H.-S. Lee, A. P. Chandrakasan, and C. G. Sodini, "Ultrasonic Imaging Transceiver Design for CMUT: A Three-Level 30-Vpp Pulse-Shaping Pulser With Improved Efficiency and a Noise-Optimized Receiver," *IEEE Journal of Solid-State Circuits*, vol. 48, no. 11, pp. 2734–2745, Nov. 2013. DOI: 10.1109/JSSC.2013.2274895.
- [4] Z. Chen, M. Soozande, H. J. Vos, et al., "Impact of Bit Errors in Digitized RF Data on Ultrasound Image Quality," *IEEE Transactions on Ultrasonics*, *Ferroelectrics, and Frequency Control*, vol. 67, no. 1, pp. 13–24, Jan. 2020. DOI: 10.1109/TUFFC.2019.2937462.
- [5] J. Jensen, "Simulation of advanced ultrasound systems using Field II," in 2004 2nd IEEE International Symposium on Biomedical Imaging: Nano to Macro, vol. 1, Apr. 2004, pp. 636–639. DOI: 10.1109/ISBI.2004.1398618.
- [6] G. Pilikos, L. Horchens, J. Batenburg, T. van Leeuwen, and F. Lucka, "Deep data compression for approximate ultrasonic image formation," in 2020 IEEE International Ultrasonics Symposium, Sep. 2020. DOI: 10.1109/IUS46767. 2020.9251753.
- [7] N. Shlezinger, A. Amar, B. Luijten, R. J. G. van Sloun, and Y. C. Eldar, Deep Task-Based Analog-to-Digital Conversion, Jan. 2022. DOI: 10.48550/ ARXIV.2201.12634.
- [8] K. Tu, "Reliability challenges in 3D IC packaging technology," *Microelectronics Reliability*, vol. 51, no. 3, pp. 517–523, Mar. 2011. DOI: https://doi.org/10.1016/j.microrel.2010.09.031.
- [9] P. Muralt, N. Ledermann, J. Paborowski, et al., "Piezoelectric micromachined ultrasonic transducers based on PZT thin films," *IEEE Transactions on Ul*trasonics, Ferroelectrics, and Frequency Control, vol. 52, no. 12, pp. 2276– 2288, Dec. 2005. DOI: 10.1109/TUFFC.2005.1563270.
- [10] B. T. Khuri-Yakub and Ö. Oralkan, "Capacitive micromachined ultrasonic transducers for medical imaging and therapy," *Journal of Micromechanics* and Microengineering, vol. 21, no. 5, pp. 054004–054014, Apr. 2011. DOI: 10.1088/0960-1317/21/5/054004.

- [11] C. Chen, S. B. Raghunathan, Z. Yu, et al., "A Prototype PZT Matrix Transducer With Low-Power Integrated Receive ASIC for 3-D Transesophageal Echocardiography," *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 63, no. 1, pp. 47–59, Jan. 2016. DOI: 10.1109/TUFFC. 2015.2496580.
- [12] J. Li, C. Li, V. Henneken, et al., "25.8 Gb/s Submillimeter Optical Data Link Module for Smart Catheters," Journal of Lightwave Technology, vol. 40, no. 8, pp. 2456–2464, Apr. 2022. DOI: 10.1109/JLT.2021.3137981.
- [13] A. Liu, L. Liao, D. Rubin, et al., "High-speed optical modulation based on carrier depletion in a silicon waveguide," Optical Express, vol. 15, no. 2, pp. 660–668, Jan. 2007. DOI: 10.1364/0E.15.000660.

# SUMMARY

This thesis describes the design of application-specific integrated circuits (ASICs) for high-frame-rate 3D intracardiac echocardiography (ICE) probes. The work follows the whole process from the analysis of the application to the testing of a prototype with all envisioned functionality. It starts with the evaluation of the main challenges posed by ultrasound imaging from the tip of a catheter with a diameter in the order of 3 mm and its implications on the circuit design. A broad literature review in the field of ICE probe development but also regarding general ultrasound electronics provides the basis for this work and motivates later design choices. A high-frame-rate 3D imaging scheme is developed based on the needs of the application and leads to the design targets of the imager array and electronics. The final ASIC architecture is reached over the course of two prototypes to enable the evaluation of block-level functionality and advanced debugging. The thesis provides a detailed description of the circuit implementation with introduced novelties and shows characterization results in the electrical as well as acoustic domain. In the following, the contents of each chapter of the main matter are summarized in one paragraph.

#### Chapter 1

The first chapter is an introduction to the thesis and begins with a motivation regarding the significance of minimally-invasive procedures in view of the global impact of cardiovascular diseases. Out of the pool of medical imaging systems, ultrasound and in particular ICE procedures are thereafter identified as attractive tools to provide guidance for these closed-chest interventions. An overview of traditional ultrasound setups without ASICs in the probe is given and it is discussed how evolving applications lead to a requirement for in-probe circuits. This is followed by a summary of the common targets and techniques in the design of ultrasound ASICs. Zooming closer to the application, developments in the design of intracardiac imaging catheters are subsequently reviewed from its early stages in the 1950s to the current state of the art. This leads to a detailed analysis of the current challenges faced in the field, from power consumption in order to avoid excessive tissue heating to cable count in the narrow catheter, and identification

of opportunities for the presented work. The chapter concludes with an overview of the division and organization of the thesis.

#### Chapter 2

A discussion of the value of high-frame-rate 3D ICE to image phenomena like electromechanical waves in the heart is presented in this chapter and leads to the propositions of an imaging scheme for a 3D ICE probe. The design employs only 1D subarray beamforming in the elevation direction and can thus achieve a high frame rate while reducing the channel count sufficiently for realization in a 10-Fr catheter. To suppress the grating-lobe artifacts associated with subarray beamforming in the elevation direction, a limited number of fan-shaped beams with a wide azimuthal and narrow elevational opening angle are sequentially steered to insonify slices of the region of interest. An angular weighted averaging of reconstructed subvolumes further reduces the grating lobe artifacts. We optimize the transmit beam divergence and central frequency based on the required image quality for electromechanical wave imaging. Numerical simulation results show that a set of 7 fan-shaped transmission beams can provide a frame rate of 1000 volumes/s and a sufficient spatial resolution to visualize the electromechanical wave propagation on a large 3D surface.

#### Chapter 3

In this chapter, a compact high-voltage (HV) transmit circuit for dense 2D transducer arrays used in 3D ultrasonic imaging systems is presented. Stringent area requirements are addressed by a unipolar pulser with an embedded transmit/receive switch. Combined with a capacitive HV level shifter, it forms the ultrasonic HV transmit circuit with the lowest reported HV transistor count and area without any static power consumption. The balanced latched-based level shifter implementation makes the design insensitive to transients on the HV supply caused by pulsing, facilitating application in probes with limited local supply decoupling, such as imaging catheters. A prototype of  $8 \times 9$  elements was fabricated in TSMC 0.18-µm HV BCD technology and a PZT transducer matrix with a pitch of 160 µm is manufactured on the chip. Electrical characterization as well as acoustic results obtained with the 6 MHz central frequency transducer are demonstrated at up to 65 V transmit amplitude.

#### Chapter 4

An ASIC for 3D, high-frame-rate ultrasound imaging probes is presented in this

chapter. The design is the first to combine element-level, HV transmitters and analog frontends, subarray beamforming, and in-probe digitization in a scalable fashion for catheter-based probes. The integration challenge is met by a hybrid analogto-digital converter (ADC), combining an efficient charge-sharing successive-approximation-register (SAR) first stage and a compact single-slope second stage. Application in large ultrasound imaging arrays is facilitated by directly interfacing the ADC with a charge-domain subarray beamformer, locally calibrating inter-stage gain errors, and generating the SAR reference using a power-efficient local reference generator. Additional hardware-sharing between neighboring channels ultimately leads to the lowest reported area and power consumption across miniature ultrasound probe ADCs. A pitch-matched design is further enabled by an efficient split between the core circuitry and a periphery block, the latter including a datalink performing clock-data-recovery and time-division multiplexing (TDM), which leads to a 12-fold total channel-count reduction. Evaluation could be performed with the same assembly as used in the transmitter characterization and the functionality has been demonstrated electrically as well as in acoustic imaging experiments at up to 1000 volumens/s.

#### Chapter 5

This chapter describes the extension of the functionality of the first prototype in a second, also larger, version of the design. To reduce the number of cables in the catheter while maintaining a small footprint per element, transmit (TX) beamforming is realized on the chip with a combination of a shift register (SR) and a row/column approach. A compact beamformer design is further enabled by delay-encoding relative to the neighboring element and re-using SR cells for delay counting. To explore an additional cable-count reduction in the receiver part of the design, a channel with a combination of TDM, subarray beamforming, and multi-level data transmission is also included. This achieves an 18-fold cable-count reduction and minimizes the power consumption in the catheter by a load modulation (LM) cable driver. It is further explored how common-mode interference can limit beamforming gain and a strategy to reduce its impact with local regulators is discussed. The chip was fabricated in TSMC 0.18-µm HV BCD technology and a 2D PZT transducer matrix of  $16 \times 18$  elements with a pitch of 160  $\mu$ m and a center frequency of 6 MHz was manufactured on the chip again. The functionality and operation of up to 1000 volumes/s are demonstrated in electrical and acoustic imaging experiments.

#### Chapter 6

The last chapter concludes the thesis by providing an overview of the presented work and summarizing essential achievements. Several original contributions have been made, including propositions in the HV transmitter, receive frontend, and datalink of an ultrasound imaging ASIC but also the novel system architecture and imaging scheme themselves. The chapter also lists several, more general, findings on the system and circuit level design that have been obtained in the course of the work. In the end, an outlook on possible future opportunities for ultrasound imaging catheters is given, ranging from circuit-level improvements on the transmitter to 3D IC stacking and the potential of signal processing or optical interconnect.

# SAMENVATTING

Dit proefschrift beschrijft het ontwerp van toepassingsspecifieke geïntegreerde schakelingen (ASICs) voor 3D intracardiale echocardiografie (ICE) katheters met een hoge framesnelheid. Het werk beschrijft het hele analyse proces voor de applicatie tot en met het testen van een prototype met alle beoogde functionaliteit. Het proefschrift begint met de evaluatie van de belangrijkste uitdagingen van ultrasone beeldvorming vanaf de punt van een katheter met een diameter in de orde van 3 mm en de implicaties ervan voor het circuitontwerp. Een brede literatuurstudie op het gebied van de ontwikkeling van ICE-katheters en algemene ultrasone elektronica, vormt de basis voor dit werk en motiveert latere ontwerpkeuzes. Een architectuur voor 3D-beeldvorming met hoge framesnelheid is ontwikkeld op basis van de benodigdheden voor de toepassing en de eisen voor de transducer matrix en bijbehorende elektronica voor beeldvorming zijn ervan afgeleid. De definitieve ASIC architectuur is na de ontwikkeling van twee prototypes vastgesteld om de functionaliteit van het systeem op blokniveau te kunnen evalueren en een geavanceerde debugging mogelijk te maken. Het proefschrift geeft een gedetailleerde beschrijving van de implementatie van de schakeling met geïntroduceerde innovaties en toont karakteriseringsresultaten in zowel het elektrische als het akoestische domein. In de volgende paragrafen worden de hoofdzaken van elk hoofdstuk samengevat.

#### Hoofdstuk 1

Het eerste hoofdstuk is de inleiding van het proefschrift en begint met een motivatie met betrekking tot het belang van minimaal invasieve procedures, gegeven de mondiale impact van hart- en vaatziekten. Uit de verzameling van medische beeldvormingssystemen worden echografie en in het bijzonder ICE-procedures vervolgens geïdentificeerd als aantrekkelijke instrumenten om begeleiding te bieden in deze gesloten-borst-interventies. Daarna wordt een overzicht gegeven van traditionele medische ultrageluid opstellingen zonder ASICs in de katheter en er wordt behandeld hoe evoluerende toepassingen leiden tot een vereiste voor circuits in de katheter. Dit wordt gevolgd door een samenvatting van de gemeenschappelijke doelen en technieken in het ontwerp van ultrageluid ASICs. In de context van de eerdergenoemde toepassing worden de ontwikkelingen in het ontwerp van intracardiale beeldvormende katheters vervolgens behandeld vanaf de vroege stadia in de jaren 1950 tot de huidige stand van de techniek. Dit leidt tot een gedetailleerde analyse van de huidige uitdagingen in het veld, zoals het energieverbruik om overmatige weefselverwarming te voorkomen, de hoeveelheid kabels in de smalle katheter en waar het gepresenteerde werk verdere verbeteringen kan bieden. Het hoofdstuk wordt afgesloten met een overzicht van de indeling en organisatie van het proefschrift.

#### Hoofdstuk 2

De waarde van 3D ICE met hoge framesnelheid voor het afbeelden van fenomenen zoals elektromechanische golven in het hart wordt in dit hoofdstuk behandeld waarna het systeem architectuur voor een 3D ICE-katheter wordt geïntroduceerd. Het ontwerp maakt alleen gebruik van 1D subarray beamforming in de elevatierichting waardoor een hoge framesnelheid kan worden bereiken terwiil het aantal kanalen voldoende wordt verminderd om in een 10-Fr katheter te kunnen worden geïmplementeerd. Om de grating lobe artefacten geassocieerd met subarray beamforming in de elevatierichting te onderdrukken, wordt een beperkt aantal waaiervormige bundels met een brede azimutale en smalle elevatie-openingshoek achtereenvolgens gestuurd om segmenten van het interessegebied te verduisteren. Een hoekgewogen middeling van gereconstrueerde subvolumes vermindert de grating lobe artefacten verder. We optimaliseren de zendbundeldivergentie en centrale frequentie op basis van de vereiste beeldkwaliteit voor elektromechanische golfbeeldvorming. Numerieke simulatieresultaten laten zien dat een set van 7 waaiervormige transmissiestralen een framesnelheid van 1000 volumes/s en een voldoende ruimtelijke resolutie kan bieden om de elektromechanische golfvoortplanting op een groot 3D-oppervlakte te visualiseren.

#### Hoofdstuk 3

In dit hoofdstuk wordt een compact hoogspannings (HV) zendcircuit gepresenteerd voor dichte 2D-transducerarrays die worden gebruikt in 3D ultrasone beeldvormingssystemen. Een strenge oppervlakte beperking motiveert het gebruik van een unipolaire pulser met een ingebouwde zend-/ontvangstschakelaar. Gecombineerd met een capacitieve HV level shifter, vormen deze het ultrageluid HV-zendcircuit met de laagste gerapporteerde HV-transistortelling en oppervlak zonder enig statisch energieverbruik. De HV level shifter op basis van een gebalanceerde latch maakt het ontwerp ongevoelig voor transiënten op de HV-voeding veroorzaakt door pulsen, wat de toepassing in katheters met beperkte lokale voedingsontkoppeling, zoals beeldvormende katheters, vergemakkelijkt. Een prototype van  $8 \times 9$  elementen is gefabriceerd in een TSMC 0.18-µm HV BCD fabricage proces en een PZT-transducermatrix met een pitch van 160 µm is op de chip gefabriceerd. Elektrische karakterisering en akoestische resultaten die zijn verkregen met de 6-MHz centrale frequentietransducer worden gedemonstreerd bij een zendamplitude tot en met 65 V.

#### Hoofdstuk 4

In dit hoofdstuk wordt een ASIC voor 3D-katheters met hoge beeldsnelheid voor echografie gepresenteerd. Het ontwerp is het eerste dat HV-zenders en analoge frontends op elementniveau, subarray beamforming en in-katheter digitalisering op een schaalbare manier integreert binnen de katheter. De integratie-uitdaging wordt aangegaan door een hybride analoog-naar-digitaalomzetter (ADC), waarbij een successive approximation (SAR) architectuur (in een eerste trap) en een singleslope architectuur (in een tweede trap) is gecombineerd. De toepassing in grote ultrasone beeldvormingsarrays wordt vergemakkelijkt door de ADC rechtstreeks te verbinden met een ladingsdomein subarray beamformer, lokaal versterkingsfouten tussen de fasen te kalibreren en de SAR-referentie te genereren met behulp van een energiezuinige lokale referentiegenerator. Door onder andere aangrenzende kanalen hardware te laten delen, bereikt het ontwerp het laagste oppervlakte en energieverbruik van alle gerapporteerde ADCs voor ultrageluid katheters. Een op oppervlakte afgestemd ontwerp wordt verder mogelijk gemaakt door een efficiënte splitsing tussen de kerncircuits en een periferieblok, waarbij de laatste een datalink omvat die klok-gegevensherstel en time domain multiplexing (TDM) uitvoert, wat de totale hoeveelheid kanalen 12-voudig verminderd. Evaluatie is uitgevoerd met dezelfde testopstelling als gebruikt bij de karakterisering van de zender en de functionaliteit is zowel elektrisch als in akoestische beeldvormingsexperimenten aangetoond tot 1000 volumes/s.

#### Hoofdstuk 5

Dit hoofdstuk beschrijft de uitbreiding van de functionaliteit van het eerste prototype in een vergrote tweede versie van het ontwerp. Om het aantal kabels in de katheter te verminderen met behoud van een kleine footprint per element, wordt op de chip zend (TX) bundelvorming gerealiseerd met een combinatie van een schuifregister (SR) en een rij/kolombenadering. Een compact bundelvormer-ontwerp wordt verder mogelijk gemaakt door delay encoding ten opzichte van het naburige element en het hergebruik van SR-cellen voor het tellen van de vertragingsstappen. Om een extra reductie van het aantal kabels in het ontvangergedeelte van het ontwerp te onderzoeken, is ook een kanaal geïmplementeerd met een combinatie van TDM, subarray beamforming en een multi-level datatransmissie. Dit zorgt voor een 18-voudige vermindering van het aantal kabels en minimaliseert het energieverbruik in de katheter door een last modulatie (LM) kabeldriver. Verder wordt behandeld hoe common-mode interferentie de versterking van de bundelvorming kan beperken en hoe de impact hiervan verminderd kan worden met lokale spanningsregelaars. De chip is gefabriceerd in een TSMC 0.18- $\mu$ m HV BCD fabricage proces en een 2D PZT transducer matrix van 16×18 elementen met een pitch van 160  $\mu$ m en een middenfrequentie van 6 MHz is opnieuw op de chip gefabriceerd. De functionaliteit en werking tot maximaal 1000 volumes/s worden gedemonstreerd in elektrische en akoestische beeldvormingsexperimenten.

#### Hoofdstuk 6

Het laatste hoofdstuk sluit het proefschrift af met een overzicht van het gepresenteerde werk en een samenvatting van de belangrijkste verworvenheden. Er zijn meerdere originele bijdragen geleverd, waaronder in de HV-zender, de ontvangstfrontend en datalink van een ASIC voor ultrasone beeldvorming. Echter, ook de nieuwe systeemarchitectuur en wijze van beeldvorming bevat innovaties. Het hoofdstuk somt ook een aantal, meer algemene bevindingen op over het ontwerp op systeem- en circuitniveau die in de loop van het werk zijn gepresenteerd. Uiteindelijk wordt een blik geworpen op toekomstige mogelijkheden voor ultrasone beeldvormingskatheters, variërend van verbeteringen op circuitniveau aan de zender tot 3D IC-stapeling en potentiële verbeteringen met behulp van signaalbewerking of optische interconnectie.

# LIST OF ABBREVIATIONS

| 1D             | One-Dimensional                                |
|----------------|--|
| $2\mathrm{D}$  | Two-Dimensional                                |
| 3D             | Three-Dimensional                              |
| ADC            | Analog-to-Digital Converter                    |
| $\mathbf{AF}$  | Atrial Fibrillation                            |
| AFE            | Analog Frontend                                |
| ASIC           | Application-Specific Integrated Circuit        |
| AWG            | Arbitrary Waveform Generator                   |
| BCD            | Bipolar-CMOS-DMOS                              |
| BER            | Bit Error Rate                                 |
| $\mathbf{BF}$  | Beamformer                                     |
| CAD            | Computer-Aided Design                          |
| CDAC           | Capacitive Digital-to-Analog Converter         |
| $\mathbf{CDR}$ | Clock-Data Recovery                            |
| CMOS           | Complementary Metal-Oxide-Semiconductor        |
| CMUT           | Capacitive Micromachined Ultrasound Transducer |
| $\mathbf{CNR}$ | Contrast-to-Noise Ratio                        |
| $\mathbf{CT}$  | Computed Tomography                            |
| CTC            | Continuous-Time Comparator                     |
| CVD            | Cardiovascular Disease                         |
| DAC            | Digital-to-Analog Converter                    |
| DAS            | Delay-and-Sum                                  |
| DC             | Direct Current                                 |
| DFE            | Decision Feedback Equalization                 |
| DFF            | D-Flip-Flop                                    |
| $\mathbf{DLL}$ | Delay Locked Loop                              |
| DR             | Dynamic Range                                  |
| DTC            | Discrete-Time Comparator                       |
| DTI            | Deep Trench Isolation                          |
|                |  |

| ECG                    | Electrocardiogram                   |
|------------------------|-------------------------------------|
| ESD                    | Electrostatic Discharge             |
| $\mathbf{E}\mathbf{W}$ | Electromechanical Wave              |
| EWI                    | Electromechanical Wave Imaging      |
| FDA                    | Food and Drug Administration        |
| FDM                    | Frequency-Division Multiplexing     |
| FF                     | Flip-Flop                           |
| FIFO                   | First-in First-out                  |
| FPGA                   | Field-Programmable Gate Array       |
| $\operatorname{GL}$    | Grating Lobe                        |
| GLR                    | Grating Lobe Residual               |
| HV                     | High-Voltage                        |
| IC                     | Integrated Circuit                  |
| ICE                    | Intracardiac Echocardiography       |
| IVUS                   | Intravascular Ultrasound            |
| LDO                    | Low-Dropout Regulator               |
| LFSR                   | Linear-Feedback Shift Register      |
| $\mathbf{L}\mathbf{M}$ | Load Modulation                     |
| LNA                    | Low-Noise Amplifier                 |
| LSB                    | Least-Significant Bit               |
| $\mathbf{LSV}$         | Lateral Shift-Variance              |
| LV                     | Low-Voltage                         |
| LVDS                   | Low-Voltage Differential Signaling  |
| MEMS                   | Microelectromechanical Systems      |
| MIM                    | Metal-Insulator-Metal               |
| MOM                    | Metal-Oxide-Metal                   |
| MOSCAP                 | Metal-Oxide-Semiconductor Capacitor |
| MRI                    | Magnetic Resonance Imaging          |
| MSB                    | Most-Significant Bit                |
| MUX                    | Multiplexer                         |
| NAND                   | Negative-AND                        |
| NOR                    | Negative-OR                         |
| NOV                    | Non-Overlapping                     |
| PAM                    | Pulse Amplitude Modulation          |
| PCB                    | Printed Circuit Board               |
| PE                     | Pulse-Echo                          |

| PGA                     | Programmable-Gain Amplifier                        |
|-------------------------|--|
| PMUT                    | Piezoelectric Micromachined Ultrasound Transducers |
| PRBS                    | Pseudo-Random Bit Sequence                         |
| PRF                     | Pulse Repetition Frequency                         |
| PRI                     | Pulse Repetition Interval                          |
| PRNG                    | Pseudo-Random Number Generator                     |
| PSF                     | Point-Spread-Function                              |
| PSRR                    | Power Supply Rejection Ratio                       |
| PVT                     | Process/Voltage/Temperature                        |
| PZT                     | Lead Zirconium Titanate                            |
| $\mathbf{R}/\mathbf{C}$ | $\operatorname{Row}/\operatorname{Column}$         |
| RMS                     | Root Mean Square                                   |
| RS                      | Reset-Set  |
| RX                      | Receive  |
| RZ                      | Return-to-Zero                                     |
| S/H                     | Sample-and-Hold                                    |
| S2D                     | Single-Ended to Differential                       |
| SAP                     | Subaperture Processing                             |
| SAR                     | Successive-Approximation-Register                  |
| SNDR                    | Signal-to-Noise-and-Distortion Ratio               |
| SNR                     | Signal-to-Noise Ratio                              |
| SOI                     | Silicon-on-Insulator                               |
| SR                      | Shift Register                                     |
| SS                      | Single-Slope                                       |
| $\mathbf{T}/\mathbf{R}$ | Transmit/Receive                                   |
| TDM                     | Time-Division Multiplexing                         |
| TEE                     | Transesophageal Echocardiography                   |
| TGC                     | Time Gain Compensation                             |
| $\mathbf{TSV}$          | Through Silicon Via                                |
| TTE                     | Transthoracic Echocardiography                     |
| TX                      | Transmit   |
| VCSEL                   | Vertical-Cavity Surface-Emitting Laser             |
| WHO                     | World Health Organization                          |
| $\mu BF$                | Micro-Beamforming                                  |

# LIST OF PUBLICATIONS

Y. Hopf, B. Ossenkoppele, M. Soozande, E. Noothout, Z. Chang, C. Chen, H. Vos, J. Bosch, M. Verweij, N. de Jong, M. Pertijs, "A Pitch-Matched Transceiver ASIC With Shared Hybrid Beamforming ADC for High-Frame-Rate 3-D Intracardiac Echocardiography," in *IEEE Journal of Solid-State Circuits*, vol. 57, no. 11, pp. 3228 - 3242, Nov. 2022, (invited).

Y. Hopf, B. Ossenkoppele, M. Soozande, E. Noothout, Z. Chang, C. Chen, H. Vos, J. Bosch, M. Verweij, N. de Jong, M. Pertijs, "Transceiver ASIC Design for High-Frame-Rate 3D Intracardiac Echocardiography," 2022 IEEE International Ultrasonics Symposium (IUS), Oct. 2022, (abstract & poster).

M. Soozande, B. Ossenkoppele, **Y. Hopf**, M. Pertijs, M. Verweij, N. de Jong, H. Vos, J. Bosch, "Imaging Scheme for 3-D High Frame Rate Intracardiac Echography: A Simulation Study," in *IEEE Transactions on Ultrasonics, Ferroelectrics, and Frequency Control*, vol. 69, no. 10, pp. 2862 - 2874, Oct. 2022.

**Y. Hopf**, M. Pertijs, "A Pitch-Matched ASIC with Integrated 65V TX and Shared Hybrid Beamforming ADC for Catheter-Based High-Frame-Rate 3D Ultrasound Probes," *Annual Workshop on Circuits, Systems and Signal Processing (ProRISC)*, July 2022, (Best presentation award).

Y. Hopf, B. Ossenkoppele, M. Soozande, E. Noothout, Z. Chang, H. Vos, J. Bosch, M. Verweij, N. de Jong, M. Pertijs, "A Compact Integrated High-Voltage Pulser Insensitive to Supply Transients for 3-D Miniature Ultrasound Probes," in *IEEE Solid-State Circuits Letters*, vol. 5, pp. 166-169, June 2022.

Y. Hopf, B. Ossenkoppele, M. Soozande, E. Noothout, Z. Chang, C. Chen, H. Vos, J. Bosch, M. Verweij, N. de Jong, M. Pertijs, "A Pitch-Matched ASIC with Integrated 65V TX and Shared Hybrid Beamforming ADC for Catheter-Based High-Frame-Rate 3D Ultrasound Probes," 2022 IEEE International Solid-State Circuits Conference (ISSCC), Mar. 2022, pp. 494-496.

Y. Hopf, B. Ossenkoppele, M. Soozande, E. Noothout, Z. Chang, H. Vos, J. Bosch, M. Verweij, N. de Jong, M. Pertijs, "A Compact Integrated High-Voltage Pulser for 3D Miniature Ultrasound Probes," *Annual Workshop on Circuits, Systems and Signal Processing (ProRISC)*, July 2021, (poster).

M. Soozande, B. Ossenkoppele, **Y. Hopf**, M. Pertijs, M. Verweij, H. Vos, J. Bosch, N. de Jong, "Feasibility of High Frame Rate 3-D Intracardiac Echography using Fan-Beam Transmissions," 2019 IEEE International Ultrasonics Symposium (IUS), Oct. 2019, (abstract & poster).

Y. Hopf, B. Ossenkoppele, M. Soozande, E. Noothout, Z. Chang, H. Vos, J. Bosch, M. Verweij, N. de Jong, M. Pertijs, "Enabling High Frame Rate 3D Intracardiac Echography," *Annual Workshop on Circuits, Systems and Signal Processing (ProRISC)*, July 2019, (poster).

# ACKNOWLEDGEMENTS

"Der Weg ist das Ziel", as we say in Germany, the journey is the reward. This is something that I have truly experienced in this venture over the past four years. But as the destination is still drawing closer, I would like to take the time to reflect and express my gratitude to those that paved the way. Without their great support and kind guidance, this endeavour would not have been possible.

First and foremost, I would like to thank my promotor Dr. Michiel Pertijs. Our first contact goes back to 2018, when an open position in his lab caught my attention while searching for research opportunities. During the application procedure. we had regular interaction and from the start, I had the impression that Michiel had exceptional technical expertise across a large range of fields and a remarkable ability to enter a new topic and directly identify and target the essential aspects. Paired with the skill to immediately determine the current position of his vis-à-vis in a discussion and distinct ingenuity, he struck me as the ideal supervisor and mentor. Since then, there has not been a single occasion where I didn't fully appreciate our collaboration and the insights and competences I gained during this time are priceless to me. It takes great leadership and vision to provide the freedom to explore new ideas in a self-determined way while still offering the guidance and discussion that is necessary to lead a work to a quality fit for presentation in high-impact formats. Of course, a project of this complexity is not progressing without setbacks but strong crisis management from Michiel and the environment that he has created in the group have enabled a nearly seamless continuation in times of difficult debugging, foundry congestion or covid-19. Finally, I would like to emphasize that I experienced our work together not only as productive but also very pleasant and that I am looking forward to remaining in contact in the future.

I would also like to extend my gratitude to my second promotor, Prof. Nico de Jong. While, as head of the Acoustical Wavefield Imaging Laboratory at TU Delft, he was further from my daily activities as a circuit designer, his profound knowledge of the acoustic side was essential to the success of the project. The facilities and group that he established in his lab enabled us to jointly arrive at a desirable design target, manufacture prototypes and apply them in conclusive experiments. I particularly appreciate all the discussions we had on the debugging of the transducer to ASIC interface and his personal effort in the investigation of possible error sources. Lastly, I would also like to thank Nico for his individual guidance throughout the PhD and for his continued role in the project even after transitioning into retirement during its course.

In addition, I would like to thank Dr. Johan Bosch, Dr. Martin Verweij and Dr. Hendrik Vos, whose vision and initial work together with Michiel and Nico lead to the formation of the project I came to work on in the first place. Only due to the fruitful collaboration and constant exchange could this multi-disciplinary work get to the place it is now and I have gained much insight into the world of ultrasound under their guidance.

The project out of which this thesis originated was split into three branches with me being responsible for the design of the electronics, Boudewine taking over the acoustic side and Mehdi looking into imaging aspects. In addition to that, Djalma from the Acoustical Wavefield Imaging Laboratory joined the ongoing work on acoustic characterization towards the end of the project. I am truly grateful for the strong collaboration throughout this venture and all the fundamental things they have taught me about their research areas.

This work was founded through the Dutch Research Council (NWO) by the Open Technology Program under Grant 14279 with industrial partners Sensixs Design BV, Philips, and Oldelft Ultrasound BV. I would like to extend my gratitude for enabling this research and for all the input we received.

A project of this magnitude cannot be realized within four years without the infrastructure to support it. It is thus important to me to acknowledge the most essential pillars that our work relied on within the Department of Microelectronics, starting with its head, Prof. Kofi Makinwa. The environment that he has created enables and stimulates innovation and it is not to a small degree his great reputation that has attracted people from all over the world to form a leading microelectronics research hotspot at TU Delft. I would also like to thank our team of secretaries, especially our main contact Joyce, for doing all the organization around the department and allowing us to focus on our research work. And last but not least, the crucial work of the technical staff at the Department of Microelectronics as well as the Acoustical Wavefield Imaging Laboratory should be recognized. The accumulated experience and skill among Antoon, Emile, Henry, Lukasz, Ron, and Zu Yao is absolutely mandatory in the maintenance of all research groups and the only way circuit designs, transducer assemblies and measurement setups of this complexity can be realized within the time of a PhD project.

Another institution of great benefit to this work was the Smart Ultrasound Group led by Michiel at TU Delft. Over the years, its members Chao, Douwe, Eunchul, Jaesung, Jing, Mingliang, Nikola, Nuriel, Peng, Teahoon, and Zhao have greatly contributed to my knowledge of tools and many insights emerged from detailed discussions about our respective issues. Several of the investigated techniques are based on the knowledge from prior works of the group and the input from architecture and design reviews during the group meeting was invaluable to the development of the presented design.

Also beyond the limits of the group, many fellow PhD candidates in the Department of Microelectronics have greatly shaped this work in different ways. The exchange of experience, teaching of tricks, and discussion of complicated matters, if not for anything else, just to order one's thoughts or get some relief, is something that I experienced as necessary in the process of becoming a better engineer and researcher. But also outside the office, I appreciated the time spent together with my colleagues from the group as well as Accel, Alireza, Amir, Arthur, Çağrı, Efraïm, Eren, Hui, Jeakyum, Mojtaba, Roger, Shoubhik, Sundeep, Thije, and many more a lot.

As an intense work requires good relief, I would also like to mention some friends I gained during my time in Delft. Alvaro, Arthur, Canmanie, Carmel, Dim, Eleni, Giota, Henrique, Ini, Jaime, Luis, Lyuba, Mari, Maud, Nuria, Roger, Samanta, Susannah and many more enabled me to disconnect from work and refresh in the weekends and evenings before heading back in. Additionally, I would like to thank my closest friends from Germany, who have shaped my life in major ways and shown that distance is merely affecting the frequency with which you physically meet.

It is moreover important for me to express my appreciation for all the support I have received from my family. From my parents that have raised me to be a confident, independent person and enabled me to follow my interests regardless of where it took me. But also from my three brothers, whose weekly meeting has become a constant in my life and whose company I value a lot.

In the end, I would like to thank Lea for all the love and unconditional support since we met. Thank you for being with me despite the long distance, your un-
derstanding nature, and sharing so many precious moments. Seeming like a rock is easy when you know you have so much to fall back on.

Yannick Hopf Delft, October 2022

## ABOUT THE AUTHOR



Yannick Hopf was born in Seligenstadt, Germany in July 1991.

He received his B.Sc. and M.Sc. degree (*cum laude*) in electrical engineering from the Technical University of Darmstadt, Darmstadt, Germany, in 2014 and 2017 with the M.Sc. graduation being honored with the Eckelmann AG price for the best degree in his year. Since 2018, he is pursuing a Ph.D. degree in electrical engineering at the Electronic Instrumentation Laboratory, Delft University of Technology, Delft, The Netherlands, with a focus on ASIC design for 3D high-frame-rate medical ultrasound imaging.

In his spare time, he enjoys socializing, dancing, cultural events, and traveling.